UNIVERSITY OF ALBERTA

AN ELASTIC INHOMOGENEITY WITH PERFECTLY BONDED INTERFACE SUBJECTED TO A GENERAL CLASS OF NONUNIFORM REMOTE LOADING

BY



A thesis submitted to the Faculty of Graduate Studies and Research in partial fulfillment of the requirements for the degree of Master of Science

Department of Mechanical Engineering

Edmonton, Alberta

Fall 2007



Library and Archives Canada

Published Heritage Branch

395 Wellington Street Ottawa ON K1A 0N4 Canada Bibliothèque et Archives Canada

Direction du Patrimoine de l'édition

395, rue Wellington Ottawa ON K1A 0N4 Canada

> Your file Votre référence ISBN: 978-0-494-33279-5 Our file Notre référence ISBN: 978-0-494-33279-5

NOTICE:

The author has granted a nonexclusive license allowing Library and Archives Canada to reproduce, publish, archive, preserve, conserve, communicate to the public by telecommunication or on the Internet, loan, distribute and sell theses worldwide, for commercial or noncommercial purposes, in microform, paper, electronic and/or any other formats.

The author retains copyright ownership and moral rights in this thesis. Neither the thesis nor substantial extracts from it may be printed or otherwise reproduced without the author's permission.

AVIS:

L'auteur a accordé une licence non exclusive permettant à la Bibliothèque et Archives Canada de reproduire, publier, archiver, sauvegarder, conserver, transmettre au public par télécommunication ou par l'Internet, prêter, distribuer et vendre des thèses partout dans le monde, à des fins commerciales ou autres, sur support microforme, papier, électronique et/ou autres formats.

L'auteur conserve la propriété du droit d'auteur et des droits moraux qui protège cette thèse. Ni la thèse ni des extraits substantiels de celle-ci ne doivent être imprimés ou autrement reproduits sans son autorisation.

In compliance with the Canadian Privacy Act some supporting forms may have been removed from this thesis.

While these forms may be included in the document page count, their removal does not represent any loss of content from the thesis. Conformément à la loi canadienne sur la protection de la vie privée, quelques formulaires secondaires ont été enlevés de cette thèse.

Bien que ces formulaires aient inclus dans la pagination, il n'y aura aucun contenu manquant.



This dissertation is dedicated to my father, Jo In Kim

ABSTRACT

Problems involving elastic inhomogeneities have received considerable attention in the literature. In many of their case, complex-variable methods are used extensively and successfully to produce exact solutions and significant practical results in problems of linear plane and anti-plane elastostatics. However, such exact analyses have been more or less absent in the analogous problems from *finite* elasticity. Recent studies have provided relatively simple complex-variable formulations for problems involving elastic inhomogeneities for the class of *harmonic materials*. The results, however, are limited exclusively to uniform remote loading, whereas, in nature, remote loadings are mostly nonuniform.

In present study, we develop a general methodology for the problems of plane *finite* deformations of harmonic composite materials containing an elastic inhomogeneity subjected to nonuniform remote loading. In addition, we also derive important results of mechanics of materials concerning the subject of maintaining uniform stress distribution inside the inhomogeneity subjected to nonuniform remote loading.

ACKNOWLEDGEMENTS

I would like to extend my deepest gratitude to Dr. Peter Schiavone for his invaluable guidance, support and kind understanding, as a result of which this dissertation has been accomplished.

I am also grateful to all my friends for their everlasting encouragement. Special thanks to my family for their consistent love, support and encouragement throughout my studies.

> Chun Il Kim Edmonton, AB June 2007

CONTENTS

1	IN	INTRODUCTION1			
	1.1	GENERAL BACKGROUND 1			
		1.1.1 INTRODUCTION			
		1.1.2 COMPOSITE MATERIAL			
	1.2	SINGLE INHOMOGENEITY MODEL			
	1.3	INTERPHASE LAYER			
		1.3.1 PERFECT BONDING			
		1.3.2 IMPERFECT BONDING			
	1.4	PURPOSE OF STUDY AND DISSERTATION OVERVIEW			
2	GE	NERAL FORMULATION AND PRELIMINARIES 12			
	2.1	INTRODUCTION			
	2.2	HARMONIC MATERIALS			
		2.2.1 COMPLEX-VARIABLE FORMULATION			
		2.2.2 A PARTICULAR CLASS OF HARMONIC MATERIAL			
	2.3	PREREQUISITES			
		2.3.1 NOTATION AND INTERFACE FORMULATION			

		2.3.2	CHARACTERIZING REMOTE LOADING AND THE UNDEFORMED STAGE		
		2.3.3	SUBSIDIARY RESULTS		
3 COMPLETE SOLUTION OF AN ELLIPTIC INHO- MOGENEITY SUBJECTED TO A GENERAL CLASS OF NONUNIFORM REMOTE LOADINGS IN FINITE ELASTICITY					
	3.1	INTR	ODUCTION27		
	3.2	NOTA	ATION AND PREREQUISITES		
	3.3	COM	PLETE SOLUTION FOR THE ELLIPTIC INHOMOGENEITY 29		
	3.4	FURT	THER DISCUSSION		
		3.4.1	Expressions for I and J Inside the Elliptic Inhomogeneity		
		3.4.2	Stress Distribution		
	3.5	EXAN	MPLE: The Case $a = b = 3$, $\delta_o = 1$: Quadratic Remote Loading		
		3.5.1	Complete Solutions		
		3.5.2	Determination of I and J Inside the Elliptic Inhomogeneity		
		3.5.3	Deformed Contour		
		3.5.4	Admissibility Conditions		
		3.5.5	The Piola Stress Distribution		
	3.6	THE	CASE OF THE CIRCULAR INHOMOGENEITY		
		3.6.1	Complete Solution in the Case of the Circular Inhomogeneity		
		3.6.2	Expressions for I and J Inside the Circular Inhomogeneity		
		3.6.3	Stress Distribution		
		3.6.4	Example: The Case $a = b = 2$		
	3.7	CON	CLUSIONS		

4	ESI	SHELBY'S CONJECTURE			
	4.1	1 INTRODUCTION		63	
	4.2	2 COMPLEX VARIABLE FORMULATION			
		4.2.1	Case 1 : When $S\overline{D_n} = 0$	72	
		4.2.2	Case 2 : When $(\alpha_1 K - \alpha_2) C_n = 0$	73	
		4.2.3	Case 3 : When $S\overline{D_n} = (\alpha_1 K - \alpha_2) C_n = 0$	73	
		4.2.4	Case 4 : When $S\overline{D_n}$, $(\alpha_1 K - \alpha_2) C_n \neq 0$	73	
	4.3	SOLU	TION FOR THE UNIFORM INTERIOR STRESS CASE	78	
	4.4	FURT	HER DISCUSSION	81	
		4.4.1	Expressions for I and J Inside the Inhomogeneity	81	
		4.4.2	Stress Distribution Inside Inhomogeneity	83	
		4.4.3	Deformed Contour	84	
		4.4.4	Admissibility Conditions	85	
	4.5	CONC	CLUSIONS	85	
5	5 DESIGNING AN INHOMOGENEITY WITH UNIFORM INTERIOR STRESS				
	5.1	1 INTRODUCTION		86	
	5.2	UNIF	ORM STRESS DISTRIBUTION INSIDE THE INHOMOGENEITY	87	
		5.2.1	Case 1: When $SD = 0$, $KC + \frac{S_{\beta_2}}{\overline{C}} \neq 0$	89	
		5.2.2	Case 2: When $KC + \frac{\mathscr{B}_{\beta_2}}{\overline{C}} = 0, \mathscr{B}D \neq 0$	90	
		5.2.3	Case 3: When $SD \neq 0$ and $KC + \frac{S_{\beta_2}}{C} \neq 0$	90	
	5.3	COMPLETE SOLUTION OF A HYPOTROCHOIDAL INHOMOGENEITY . 92			
	5.4	SOLU	TION OF A HYPOTROCHOIDAL INHOMOGENEITY (GENERAL).	98	

5.	5 FUF	FURTHER DISCUSSION10	
	5.5.	1 Expressions for I and J Inside the Inhomogeneity	
	5.5.2	2 Stress Distribution	
5.	.6 EXA	AMPLE: THE CASE $n = 3$	
	5.6.	1 Complete Solutions106	
	5.6.2	2 Determination of I and J Inside the Inhomogeneity	
	5.6.3	3 Deformed Contour	
	5.6.4	4 Admissibility Conditions108	
	5.6.	5 The Piola Stress Distribution109	
5.	.7 COI	NCLUSIONS	
6 C	CONCI	LUSIONS AND FUTURE WORK 111	
6.	.1 COI	NCLUSIONS	
6.	.2 FU1	TURE WORK	
REJ	FERE	NCES	
A D	DERIV	ATIONS	
A	.1 DEF	RIVATION OF RESULTS IN CHAPTER. 3	
А	.2 DEF	RIVATION OF RESULTS IN CHAPTER. 4	

LIST OF FIGURES

FIGURE 1: Showing Three Types of Composite Materials 2
FIGURE 2: Several Orientations of Fiber-Reinforced Composites 4
FIGURE 3: The Conformal Mapping from $z - plane$ to $\xi - plane$
FIGURE 4: Example of a Deformed Elliptic Inhomogeneity Subjected to a Remote Loading Characterized by (3.28)
FIGURE 5: Example of a Deformed Circular Inhomogeneity Subjected to a Remote Loading Characterized by (3.48)
FIGURE 6: The Conformal Mapping from $z - plane$ to $\xi - plane$
FIGURE 7: First Root (equation 4.27) ····· 75
FIGURE 8: Second Root (equation 4.27) ······ 76
FIGURE 9: Deformed Contour of the Elliptic Inhomogeneity
FIGURE 10: Deformed Contour of the Hypotrochoidal Inhomogeneity with 3 Cusps $(n = 2)$
FIGURE 11: Example of the Hypotorchoidal Inhomogeneity with $n+1$ cusps
FIGURE 12: Deformed Contour of the Hypotrochoidal Inhomogeneity with 4 Cusps $(n = 3)$

.

Chapter 1 INTRODUCTION

1.1 GENERAL BACKGROUND

1.1.1 INTRODUCTION

Materials are widely used in modern industry for various purposes. Many of their successful applications can be found in, for example, construction fields, aerospace, biomedical applications, automobile industries and space structures. However, not every application of materials is successful at the initial stage. For example, a beverage can was first made from galvanized iron. Yet, a galvanized iron can is heavy and inadequately strong for a beverage can and hence, later, was replaced by an aluminum can. The process of finding "proper materials" for each specific purpose is not an easy task and often requires trial and error. Therefore, it is of fundamental importance that one really needs to understand the mechanical behavior of materials before one uses them.

Among various modern materials, composite materials have gained particular interest, due to the fact that technological developments in the use of materials inevitably induce severe and highly demanding working conditions so that conventional materials such as ceramics and polymeric materials may no longer meet such intensive service requirements.

1.1.2 COMPOSITE MATERIAL

Composite materials are generally referred to as multiphase materials consisting of two or more distinct constituents, which are combined to achieve desirable mechanical, thermal and/or electronic properties depending on specific service requirements. Based on the structural shape of the materials, composite materials are classified into three different classes namely; particle-reinforced, fiber-reinforced and laminar composites (see Figure 1).



Figure 1: Showing Three Types of Composite Materials: (a) Particulate Composite, (b) Fiber-Reinforced Composite, and (c) Laminar Composite (From Sudak, 2000)

Particle-reinforced composites consist of a matrix material into which particles are embedded. These composites are generally subdivided into two classes referred to as largeparticle composites and dispersed-strengthened composites. The large particle composites containing large amounts of coarse particles which are harder and stiffer than the matrix is manufactured to obtain specific combinations of properties rather than to improve strength. The application of the large particle composite can be found where high corrosion and fatigue-resistance are required (Tjong et al, 2005; Jian-Zhong et al, 2006). In the case of the dispersion-strength composite, the particles are extremely small in that they prevent the matrix from dislocating inducing plastic deformation and thereby increase the strength of the material (Feng et al, 2007). More practical applications of particulate composites can be found in Donald (1990).

Laminate composites consist of two or more different thin, planar and unidirectional layers called laminas. The laminas are stacked and subsequently bonded in such a way that the material properties and reinforcing directions vary with each successive layer. One particular application of laminate composites can be found in shape memory alloys (see Irie, 1989; Tupper, 2001; Zhang et al, 2007; Zhang. Y et al, 2007). Details of designing laminate composites can be found in John (1999).

Fiber-reinforced composites consist of cylindrical fibers (i.e. each cross section of the cylindrical fibers has the same planar geometry) and a surrounding bulk matrix. The reinforcement (fiber) can be embedded into the matrix material in a variety of orientations so that the composite meets each specific service requirement (see Figure 2).



Figure 2: Several Orientations of Fiber-Reinforced Composites: (a) Continuous Unidirectional Fibers, (b) Randomly Oriented Short Discontinuous Fibers, (c) Orthogonal Fibers, and (d) Multiple-Ply Fibers (From Askeland, 1990)

Applications of fiber-reinforced composites are found in many different areas including aerospace, construction, automobile industries, spacecraft structures, electronics and petrochemical (see Chung, 1984; Chen et al, 2007; Karthikeyan et al, 2007). The elastic fibers acting as reinforcements absorb the external load throughout the surrounding matrix and hence, increase the strength of the composite. The reinforcement should be a material that is stiff, whereas the surrounding matrix should be tough and sufficiently ductile to support and allow for external loads to be transmitted into each reinforcement. In addition, the surrounding matrix, serving as a barrier, protects the composite from the propagation of brittle cracks induced at the vicinity of the fibers.

Due to its wide range of application in modern industry, the fiber-reinforced composite has gained particular attention and become the most predominant composite in use today. For this reason, the fiber-reinforced composite will form the fundamental framework of this dissertation.

As mentioned above, the use of reinforcements certainly improves desirable mechanical properties such as stiffness, strength, and toughness. It is well-known, however, that the reinforcement also (referred to as an inhomogeneity) disturbs the original stress field leading to local stress intensification that may further induce the total failure of the composite system. Therefore, it is of great importance to study the state of stress within the composite material, in particular, how the distribution of stress varies with the introduction of inhomogeneities.

1.2 SINGLE INHOMOGENEITY MODEL

Since, in nature, a composite body contains numerous fibers (inhomogeneities), the actual analysis of such a composite is computationally expensive and often analytically impossible. Instead, we may imagine a small section sampled from a composite and then analyze it as if the sampled section is the original composite. The central vehicle, which makes this possible, is the existence of the representative volume element (RVE) (see Pindera et al,

1995) in which a sample of the composite material is taken to be structurally the same as the whole composite with the average properties indistinguishable from those of the statistically homogeneous composite. Therefore, the entire three-dimensional fiber-reinforced composite can be analyzed by considering a single inhomogeneity embedded in infinite medium called the "matrix" followed by an *averaging procedure*, for example, the Reuss and Voigt hypotheses, the Dilute Approximation, the Self-Consistent Scheme, the Mori-Tanaka Method, and the Eshelby Equivalent Inclusion method all of which lead to *effective properties* of the composite (see Hashin, 1983; Hill, 1965; Christenson & Lo, 1979; Christenson, 1990; Jacob, 1991; Nemat & Hori, 1993; Huang & Hu, 1995).

It is widely understood that the single-inhomogeneity problem is of fundamental importance in a composite (Eshelby, 1957, 1959; Hashin, 1991). More importantly, the single-inhomogeneity model is a much simpler and more practical model for analytical approaches than other available multi-inhomogeneity composite models (Achenbach & Zhu, 1990). Therefore, it is suggested that the RVE of the composite material can be treated as a single inhomogeneity-matrix system.

In our present study, we further consider the single inhomogeneity-matrix system undergoing plane-strain deformations in that the cross section of the three-dimensional composite is subjected to plane deformations (Timoshenko & Goodier, 1970; Truesell & Toupin, 1960; Malvern, 1969; Fung, 1977; Gurtin, 1981) for simplification. The planestrain deformation is possible when the dimension of the body in the z-direction is very large and the conditions are the same at all cross sections, so that it is sufficient to consider only a slice between two sections a unit distance apart. There are many important problems of this kind, for example, a fiber-reinforced composite (see Figure 1(b)) is stretched in the uniaxial or biaxial direction, a cylindrical tube with internal pressure and a retaining wall (e.g. dam) with lateral pressure.

For single-inhomogeneity problems most works so far, have been concerned with *linear plane elasticity* (Materials undergoing relatively small deformations see, for example, Timoshenko & Goodier, 1970; Ru & Schiavone, 1996, 1997; Ru, 1998a, 1998b; Schiavone, 2003; Van Vliet et al, 2003; Ru, et al, 2005). However, very little researches have been conducted in *nonlinear elasticity* which involves large or finite deformations (see C. Truesdell, 1965). The actual analysis of relevant problems has eluded researchers due to the existing nonlinearity in mathematical models. Yet, there is one special class of materials which mathematically behaves as a linear material. This class is the class of *harmonic* materials, first introduced by John (1960) and later explored by a number of authors see, for example, Knowles & Sternberg (1975), Varley & Cumberbatch (1980), Abeyaratne & Horgan (1984), Horgan (1989) and Li & Steigmann (1993). Therefore, in this dissertation, we will study the single inhomogeneity problem for finite deformations of a fiber-reinforced composite system in nonlinear plane elasticity using harmonic materials.

We also mention here that we consider only a well-spaced fiber-reinforced composite with fiber volume fractions up to 40% (Schmauder et al, 1992). In that way, the interaction among neighboring fibers and its influence on the stress fields on the entire composite system can be neglected so that we can isolated one fiber and study this in an infinite matrix. Finally, the effective property of the composite can be obtained by using one of the averaging procedures mentioned above.

1.3 INTERPHASE LAYER

The fiber-reinforced composite material consists of two distinctive phases, namely; a fiber (inhomogeneity) and a matrix phase. However, in nature, there is an additional phase existing between the fiber and the matrix called an " interphase or interphase layer". The interphase layer may be considered as the by-product of chemical reactions during the fabrication process or as a thin layer introduced in the design stage to improve the performance of the composite material (Horgan & Chan, 1999). Consequently, the mechanical properties of certain fiber-reinforced composite materials depend not only on the properties of the two constituent phases (fiber and matrix) but also on the nature of the bond between the fiber and the surrounding matrix (Jayaraman et al, 1993).

The existence of an interphase layer implies that the composite material is required to be considered as a three-phase constituent system (matrix phase, interphase layer and fiber). Nevertheless, due to its intrinsic complexity in forming, it is extremely difficult to provide an exact mathematical description of the behavior of the interphase layer. Extensive research has been done in the fields of micromechanics, chemistry and material sciences on the interphase layer (Brennan, 1988; Kim & Bau, 1992) and various approximation models have been proposed in an attempt to characterize the complex behavior of the interphase layer. For example, the imperfect interface model (Achenbach & Zhu, 1990; Ghahremani, 1980; Hashin, 1991) assumes that a very thin interfacial zone exists between the fiber and the matrix. In the physical sense, an 'interface' can be considered as a limiting case of the interphase layer as its thickness tends to zero. This interface, defined as a twodimensional boundary (a curved plane), distinguishes the bulk materials and, hence, the material properties are changed abruptly across the interface.

1.3.1 PERFECT BONDING

A state of perfect bonding assumes that the interface is considered as a surface across which both tractions and displacements are continuous. Such an interface is generally referred to as a "perfectly bonded interface" (see Ru &Schiavone, 1996; Ru et al, 2005). The perfect bonding assumption is an idealization for it is known that the bonding, in nature, is never perfect due to the existing cracks, voids and cavities in the interphase layer between a fiber and its surrounding matrix. In general, perfect bonding is rather conventional, yet comprehensive studies reveal that (Knowles & Sternberg, 1975; Ru, 2002) the model shows satisfactory physical correspondence in many cases. In addition, the existence of a prefect bond enables an idealization of very complex behavior such as finite plane deformations accompanying large deformations in nonlinear elasticity.

1.3.2 IMPERFECT BONDING

An imperfect bond, as the name suggests, was introduced to take into account the imperfections of the interface such as voids, cavities, cracks, and dislocations. The fundamental premise for the imperfect bonding model is that the interphase, in general, is found to be softer and weaker than both fiber and matrix (Jones & Whitter, 1967). Under this condition, the imperfect bonding can be modeled by introducing the interface across which tractions are continuous but displacements are discontinuous. A considerable amount of studies have been conducted to providing mathematical models effectively describing the behavior of the imperfectly bonded interphase. The simplest imperfect interface model is the elastic spring-type interface model in which the tractions are continuous yet, displacement jumps across the interface are directly proportional to their respective interface traction components (see, for example, Bigoni et al, 1998; Lipton & Serkove et al, 1998; steif & Hoysan, 1987; Aboudi, 1987; Gosz et al, 1992). We do not discuss this more complicated scenario in this thesis but instead note that this is an interesting area for future study especially in the area of finite deformations.

1.4 PURPOSE OF STUDY AND DISSERTATION OVERVIEW

The objectives of the present study are to consider the finite plane deformations of a harmonic composite material containing an elastic inhomogeneity embedded in a surrounding matrix of similar elastic material. The bonding at the inhomogeneity-matrix interface is considered to be perfect so that both tractions and displacements are continuous across the surface. For this purpose, complex variable techniques (Muskhelishvili, 1953; England, 1971; Brown & Churchill, 1996) are extensively incorporated in the current work.

The results derived in this dissertation will enable one to have a clear understanding of the state of stress throughout the entire fiber-reinforced composite system subjected to varying applied loadings. In particular, the complete solutions of the corresponding problems of these kinds (see Chapters. 3-5) provide physical relevance to those corresponding "real-world problems" and hence will enhance the future design of composite materials. The thesis proceeds as follows. Chapter 2 outlines the complex-variable formulations for a special class of harmonic materials. In addition, general preliminaries such as notations, interface formulations and the state of the undeformed stage are extensively discussed for further purpose.

Chapter 3 provides the complete solution for a perfectly bonded elliptic inhomogeneity when the system is subjected to classes of nonuniform remote stress distributions. We note, in particular, the generality of our results and that existing results in the literature are obtained as special cases of the solutions derived here.

Chapter 4 examines, in detail, the analogy of Eshelby's conjecture for the particular class of materials of harmonic-type undergoing finite plane deformations. In particular, we show that, under the analogous constraints of perfect bonding and uniform remote (Piola) stress, if an inhomogeneity matrix-system is subjected to finite plane deformations, the Piola stress distribution within an elliptic inhomogeneity is necessarily uniform.

Chapter 5 discusses the idea that by adjusting the remote (Piola) stress we can design the shape of the inhomogeneity in such a way that the interior (Piola) stress distribution remains uniform.

Finally, Chapter 6 provides a summary of results obtained and some concluding remarks with some suggestions for future research.

Chapter 2 GENERAL FORMULATION AND PRELIMINARIES

2.1 INTRODUCTION

As discussed in Chapter 1, we consider the single inhomogeneity problems in which the inhomogeneity is perfectly bonded to the surrounding material referred to henceforth as 'the matrix'. Assume that all materials are 'harmonic' (see below) and that the corresponding inhomogeneity-matrix system is subjected to large plane-strain deformations. In addition, we further assume that the inhomogeneity-matrix system undergoes plane-strain deformation. Consequently, the problem of the entire system in \mathbb{R}^3 can be reduced the equivalent problem concerning the cross section of the system with unit thickness in \mathbb{R}^2 .

The class of compressible *hyperelastic harmonic materials* or just *harmonic materials* als was proposed by John (1960) and has attracted considerable attention in the literature (see for example, Knowles & Sternberg, 1975; Varley & Cumberbatch, 1980; Abeyaratne, 1983; Abeyaratne & Horgan, 1984; Jafari, Abeyaratne & Horgan 1984; Li & Steigmann, 1993; Horgan, 1989, 1995, 2001). For example, experimental results by Varley & Cumberbatch (1980) indicate close agreement between this class of harmonic materials and the be-

havior of certain rubber-like materials. In addition, Li & Steigmann (1993) and Abeyaratne (1983) have used these harmonic materials to study, respectively, the finite deformations of an annular membrane induced by the rotation of a rigid hub and the finite deformations of a crack. Recently Wang et al (2005) have extended and analyzed the notion of 'harmonic shapes' (holes and inclusions) to this class of materials.

In the following section, we present the complex-variable formulation of a model of the behavior of harmonic materials.

2.2 HARMONIC MATERIALS

2.2.1 COMPLEX-VARIABLE FORMULATION

We consider a plane cross-section of the single inhomogeneity-matrix system in which it is assumed that the matrix is unbounded. Consequently, let the inhomogeneity-matrix system occupy \mathbb{R}^2 (described here by the generic coordinates (x_1, x_2)) in its undeformed (reference) configuration. The deformation describing the deformed (or current) configuration is given by:

$$w(z) = y_1(x_1, x_2) + iy_2(x_1, x_2), \text{ for all } (x_1, x_2) \in \mathbb{R}^2.$$
 (2.1)

The components of the deformation-gradient tensor are given by (see Knowles & Sternberg, 1975):

$$F_{ij} = \frac{\partial y_i}{\partial x_j} = y_{i,j}, \quad J = \det \mathbf{F} > 0 \quad , i, j = 1, 2$$
(2.2)

and let G be the left Cauchy -Green strain tensor defined by

$$\mathbf{G} = \mathbf{F}\mathbf{F}^T$$
 on \mathbb{R}^2 .

Then, from (2.1-2), we can calculate the two fundamental scalar invariants of G:

$$R = \lambda_1^2 + \lambda_2^2 = tr(\mathbf{G}) = tr[F_{ps}F_{qs}(e_p \otimes e_q)] = F_{ps}F_{ps} = y_{p,s}y_{p,s}, \ p, s = 1, 2(2.3)$$
$$J = \lambda_1\lambda_2 = \sqrt{\det \mathbf{G}} = F_{11}F_{22} - F_{12}F_{21} = y_{1,1}y_{2,2} - y_{1,2}y_{2,1},$$

where λ_1 and λ_2 are the principal stretches of G. From Knowles & Sternberg (1975), the governing constitutive relations and the equilibrium equation for the harmonic material are given by

$$\tau_{ij} = \frac{2}{J} \frac{\partial \Theta}{\partial R} F_{i\rho} F_{j\rho} + \frac{\partial \Theta}{\partial J} \delta_{ij},$$

$$\sigma_{ij} = 2 \frac{\partial \Theta}{\partial R} F_{ij} + \frac{\partial \Theta}{\partial J} \epsilon_{i\rho} \epsilon_{j\gamma} F_{\rho\gamma}$$

$$\frac{\partial \sigma_{ij}}{\partial x_j} = 0, \quad \text{on } \mathbb{R}^2, \quad i, j, \rho, \gamma = 1, 2, \tag{2.4}$$

$$W(x_1, x_2) = \Theta(R(x_1, x_2), J(x_1, x_2)), \text{ for all } (x_1, x_2) \in \mathbb{R}^2,$$

where W is the strain energy density of a homogeneous and isotropic (hyper-elastic) solid. It is obvious from (2.4) that the mechanical response of a homogeneous and isotropic solid to a plane deformation is dominated entirely by the material response defined by the function Θ .

The strain energy density W for this class of harmonic materials introduced by John (1960) is characterized per unit volume of the deformed configuration by

$$W = \Theta(I, J) = 2\mu[F(I) - J], \ I = \sqrt{R + 2J}.$$
(2.5)

where μ is a given positive material constant and F(I) a material function of I (note that F(I) is different from F). Then (2.1-3) and (2.5) lead to the representations

$$I = \sqrt{\lambda_1^2 + \lambda_2^2 + 2\lambda_1\lambda_2} = \lambda_1 + \lambda_2 = |w_{,2} + iw_{,1}|, \qquad (2.6)$$
$$J = \lambda_1\lambda_2 = \det \mathbf{F} = -\operatorname{Im}[w_{,1}\overline{w}_{,2}]$$
$$R = tr(\mathbf{F}\mathbf{F}^T) = |w_{,1}|^2 + |w_{,2}|^2.$$

From (2.5), the constitutive relations (2.4) become

$$\tau_{ij} = 2\mu \left[\frac{F'(I)}{IJ} F_{i\rho} F_{j\rho} + \left(\frac{F'(I)}{I} - 1 \right) \delta_{ij} \right], \qquad (2.7A)$$

$$\sigma_{ij} = 2\mu \left[\frac{F'(I)}{I} F_{ij} + \left(\frac{F'(I)}{I} - 1 \right) \epsilon_{i\rho} \epsilon_{j\gamma} F_{\rho\gamma} \right], \qquad (2.7B)$$

where

$$\frac{\partial \Theta}{\partial R} = \frac{\partial 2\mu [F\left(\sqrt{R+2J}\right) - J]}{\partial R} = 2\mu F'\left(\sqrt{R+2J}\right) \frac{\partial \sqrt{R+2J}}{\partial R} = \mu \frac{F'(I)}{I},$$

$$\frac{\partial \Theta}{\partial J} = \frac{\partial 2\mu [F\left(\sqrt{R+2J}\right) - J]}{\partial J} = 2\mu \left[F'\left(\sqrt{R+2J}\right)\frac{\partial \sqrt{R+2J}}{\partial J} - 1\right] = 2\mu \left[\frac{F'(I)}{I} - 1\right]$$

Equating (2.7B) for

$$i, j, \rho, \gamma = 1, 2, \epsilon_{11} = \epsilon_{22} = 0, \epsilon_{12} = 1, \epsilon_{21} = -1,$$

we obtain that

$$\sigma_{11} = 2\mu \left(\frac{F'(I)}{I} \left(F_{11} + F_{22} \right) - F_{22} \right), \quad \sigma_{22} = 2\mu \left(\frac{F'(I)}{I} \left(F_{11} + F_{22} \right) - F_{11} \right),$$

$$\sigma_{12} = 2\mu \left(\frac{F'(I)}{I} \left(F_{12} - F_{21} \right) + F_{21} \right), \quad \sigma_{21} = 2\mu \left(-\frac{F'(I)}{I} \left(F_{12} - F_{21} \right) + F_{12} \right),$$
(2.8)

From (2.1) in conjunction with (2.2), we derive the following relation

$$\frac{F'(I)}{I}(F_{12} - F_{21}) + i\frac{F'(I)}{I}(F_{11} + F_{22}) = \frac{F'(I)}{I}(w_{,2} + iw_{,1}), \qquad (2.9)$$
$$(\because w_{,2} = F_{12} + iF_{22}, \quad w_{,1} = F_{11} + iF_{21}).$$

and from (2.7A) we have that for, $i, j, \rho = 1, 2$,

$$\tau_{11} = 2\mu \left[\frac{F'(I)}{IJ} \left(F_{11}^2 + F_{12}^2 \right) + \left(\frac{F'(I)}{I} - 1 \right) \right],$$

$$\tau_{22} = 2\mu \left[\frac{F'(I)}{IJ} \left(F_{22}^2 + F_{21}^2 \right) + \left(\frac{F'(I)}{I} - 1 \right) \right],$$
 (2.10)

$$\tau_{12} = 2\mu \frac{F'(I)}{IJ} \left(F_{12}F_{22} + F_{11}F_{21} \right), \quad (\because \delta_{11} = \delta_{22} = 1, \ \delta_{12} = 0),$$

Therefore, from (2.1-2) and (2.9), Equation (2.8) can be written the form

$$\sigma_{12} + i\sigma_{22} = 2\mu \left[\frac{F'(I)}{I} \left(w_{,2} + iw_{,1} \right) - iw_{,1} \right],$$

$$\sigma_{11} + i\sigma_{21} = 2\mu i \left[w_{,2} - \frac{F'(I)}{I} \left(w_{,2} + iw_{,1} \right) \right], \qquad (2.11)$$

Also, from(2.1-3), Equation (2.10) leads to the representations

$$\tau_{11} + \tau_{22} = 2\mu \left[\frac{IF'(I)}{J} - 2 \right],$$

$$\tau_{11} - \tau_{22} + 2i\tau_{12} = 2\mu \frac{F'(I)}{IJ} \left(w_{,1}^2 + w_{,2}^2 \right), \qquad (2.12)$$

where

$$R = F_{ij}F_{ij} = F_{11}^2 + F_{12}^2 + F_{22}^2 + F_{21}^2, \quad i, j = 1, 2,$$

$$w_{,1}^2 + w_{,2}^2 = F_{11}^2 - F_{22}^2 + F_{12}^2 - F_{21}^2 + 2i(F_{12}F_{22} + F_{11}F_{21}).$$

Here, σ and τ in (2.11-12) are referred to as the Piola stresses and the Cauchy stresses, respectively. The Piola stress is the measurement of stress per unit area of in the undeformed (reference) configuration and the Cauchy stress is measured per unit area of the deformed (current) configuration (see Chadwick, 1999; Fung, 1977; Gurtin, 1981).

Finally, in view of (2.2), (2.8) and the plane deformation (2.1), the equilibrium equation from the third equation of (2.4) now takes the form

$$\left[\frac{F'(I)}{I}(w_{,2}+iw_{,1})\right]_{,1}+i\left[\frac{F'(I)}{I}(w_{,2}+iw_{,1})\right]_{,2}=0$$
(2.13)

To deal with Equations (2.11-13), we introduce a 'stress function' (or potential) ϕ analytic with respect to the complex variable z (see Ru, 2002):

$$\frac{F'(I)}{I}(w_{,2} + iw_{,1}) = \phi'(z), \quad F'(I) = |\phi'(z)|, \quad w'(z) = [\phi'(z)]^{\delta+1}, \quad (2.14)$$

$$I = P(|\phi'|), \quad P(^*) \equiv [F'(^*)]^{-1},$$

$$z = x_1 + ix_2, \quad \text{for all } (x_1, x_2) \in \mathbb{R}^2,$$

where P(*) denotes the inverse function of F'(*). For harmonic materials, the uniaxial Piola stress is given by (see Ru, 2002)

$$T = 2\mu \left[F'(I) - \lambda_2\right], \qquad (2.15)$$
$$F'(I) = \lambda_1, \quad \lambda_2 = P(\lambda_1) - \lambda_1,$$

where λ_2 is the transverse stretch of the material. Experimental results concerning Equation (2.15) reveal that the uniaxial relations can be written in the form (see Varley & Cumberbatch, 1980)

$$T = 4\mu \frac{b\lambda_1^k + c}{\lambda_1}, \quad k = 2(1+\delta).$$
 (2.16)

We note from Varley & Cumberbatch (1980) that this uniaxial relation (2.16) is designed to accommodate the undeformed state $\lambda_1 = \lambda_2 = 1$ at which T = 0. In addition, the results based on the model (2.16) were found to be in good agreement with experimental data, for example, those obtained for the large deformations of a rubber sheet containing an elliptic hole (Varley & Cumberbatch, 1980).

It is shown in Ru (2002) that we can write the deformation w(z) in terms of two complex potentials $\phi(z)$, $\psi(z)$ as

$$w(z) = -i\phi(z) + i\frac{bw(z)\overline{w'(z)} + cz}{\overline{\phi'(z)}} + \overline{\psi(z)},$$
(2.17)

and from (2.11), (2.14) and (2.18) the Piola stresses in the form:

$$-\sigma_{21} + i\sigma_{11} = \chi_{,2}, \quad \sigma_{22} - i\sigma_{12} = \chi_{,1}, \tag{2.18}$$

where the stress function χ is defined by

$$\chi(z) \equiv \chi_1(z) + i\chi_2(z) \equiv -2\mu [i\phi(z) + w(z)].$$
(2.19)

From (2.17) and (2.19), χ can be written as:

$$\chi(z) = -2\mu \left[i \frac{bw(z)\overline{w'(z)} + cz}{\overline{\phi'(z)}} + \overline{\psi(z)} \right].$$
(2.20)

The stress function $\chi(z)$ in Equation (2.20) is referred to as the Piola stress function.

2.2.2 A PARTICULAR CLASS OF HARMONIC MATERIAL

According to Varley & Cumberbatch (1980), the value k = 2.28 (in Equation (2.16)) provides the best agreement with experimental results obtained from some rubber-like materials. In addition, due to the mathematical simplicity, the special class of harmonic materials (2.16) defined by k = 2 have gained particular attention (Horgan, 1989, 1995; Li & Steigmann, 1993). From (2.14) and (2.16), this special class of harmonic materials is characterized by

$$F'(I) = \frac{1}{4\alpha} \left[I + \sqrt{I^2 - 16\alpha\beta} \right], \quad P(\lambda) = 2\left(\alpha\lambda + \frac{\beta}{\lambda}\right) \ge 4\sqrt{\alpha\beta},$$

$$T(\lambda_1) = 4\mu \left[(1-\alpha)\lambda_1 - \frac{\beta}{\lambda_1} \right], \quad 1 > \alpha \ge \frac{1}{2}, \quad \beta > 0,$$
(2.21)

where α and β are two material constants, related to b and c by $\alpha = 1-b$ and $\beta = -c$. Here, to get a positive Piola stress and transverse stretch at very large stretching, we have that $1 > \alpha \ge \frac{1}{2}$. Similarly, to get a negative Piola stress at very large compression, we have $\beta > 0$ (Varley & Cumberbatch, 1980). To incorporate the undeformed stage, from (2.21) we require that, for $\lambda_1 = 1, T = 0$. That is,

$$T(1) = 4\mu \left[(1-\alpha) 1 - \frac{\beta}{1} \right] = 0,$$

which gives (Ru, 2002)

$$\alpha + \beta = 1. \tag{2.22}$$

For example the values, $\alpha = \beta = \frac{1}{2}$ have been adopted by Li & Steigmann (1993) to study the finite deformation of an annular membrane induced by the rotation of a rigid body, and by Abeyaratne (1983) to study the finite deformation of a crack. In addition, when k = 2, it is noted from (2.14) that

$$w'(z) = \phi'(z), \quad \delta = 0$$

 $(\because k = 2(1 + \delta) = 2),$

and (2.17) and (2.20) become

$$iw(z) = \alpha \phi(z) + i\overline{\psi(z)} + \frac{\beta z}{\overline{\phi'(z)}}$$
(2.23)

$$\chi(z) = 2\mu i [(\alpha - 1)\phi(z) + i\overline{\psi(z)} + \frac{\beta z}{\overline{\phi'(z)}}], \qquad (2.24)$$

which gives the two fundamental (to this study) quantities w and χ entirely in terms of two analytic functions ϕ and ψ .

2.3 PREREQUISITES

2.3.1 NOTATION AND INTERFACE FORMULATION

Consider a domain in \mathbb{R}^2 , infinite in extent, containing a single internal elastic inhomogeneity with elastic properties different from those of the surrounding matrix. We represent the matrix by the domain S_1 and assume that the inhomogeneity occupies the region S_2 . The curve Γ will denote the inhomogeneity-matrix interface. In what follows, the subscripts 1 and 2 will refer to the regions S_1 and S_2 , respectively. The elastic materials occupying the matrix and the inhomogeneity belong to the class of harmonic materials characterized by Equation (2.5) and (2.21) above with associated elastic constants μ_1 , α_1 , β_1 and μ_2 , α_2 , β_2 , respectively. From the previous section, the (plane) deformation function w in the matrix and in the inhomogeneity can be written, respectively, in terms of two analytic functions ϕ and ψ as:

$$iw_{\gamma}(z) = \alpha_{\gamma}\phi_{\gamma}(z) + i\overline{\psi_{\gamma}(z)} + \frac{\beta_{\gamma}z}{\overline{\phi_{\gamma}'(z)}}$$
 $\gamma = 1,2$ (no sum over repeated indices). (2.25)

Similarly, the complex Piola stress function χ can be written in terms of φ and ψ in the matrix and in the inhomogeneity as:

$$\chi_{\gamma}(z) = 2\mu_{\gamma}i[(\alpha_{\gamma} - 1)\phi_{\gamma}(z) + i\overline{\psi_{\gamma}(z)} + \frac{\beta_{\gamma}z}{\phi_{\gamma}'(z)}] \quad \gamma = 1,2 \text{ (no sum over repeated indices)}.$$
(2.26)

Since, in the present study, we adopt the perfectly bonded interface model that both tractions and displacements are continuous across the interface, the continuity conditions for the displacement and Piola stresses across the (perfectly) bonded interface Γ lead to

$$\alpha_1\phi_1(z) + i\overline{\psi_1(z)} + \frac{\beta_1 z}{\overline{\phi_1'(z)}} = \alpha_2\phi_2(z) + i\overline{\psi_2(z)} + \frac{\beta_2 z}{\overline{\phi_2'(z)}}, \qquad (2.27)$$

$$\mu_{1}[(\alpha_{1}-1)\phi_{1}(z)+i\overline{\psi_{1}(z)}+\frac{\beta_{1}z}{\overline{\phi_{1}'(z)}}] = \mu_{2}[(\alpha_{2}-1)\phi_{2}(z)+i\overline{\psi_{2}(z)}+\frac{\beta_{2}z}{\overline{\phi_{2}'(z)}}], \quad (2.28)$$
respectively, where $z \in \Gamma$.

2.3.2 CHARACTERIZING REMOTE LOADING AND THE UNDEFORMED STAGE

Earlier, we assumed that the matrix was unbounded. Consequently, it is necessary to impose some asymptotic conditions on the inhomogeneity-matrix system. These conditions will, in reality, represent the remote loading applied to the system when subjected to planestrain deformations. For example, in Ru et al (2005), the inhomogeneity-matrix system has been studied exclusively in the case of uniform remote loading. This scenario is not sufficiently general to accommodate the many different forms of remote loading to which such a system can be subjected. Consequently, we attempt to analyze the inhomogeneity-matrix system with as general as class of remote loadings as possible.

To consider an elliptic inhomogeneity-matrix system subjected to a class of remote stresses, it is necessary to first identify what possible forms of remote stresses can be imposed. Also, as mentioned before, it is required to ensure that the complex-variable formulations (2.21) and (2.23-24) do in fact admit the undeformed stage when all external and internal forces are removed.

From (2.18) together with (2.26), the Piola stresses in the matrix and in the inhomogeneity can be written, respectively, in terms of the two analytic functions ϕ and ψ as:

$$-\sigma_{21} + i\sigma_{11} = 2\mu_{2}i \left[(\alpha_{2} - 1) (\phi_{2}(z))_{,2} + i\overline{(\psi_{2}(z))_{,2}} + \left(\frac{\beta_{2}z}{\overline{\phi_{2}'(z)}}\right)_{,2} \right] ,$$

$$\sigma_{22} - i\sigma_{12} = 2\mu_{2}i \left[(\alpha_{2} - 1) (\phi_{2}(z))_{,1} + i\overline{(\psi_{2}(z))_{,1}} + \left(\frac{\beta_{2}z}{\overline{\phi_{2}'(z)}}\right)_{,1} \right] ,$$

$$-\sigma_{21}^{\infty} + i\sigma_{11}^{\infty} = 2\mu_{1}i \left[(\alpha_{1} - 1) (\phi_{1}(z))_{,2} + i\overline{(\psi_{1}(z))_{,2}} + \left(\frac{\beta_{1}z}{\overline{\phi_{1}'(z)}}\right)_{,2} \right] , \quad (2.29)$$

$$\sigma_{22}^{\infty} - i\sigma_{12}^{\infty} = 2\mu_{1}i \left[(\alpha_{1} - 1) (\phi_{1}(z))_{,1} + i\overline{(\psi_{1}(z))_{,1}} + \left(\frac{\beta_{1}z}{\overline{\phi_{1}'(z)}}\right)_{,1} \right] .$$

Since the Piola stresses from (2.29) are functions of $\phi_1(z)$ and $\psi_1(z)$ and in the view of (2.25), $\psi_1(z)$ is related to $\phi_1(z)$ through $w_1(z)$, it is obvious that the complex function $\phi_1(z)$ dominates the remote stresses. It follows from (2.14) that the equilibrium equation

(2.13) can be rewritten as:

$$\phi_1'(z)_{,1} + i\phi_1'(z)_{,2} = 0.$$
(2.30)

Therefore, the complex function $\phi(z)$ must be chosen in such a way to satisfy the equilibrium equation (2.30). One of the successful candidates for the above equation is a monomial representation of z (e. g. z^n). In fact,

Remark 1. Using the principal of superposition, we find that any series representation of $\phi(z)$ (i.e. $\phi(z) = A_1 z + A_2 z^2 + A_3 z^3 + \cdots, A_{\gamma} \in \mathbb{C}$, $(\gamma = 1, 2, 3, \cdots)$) satisfies the equilibrium equation,

$$\phi'(z)_{,1} + i\phi'(z)_{,2} = 0,$$

and from (2.14-15) and Knowles & Sternberg (1975),

 $\lambda_1 = |\phi'(z)| = F'(I) \neq 0$ throughout the entire plane.

The result in Remark 1 is extremely important in that it allows us to accommodate a large class of possible remote stresses, This statement is based on the well-known result from the theory of functions (Weierstrass) which states that any continuous function in a bounded domain can be uniformly approximated by a polynomial.

From Remark 1 (or (2.14-15)), we must have that, for the undeformed stage characterized by $\lambda_1 = \lambda_2 = 1$,

$$1 = \lambda_1 = F'(I) = |\phi'(z)|.$$
(2.31)

Also, from the deformation function (2.25), we require that, for the undeformed stage,

$$w(z) = -i\alpha\phi(z) + \overline{\psi(z)} - \frac{i\beta z}{\overline{\phi'(z)}} = z.$$
(2.32)

For the class of harmonic materials considered here, to ensure the desired mechanical behavior (i.e. to ensure that the undeformed stage is included), it is required from (2.21-22) that (Ru, 2002)

$$\alpha + \beta = 1. \tag{2.33}$$

Thus, it follows from (2.31-33) that the stress functions characterizing the undeformed stage are identified as:

$$\phi(z) = iz, \quad \psi(z) = 0.$$
 (2.34)

Remark 2. We remark that, for the class of harmonic materials considered here, a state of zero displacement or equivalently, the undeformed stage is characterized by

$$1 = \left|\phi_{\gamma}'\left(z\right)\right| = F'\left(I\right),$$

 $iw_{\gamma}(z) = \alpha_{\gamma}\phi_{\gamma}(z) + i\overline{\psi_{\gamma}(z)} + \frac{\beta_{\gamma}z}{\overline{\phi'_{\gamma}(z)}} = iz, \quad \gamma = 1,2$ (no sum over repeated indices),

which, in turn leads to

$$\phi_1(z) = iz, \quad \psi_1(z) = 0,$$

 $\phi_2(z) = iz, \quad \psi_2(z) = 0.$

2.3.3 SUBSIDIARY RESULTS

As stated before, it is of fundamental importance to fully understand the undeformed stage in an attempt to analyze an inhomogeneity-matrix system subjected to nonuniform remote stresses. In the present section, we provide physical interpretations of the undeformed stage

in terms of mathematical representations derived in the previous section. We first note from Remark 2 that, for the undeformed stage,

$$\phi_1(z) = iz, \quad \psi_1(z) = 0,$$

 $\phi_2(z) = iz, \quad \psi_2(z) = 0,$

(2.35)

 $\lambda_1 = \lambda_2 = 1.$

From (2.14) and (2.21) we now have that

$$I = P(|\phi'(z)|) = 2\left(\alpha |\phi'(z)| + \frac{\beta}{|\phi'(z)|}\right).$$
 (2.36)

Then, the undeformed stage condition yields only that

$$I = P(|i|) = 2(\alpha + \beta) = 2,$$

where

$$1 = \lambda_1 = F'(I) = |\phi'(z)| = |i|, \quad \alpha + \beta = 1, \quad \text{(from 2.33)}.$$

This result can be easily verified directly from (2.14-15) that

$$I = P\left(|i|\right) = \lambda_1 + \lambda_2 = 2.$$

The Piola stresses inside the inhomogeneity and at the remote boundary for the undeformed stage can then be evaluated from (2.29), (2.33) and (2.35). That is

$$-\sigma_{21} + i\sigma_{11} = 2\mu_2 i \left[(\alpha_2 - 1) i z_{,2} + \beta_2 \left(\frac{1}{-i} \right) z_{,2} \right] = 2\mu_2 i \left[1 - (\alpha_2 + \beta_2) \right] = 0,$$

$$\sigma_{22} - i\sigma_{12} = 2\mu_2 i \left[(\alpha_2 - 1) i z_{,1} + \beta_2 \left(\frac{1}{-i} \right) z_{,1} \right] = 2\mu_2 i \left[i \left(\alpha_2 + \beta_2 - 1 \right) \right] = 0,$$

$$-\sigma_{21}^{\infty} + i\sigma_{11}^{\infty} = 2\mu_1 i \left[(\alpha_1 - 1) i z_{,2} + \beta_1 \left(\frac{1}{-i} \right) z_{,2} \right] = 2\mu_1 i \left[1 - (\alpha_1 + \beta_1) \right] = 0,$$

$$\sigma_{22}^{\infty} - i\sigma_{12}^{\infty} = 2\mu_1 i \left[(\alpha_1 - 1) i z_{,1} + \beta_1 \left(\frac{1}{-i} \right) z_{,1} \right] = 2\mu_1 i \left[i \left(\alpha_1 + \beta_1 - 1 \right) \right] = 0,$$

where, $\alpha_\gamma+\beta_\gamma~=~1~\gamma=1,2~$ (no sum over repeated indices),

(i.e. $\sigma_{11} = \sigma_{22} = \sigma_{12} = \sigma_{21} = \sigma_{11}^{\infty} = \sigma_{22}^{\infty} = \sigma_{21}^{\infty} = \sigma_{12}^{\infty} = 0$). For the deformation function, again, from (2.25), we have that

$$w_{\gamma}(z) = -i\frac{1}{2}iz - \frac{1}{2}\frac{iz}{-i} = z, \quad \gamma = 1,2$$
 (no sum over repeated indices),

which implies that the material points in the reference configuration remain intact. Since we assume that there is no residual stress, the results obtained above are found to be physically correct in that if there is no load applied to the system, then, the corresponding Piola stresses and displacements are equal to zero.

Finally, (2.30) (or (Remark 1)) and (2.35) yield for the undeformed stage:

 $\phi_{\gamma}^{\prime}\left(z
ight)_{,1}+i\phi_{\gamma}^{\prime}\left(z
ight)_{,2}=0, \hspace{0.2cm} \gamma=1,2 \hspace{0.2cm}$ (no sum over repeated indices),

which clearly, through (2.14), shows that the equilibrium equation (2.13) is satisfied.
Chapter 3 COMPLETE SOLUTION OF AN ELLIPTIC INHOMOGENEITY SUBJECTED TO A GENERAL CLASS OF NONUNIFORM REMOTE LOADINGS IN FINITE ELASTICITY

3.1 INTRODUCTION

The mathematical analysis of problems involving plane-strain deformations of harmonic materials has recently been facilitated by the relatively simple complex-variable formulation presented in Ru (2002). For example, Wang et al (2005) have extended and analyzed the notion of 'harmonic shapes' (holes and inclusions) to this class of materials. In addition, in Ru et al (2005), the authors have used this formulation to study an important class of introductory problems involving elastic inhomogeneities embedded in harmonic materials. More precisely, the authors have obtained complete solutions of problems concerning the plane deformations of an harmonic composite material containing an elastic elliptic inhomogeneity perfectly bonded to a surrounding matrix of similar elastic material. The results presented in Ru et al (2005), however, are limited to the case where the inhomogeneity-matrix system is subjected exclusively to *uniform* remote loading. In many practical problems, however, composite materials, in particular those subjected to finite deformations, are subjected to many different forms of loading most of which are nonuniform in nature. The analysis required to study this more general class of problems, however, is extremely challenging, mainly as a result of the complications arising from the use of conformal mapping techniques with the ensuing nonlinearities in the mapped plane.

In the present Chapter, we overcome the above-mentioned difficulties and address the issues relating to nonuniform remote loading of harmonic composite materials. In particular, we extend the results in Ru et al (2005) to the case of plane finite deformations of a composite material in which an elliptic elastic inhomogeneity is embedded in the same class of harmonic materials under the assumption of nonuniform remote loading (Kim & Schiavone, 2007a). Using complex-variable methods (Muskhelishvili, 1953; England, 1971; Brown & Churchill, 1996), we obtain the complete solution for a perfectly bonded elliptic inhomogeneity when the system is subjected to classes of nonuniform remote stress distributions characterized by stress functions described by general polynomials of degree $n \ge 1$ in the corresponding complex variable z describing the matrix. The results obtained are extremely important in that, essentially, they lead to the solution of a class of problems in which the remote loading is characterized by a much wider and more practically realistic class of functions.

3.2 NOTATION AND PREREQUISITES

Consider a domain in \mathbb{R}^2 , infinite in extent, containing a single internal elliptic elastic inhomogeneity with elastic properties different from those of the surrounding matrix. We represent the matrix by the domain S_1 and assume that the inhomogeneity occupies the region S_2 . The ellipse Γ will denote the inhomogeneity-matrix interface. In what follows, the subscripts 1 and 2 will refer to the regions S_1 and S_2 , respectively. The elastic materials occupying the matrix and the inhomogeneity belong to the class of harmonic materials characterized by Equation (2.5) and (2.21) above with associated elastic constants μ_1, α_1, β_1 and μ_2, α_2, β_2 , respectively.

In the view of Remarks. 1 and 2, we consider problems corresponding to cases in which the inhomogeneity-matrix system is subjected to a general class of remote loadings described by analytic functions ϕ_1 and ψ_1 of the (polynomial) form:

$$\phi_1(z) = A_1 z + \sum_{n=a}^{b} A_n z^n, \psi_1(z) = B_1 z + \sum_{n=a}^{b} B_n z^n, \ |z| \to \infty, a \ge 2, b \ge a, (3.1)$$

where A_1, A_n, B_1 and B_n are known complex constants.

3.3 COMPLETE SOLUTION FOR THE ELLIPTIC INHOMOGENEITY

We proceed by first using the mapping function $z = \omega(\xi)$ to map (conformally) the infinite region outside an ellipse in the z - plane to the infinite region outside the unit circle in the

 ξ – plane. In fact (see Muskhelishvili, 1953)

$$z = \omega\left(\xi\right) = R\left(\xi + \frac{m}{\xi}\right), \quad R > 0, \ 1 \ge m \ge 0, \tag{3.2}$$

with $\xi = \rho e^{i\theta}$ and $\rho = 1$ will map the ellipse

$$x_1 = R\left(\rho + \frac{m}{\rho}\right)\cos\theta, \quad x_2 = R\left(\rho - \frac{m}{\rho}\right)\sin\theta,$$
 (3.3)

(where $R \in \mathbb{R}$ and $\theta \in [0, 2\pi]$) and its exterior region in the complex z - plane, onto and outside, respectively, the unit circle in the complex $\xi - plane$.



Figure 3: The Conformal Mapping from z - plane to $\xi - plane$

In our present case of an elliptic inhomogeneity, we assume that $1 > m \ge 0$ (for m = 1, the ellipse becomes a straight slit; for m = 0, a circle). Using this mapping function and the binomial theorem, we write z^n in the ξ - plane as

$$z^{n} = \left(\omega\left(\xi\right)\right)^{n} = \left(R\left(\xi + \frac{m}{\xi}\right)\right)^{n} = R^{n} \sum_{r=0}^{n} \binom{n}{r} \xi^{n-2r} m^{r}.$$
 (3.4)

It is clear from (3.4) that each ocurrence of z^n in the governing equations will correspond to n + 1 terms in the $\xi - plane$. Consequently, we suggest the following representation of the corresponding functions ϕ_2 and ψ_2 inside the inhomogeneity:

$$\phi_{2}(z) = C_{1}z + \sum_{n=a}^{b} \left[C_{n}z^{n} + m \sum_{\alpha=0}^{\frac{n-(2+\delta_{o})}{2}} C_{n,\alpha}z^{2\alpha+\delta_{o}} \right]$$

$$\psi_{2}(z) = D_{1}z + \sum_{n=a}^{b} \left[D_{n}z^{n} + m \sum_{\alpha=0}^{\frac{n-(2+\delta_{o})}{2}} D_{n,\alpha}z^{2\alpha+\delta_{o}} \right], \quad z \in S_{2}, \ a \ge 2, b \ge a,$$
(3.5)

where $C_{1,C_{n}}, C_{n,\alpha}, D_{1}, D_{n}$ and $D_{n,\alpha}$ are complex constants (the 'comma' notation used to denote the constants $C_{n,\alpha}$ and $D_{n,\alpha}$ is chosen only for convenience and does not denote differentiation) to be determined and δ_{o} is defined as:

$$\delta_o = \begin{cases} 1, & n \text{ odd,} \\ 0, & n \text{ even.} \end{cases}$$

From (2.27-28), the interface conditions can now be written in the form

$$\phi_1(z) = K\phi_2(z) + iS\overline{\psi_2(z)} + \frac{S\beta_2 z}{\phi_2'(z)},$$
(3.6)

$$\psi_{1}(z) = i\left(-\alpha_{1}\overline{\phi_{1}(z)} - \frac{\beta_{1}\overline{z}}{\phi_{1}'(z)} + \alpha_{2}\overline{\phi_{2}(z)} + \frac{\beta_{2}\overline{z}}{\phi_{2}'(z)}\right) + \psi_{2}(z), \quad |z| = \Gamma.$$

For convenience, we write $\phi_{\gamma}(z) = \phi_{\gamma}(w(\xi)) = \phi_{\gamma}(\xi)$, $\gamma = 1, 2$ and similarly for the functions ψ_{γ} . Then, the interface condition (3.6) becomes in the $\xi - plane$:

$$\phi_1(\xi) = K\phi_2(\xi) + iS\overline{\psi_2(\xi)} + \frac{S\beta_2\omega(\xi)\omega'(\xi)}{\overline{\phi_2'(\xi)}}, \qquad (3.7)$$

$$\psi_1\left(\xi\right) = i\left(-\alpha_1\overline{\phi_1\left(\xi\right)} - \frac{\beta_1\overline{\omega\left(\xi\right)}\omega'\left(\xi\right)}{\phi_1'\left(\xi\right)} + \alpha_2\overline{\phi_2\left(\xi\right)} + \frac{\beta_2\overline{\omega\left(\xi\right)}\omega'\left(\xi\right)}{\phi_2'\left(\xi\right)}\right) + \psi_2\left(\xi\right), \quad |\xi| = 1,$$

$$\therefore \quad \phi_{\gamma}'(z) = \frac{d\phi(z)}{dz} = \frac{d\phi(\xi)}{d\xi} \frac{d\xi}{dz} = \frac{\phi'(\xi)}{\omega'(\xi)}, \quad \overline{\phi_{\gamma}'(z)} = \overline{\left(\frac{\phi'(\xi)}{\omega'(\xi)}\right)} = \frac{\overline{\phi'(\xi)}}{\overline{\omega'(\xi)}},$$

where the constants K and S are defined by $K = \frac{\mu_2}{\mu_1} + \alpha_2 \left(1 - \frac{\mu_2}{\mu_1}\right), \quad S = \left(1 - \frac{\mu_2}{\mu_1}\right).$

Remark 3. Since, from Remark 1

$$|\phi'_{\gamma}(z)| = F'(I) \neq 0, \quad A_1, \ C_1 \neq 0,$$

 $\gamma = 1,2$ (no sum over repeated indices),

and

$$\omega'(\xi) = R\left(1 - \frac{m}{\xi^2}\right) \neq 0, \quad |\xi| \ge 1, \ (\because m \neq 1),$$

it is clear that we must have

$$\phi_{2}'(\xi) \neq 0$$
$$\iff \overline{\phi_{2}'(\xi)} \neq 0, \quad |\xi| = 1$$

In other words, from (3.4-5), we must have that

$$C_{1} + \sum_{n=a}^{b} \left[nC_{n}R^{n-1} \left(\xi + \frac{m}{\xi}\right)^{n-1} + m \sum_{\alpha=0}^{\frac{n-(2+\delta_{o})}{2}} \left[(2\alpha + \delta_{0}) C_{n,\alpha}R^{2\alpha + \delta_{o}-1} \left(\xi + \frac{m}{\xi}\right)^{2\alpha + \delta_{o}-1} \right] \neq 0$$

$$\overline{C_1} + \sum_{n=a}^{b} \left[n \overline{C_n} R^{n-1} \left(\overline{\xi} + \frac{m}{\overline{\xi}} \right)^{n-1} + m \sum_{\alpha=0}^{\frac{n-(2+\delta_o)}{2}} \left[(2\alpha + \delta_0) \overline{C_{n,\alpha}} R^{2\alpha + \delta_o - 1} \left(\overline{\xi} + \frac{m}{\overline{\xi}} \right)^{2\alpha + \delta_o - 1} \right] \right] \neq 0.$$
for $|\xi| = 1.$

Since $\xi \overline{\xi} = |\xi|^2 = 1$, on the circular boundary in the ξ – *plane*, it follows from (3.2), (3.5) and (3.7) that the interface condition in the ξ – *plane* becomes:

 \Leftrightarrow

$$\begin{split} \phi_{1}\left(\xi\right) &= K \left[C_{1}R\left(\xi + \frac{m}{\xi}\right) + \sum_{n=a}^{b} \left[C_{n}R^{n}\left(\xi + \frac{m}{\xi}\right)^{n} + m \sum_{\alpha=0}^{\frac{n-(2+\delta_{0})}{2}} \left[C_{n,\alpha}R^{2\alpha+\delta_{0}}\left(\xi + \frac{m}{\xi}\right)^{2\alpha+\delta_{0}} \right] \right] \right] \\ &+ iS \left[\overline{D_{1}}R\left(\frac{1}{\xi} + \xi m\right) + \sum_{n=a}^{b} \left[\overline{D_{n}}R^{n}\left(\frac{1}{\xi} + m\xi\right)^{n} + m \sum_{\alpha=0}^{\frac{n-(2+\delta_{0})}{2}} \left[\overline{C_{n,\alpha}}R^{2\alpha+\delta_{0}}\left(\frac{1}{\xi} + m\xi\right)^{2\alpha+\delta_{0}} \right] \right] \right] \\ &+ \frac{S\beta_{2}R\left(\xi + \frac{m}{\xi}\right)}{\overline{C_{1}} + \sum_{n=a}^{b} \left[n\overline{C_{n}}R^{n-1}\left(\frac{1}{\xi} + \xi m\right)^{n-1} + m\sum_{\alpha=0}^{\frac{n-(2+\delta_{0})}{2}} \left[(2\alpha+\delta_{0})\overline{C_{n,\alpha}}R^{2\alpha+\delta_{0}-1}\left(\frac{1}{\xi} + \xi m\right)^{2\alpha+\delta_{0}-1} \right] \right] \end{split}$$

$$(3.8)$$

for $|\xi| = 1$.

From (3.4) together with (3.8), we finally obtain , for $|\xi| = 1$,

$$\begin{split} \phi_1\left(\xi\right) &= KC_1 R\left(\xi + \frac{m}{\xi}\right) + iS\overline{D_1} R\left(\frac{1}{\xi} + \xi m\right) + \sum_{n=a}^b \left[R^n \sum_{r=0}^n \binom{n}{r} \xi^{n-2r} \left(KC_n m^r + iS\overline{D_n} m^{n-r} \right) \right] \\ &+ \sum_{n=a}^b \left[\sum_{\alpha=0}^{\frac{n-(2+\delta_o)}{2}} \left[R^{2\alpha+\delta_o} \sum_{r=0}^{2\alpha+\delta_o} \binom{2\alpha+\delta_o}{r} \xi^{2(\alpha-r)+\delta_o} \left(KC_{n,\alpha} m^{r+1} + iS\overline{D_{n,\alpha}} m^{2\alpha+\delta_o-r+1} \right) \right] \right] + \end{split}$$

$$\frac{S\beta_2 R\left(\xi+\frac{m}{\xi}\right)}{\overline{C_1}+\sum_{n=a}^{b}\left[n\overline{C_n}R^{n-1}\left(\frac{1}{\xi}+\xi m\right)^{n-1}+m\sum_{\alpha=0}^{\frac{n-(2+\delta_o)}{2}}\left[(2\alpha+\delta_0)\overline{C_{n,\alpha}}R^{2\alpha+\delta_o-1}\left(\frac{1}{\xi}+\xi m\right)^{2\alpha+\delta_o-1}\right]\right]},$$
(3.9)

where

$$R^n\left(\xi+\frac{m}{\xi}\right)^n = R^n \sum_{r=0}^n \binom{n}{r} \xi^{n-2r} m^r,$$

$$R^{2\alpha+\delta_o}\left(\xi+\frac{m}{\xi}\right)^{2\alpha+\delta_o} = R^{2\alpha+\delta_o}\sum_{r=0}^{2\alpha+\delta_o} \binom{2\alpha+\delta_o}{r} \xi^{2(\alpha-r)+\delta_o} m^r$$

$$R^n \left(\frac{1}{\xi} + m\xi\right)^n = R^n \left(m\xi + \frac{1}{\xi}\right)^n = R^n \sum_{r=0}^n \binom{n}{r} \xi^{n-2r} m^{n-r}$$

$$R^{2\alpha+\delta_o}\left(\frac{1}{\xi}+m\xi\right)^{2\alpha+\delta_o} = R^{2\alpha+\delta_o}\left(m\xi+\frac{1}{\xi}\right)^{2\alpha+\delta_o} = R^{2\alpha+\delta_o}\sum_{r=0}^{2\alpha+\delta_o}\binom{2\alpha+\delta_o}{r}\xi^{2(\alpha-r)+\delta_o}m^{n-r}.$$

If we adopt the form ϕ_1 from (3.9) for the exterior region $|\xi| > 1$, we have that

$$\phi_{1}\left(\xi\right) = \xi\left(KC_{1}R + iS\overline{D_{1}}Rm\right) + \frac{1}{\xi}\left(KC_{1}Rm + iS\overline{D_{1}}R\right) + SR_{(n,\xi)}$$
$$+ \sum_{n=a}^{b} \left[R^{n}\sum_{r=0}^{n}\left[\binom{n}{r}\xi^{n-2r}X_{n,(n,r)}\right] + \sum_{\alpha=0}^{\frac{n-(2+\delta_{o})}{2}}\left[R^{2\alpha+\delta_{o}}\sum_{r=0}^{2\alpha+\delta_{o}}\binom{2\alpha+\delta_{o}}{r}\xi^{2(\alpha-r)+\delta_{o}}X_{n,(\alpha,r)}\right]\right]$$

$$|\xi| > 1,$$
 (3.10)

where the constants,

$$X_{n,(n,r)} = KC_n m^r + iS\overline{D_n} m^{n-r}, \quad X_{n,(\alpha,r)} = KC_{n,\alpha} m^{r+1} + iS\overline{D_{n,\alpha}} m^{2\alpha+\delta_o-r+1},$$

and

$$R_{(n,\xi)} = \frac{\beta_2 R\left(\xi + \frac{m}{\xi}\right)}{\overline{C_1} + \sum_{n=a}^{b} \left[n\overline{C_n} R^{n-1} \left(\frac{1}{\xi} + \xi m\right)^{n-1} + m \sum_{\alpha=0}^{\frac{n-(2+\delta_o)}{2}} \left[(2\alpha + \delta_0) \overline{C_{n,\alpha}} R^{2\alpha + \delta_o - 1} \left(\frac{1}{\xi} + \xi m\right)^{2\alpha + \delta_o - 1} \right] \right]}.$$

From (3.1) and (3.4), the remote loading condition in the ξ – plane is taken to be

$$\phi_1(\xi) = A_1 R \xi + \sum_{n=a}^{b} \left[A_n R^n \sum_{r=0}^{\frac{n-\delta_0}{2}} \binom{n}{r} \xi^{n-2r} m^r \right], \quad |\xi| \to \infty,$$
(3.11)

Consequently, from (3.10), we have that

$$\phi_1\left(\xi\right) = \xi\left(KC_1R + iS\overline{D_1}Rm\right) + \lim_{|\xi| \to \infty} \left(SR_{(n,\xi)}\right)$$
$$+ \sum_{n=a}^b \left[R^n \sum_{r=0}^{\frac{n-\delta_0}{2}} \left[\binom{n}{r} \xi^{n-2r} X_{n,(n,r)}\right] + \sum_{\alpha=0}^{\frac{n-(2+\delta_0)}{2}} \left[R^{2\alpha+\delta_0} \sum_{r=0}^{\alpha} \binom{2\alpha+\delta_0}{r} \xi^{2(\alpha-r)+\delta_0} X_{n,(\alpha,r)}\right]\right]$$

$$, |\xi| \to \infty,$$

$$\therefore n - 2r \ge \delta_o \implies r \le \frac{n - \delta_o}{2},$$

$$2(\alpha - r) + \delta_o \ge \delta_o \implies r \le \alpha, \quad |\xi| \to \infty.$$

$$(3.12)$$

Clearly, this must also satisfy the remote loading condition (3.11), from which we derive the equations defining the unknown complex constants:

$$A_{1}R\xi + \sum_{n=a}^{b} \left[A_{n}R^{n} \sum_{r=1}^{\frac{n-\delta_{0}}{2}} \binom{n}{r} \xi^{n-2r} m^{r} \right] = \xi \left(KC_{1}R + iS\overline{D_{1}}Rm \right) + \lim_{|\xi| \to \infty} \left(SR_{(n,\xi)} \right) \\ + \sum_{n=a}^{b} \left[R^{n} \sum_{r=1}^{\frac{n-\delta_{0}}{2}} \left[\binom{n}{r} \xi^{n-2r} X_{n,(n,r)} \right] + \sum_{\alpha=0}^{\frac{n-(2+\delta_{0})}{2}} \left[R^{2\alpha+\delta_{0}} \sum_{r=0}^{\alpha} \binom{2\alpha+\delta_{0}}{r} \xi^{2(\alpha-r)+\delta_{0}} X_{n,(\alpha,r)} \right] \right],$$
(3.13A)

$$\sum_{n=a}^{b} [A_n] \xi^n = \sum_{n=a}^{b} \left[X_{n,(n,r=0)} \right] \xi^n = \sum_{n=a}^{b} \left[KC_n + iS\overline{D_n}m^n \right] \xi^n.$$
(3.13B)

Remark 4. As a direct consequence of Remark 3, we have that

$$KC_{1}R + iS\overline{D_{1}}Rm - \frac{1}{\xi^{2}} \left(KC_{1}Rm + iS\overline{D_{1}}R \right) + SR'(n,\xi)$$

$$+ \sum_{n=a}^{b} \left[R^{n} \sum_{r=0}^{n} (n-2r) \binom{n}{r} \xi^{n-2r-1} X_{n,(n,r)} \right]$$

$$+ \sum_{n=a}^{b} \left[\sum_{\alpha=0}^{\frac{n-(2+\delta_{o})}{2}} \left[R^{2\alpha+\delta_{o}} \sum_{r=0}^{2\alpha+\delta_{o}} (2(\alpha-r)+\delta_{o}) \binom{2\alpha+\delta_{o}}{r} \xi^{2(\alpha-r)+\delta_{o}-1} X_{n,(\alpha,r)} \right] \right] \neq 0,$$

$$\forall \xi : |\xi| \geq 1.$$

This implies, together with Remark 3 and Equations (3.13A-B), that there are restrictions on the constants A_1, A_n, B_1 and B_n characterizing the remote stresses.

It remains to determine the analytic function $\psi_1(\xi)$. From the interface conditions (3.7), (3.4-5) and (3.10) it follows that, for $|\xi| = 1$,

$$\begin{split} \psi_{1}\left(\xi\right) &= -i\alpha_{1}\left[\frac{1}{\xi}\left(K\overline{C_{1}}R - iSD_{1}Rm\right) + \xi\left(K\overline{C_{1}}Rm - iSD_{1}R\right)\right] \\ &+ i\alpha_{2}\overline{C_{1}}R\left(\frac{1}{\xi} + m\xi\right) + D_{1}R\left(\xi + \frac{m}{\xi}\right) \\ &- i\alpha_{1}\sum_{n=a}^{b}\left[R^{n}\sum_{r=0}^{n}\left[\binom{n}{r}\xi^{2r-n}\overline{X_{n,(n,r)}}\right] + \sum_{\alpha=0}^{\frac{n-(2+\delta_{0})}{2}}\left[R^{2\alpha+\delta_{0}}\sum_{r=0}^{2\alpha+\delta_{0}}\binom{2\alpha+\delta_{0}}{r}\xi^{2(r-\alpha)-\delta_{0}}\overline{X_{n,(\alpha,r)}}\right]\right] \\ &+ i\alpha_{2}\sum_{n=a}^{b}\left[\overline{C_{n}}R^{n}\sum_{r=0}^{n}\left[\binom{n}{r}\xi^{2r-n}m^{r}\right] + \sum_{\alpha=0}^{\frac{n-(2+\delta_{0})}{2}}\left[\overline{C_{n,\alpha}}R^{2\alpha+\delta_{0}}\sum_{r=0}^{2\alpha+\delta_{0}}\binom{2\alpha+\delta_{0}}{r}\xi^{2(r-\alpha)-\delta_{0}}m^{r+1}\right]\right] \\ &+ \sum_{n=a}^{b}[D_{n}R^{n}\sum_{r=0}^{n}\left[\binom{n}{r}\xi^{2r-n}m^{n-r}\right] \\ &+ \sum_{\alpha=0}^{\frac{n-(2+\delta_{0})}{2}}[D_{n,\alpha}R^{2\alpha+\delta_{0}}\sum_{r=0}^{2\alpha+\delta_{0}}\binom{2\alpha+\delta_{0}}{r}\xi^{2(r-\alpha)-\delta_{0}}m^{2\alpha+\delta_{0}-r+1}]] \end{split}$$

$$+\overline{R_{(n,\xi)}}i(1-\alpha_{1}S) - \frac{i\beta_{1}R^{2}\left(\frac{1}{\xi}+m\xi\right)\left(1-\frac{m}{\xi^{2}}\right)}{\phi_{1}'(\xi)},$$
(3.14)

where

$$\overline{R_{(n,\xi)}} = \frac{\beta_2 R\left(\frac{1}{\xi} + m\xi\right)}{C_1 + \sum_{n=a}^{b} \left[nC_n R^{n-1} \left(\xi + \frac{m}{\xi}\right)^{n-1} + m \sum_{\alpha=0}^{\frac{n-(2+\delta_0)}{2}} \left[(2\alpha + \delta_0) C_{n,\alpha} R^{2\alpha + \delta_0 - 1} \left(\xi + \frac{m}{\xi}\right)^{2\alpha + \delta_0 - 1} \right] \right]},$$
 and

$$\begin{split} \phi_1'\left(\xi\right) &= KC_1R + iS\overline{D_1}Rm - \frac{1}{\xi^2}\left(KC_1Rm + iS\overline{D_1}R\right) + SR'\left(n,\xi\right) \\ &+ \sum_{n=a}^b \left[R^n\sum_{r=0}^n \left[\left(n-2r\right)\binom{n}{r}\xi^{n-2r-1}X_{n,(n,r)}\right]\right] \\ &+ \sum_{n=a}^b \left[\sum_{\alpha=0}^{\frac{n-(2+\delta_o)}{2}} \left[R^{2\alpha+\delta_o}\sum_{r=0}^{2\alpha+\delta_o}\left(2\left(\alpha-r\right)+\delta_o\right)\binom{2\alpha+\delta_o}{r}\xi^{2(\alpha-r)+\delta_o-1}X_{n,(\alpha,r)}\right]\right], \end{split}$$

and

$$R^{n}\left(\overline{\xi}+\frac{m}{\overline{\xi}}\right)^{n} = R^{n}\left(\frac{1}{\xi}+m\xi\right)^{n} = R^{n}\sum_{r=0}^{n}\binom{n}{r}\xi^{2r-n}m^{r},$$

$$R^{2\alpha+\delta_o}\left(\overline{\xi}+\frac{m}{\overline{\xi}}\right)^{2\alpha+\delta_o} = R^{2\alpha+\delta_o}\left(\frac{1}{\xi}+m\xi\right)^{2\alpha+\delta_o} = R^{2\alpha+\delta_o}\sum_{r=0}^{2\alpha+\delta_o}\binom{2\alpha+\delta_o}{r}\xi^{\delta_o-2(\alpha-r)}m^r$$

$$R^n \left(m\overline{\xi} + \frac{1}{\overline{\xi}} \right)^n = R^n \left(\frac{m}{\xi} + \xi \right)^n = R^n \sum_{r=0}^n \binom{n}{r} \xi^{2r-n} m^{n-r}$$

$$R^{2\alpha+\delta_o}\left(m\overline{\xi}+\frac{1}{\overline{\xi}}\right)^{2\alpha+\delta_o} = R^{2\alpha+\delta_o}\sum_{r=0}^{2\alpha+\delta_o}\binom{2\alpha+\delta_o}{r}\xi^{\delta_o-2(\alpha-r)}m^{n-r}, \quad \because \xi\overline{\xi} = |\xi|^2 = 1.$$

From (3.1) and (3.4), ψ_1 must also satisfy the asymptotic condition

$$\psi_1(z) = B_1 R \xi + \sum_{n=a}^{b} \left[B_n R^n \sum_{r=\frac{n+\delta_o}{2}}^n \binom{n}{r} \xi^{2r-n} m^{n-r} \right], \quad |\xi| \to \infty.$$

Hence, If we adopt ψ_1 from (3.14) for $|\xi| > 1$, we require that

$$B_1R\xi + \sum_{n=a}^{b} \left[B_n R^n \sum_{r=\frac{n+\delta_0}{2}}^{n-1} \binom{n}{r} \xi^{2r-n} m^{n-r} \right] = \xi \left[-i\alpha_1 \left(K\overline{C_1}Rm - iSD_1R \right) + i\alpha_2\overline{C_1}Rm + D_1R \right]$$

$$+\sum_{n=a}^{b} \left[R^{n} \sum_{r=\frac{n+\delta_{a}}{2}}^{n-1} \binom{n}{r} \xi^{2r-n} \left(-i\alpha_{1} \overline{X_{n,(n,r)}} + i\alpha_{2} \overline{C_{n}} m^{r} + D_{n} m^{n-r} \right) \right] +$$

$$\sum_{n=a}^{b} \left[\sum_{\alpha=0}^{\frac{n-(2+\delta_{o})}{2}} \left[R^{2\alpha+\delta_{o}} \sum_{r=\alpha+\delta_{o}}^{2\alpha+\delta_{o}} \binom{2\alpha+\delta_{o}}{r} \xi^{2(r-\alpha)-\delta_{o}} \left(-i\alpha_{1}\overline{X_{n,(\alpha,r)}} + i\alpha_{2}\overline{C_{n,\alpha}}m^{r+1} + D_{n,\alpha}m^{2\alpha+\delta_{o}-r+1} \right) \right] \right]$$

$$+\lim_{|\xi|\to\infty} \left(\overline{R_{(n,\xi)}}i\left(1-\alpha_1S\right)\right) - \lim_{|\xi|\to\infty} \left(\frac{i\beta_1 R^2\left(\frac{1}{\xi}+m\xi\right)\left(1-\frac{m}{\xi^2}\right)}{\phi_1'\left(\xi\right)}\right),\qquad(3.15\text{ A})$$

$$\sum_{n=a}^{b} [B_n] \xi^n = \sum_{n=a}^{b} \left[-i\alpha_1 \overline{X_{n,(n,r=n)}} + i\alpha_2 \overline{C_n} m^n + D_n \right] \xi^n, \quad (3.15B)$$
$$\therefore 2r - n \ge \delta_o \implies r \ge \frac{n + \delta_o}{2},$$
$$2(r - \alpha) - \delta_o \ge \delta_o \implies r \ge \alpha + \delta_o, \quad |\xi| \to \infty,$$

where $X_{n,(n,r)}$ and $X_{n,(\alpha,r)}$ are defined in (3.10).

SUMMARY

The complete solution of the inhomogeneity-matrix system subjected to the general class of remote loadings described by (3.1) is given by

$$\phi_{2}(z) = C_{1}z + \sum_{n=a}^{b} \left[C_{n}z^{n} + m \sum_{\alpha=0}^{\frac{n-(2+\delta_{o})}{2}} C_{n,\alpha}z^{2\alpha+\delta_{o}} \right]$$
(3.16)

$$\psi_{2}(z) = D_{1}z + \sum_{n=a}^{b} \left[D_{n}z^{n} + m \sum_{\alpha=0}^{\frac{n-(2+\delta_{o})}{2}} D_{n,\alpha}z^{2\alpha+\delta_{o}} \right], \ z \in S_{2}, \ a \ge 2, b \ge a,$$

$$\phi_1(\xi) = \xi \left(KC_1R + iS\overline{D_1}Rm \right) + \frac{1}{\xi} \left(KC_1Rm + iS\overline{D_1}R \right) + SR_{(n,\xi)}$$
(3.17)

$$+\sum_{n=a}^{b} \left[R^{n} \sum_{r=0}^{n} \left[\binom{n}{r} \xi^{n-2r} X_{n,(n,r)} \right] + \sum_{\alpha=0}^{\frac{n-(2+\delta_{o})}{2}} \left[R^{2\alpha+\delta_{o}} \sum_{r=0}^{2\alpha+\delta_{o}} \binom{2\alpha+\delta_{o}}{r} \xi^{2(\alpha-r)+\delta_{o}} X_{n,(\alpha,r)} \right] \right]$$
$$, |\xi| > 1,$$

and

$$\psi_1(\xi) = -i\alpha_1 \left[\frac{1}{\xi} \left(K\overline{C_1}R - iSD_1Rm \right) + \xi \left(K\overline{C_1}Rm - iSD_1R \right) \right]$$

$$+i\alpha_{2}\overline{C_{1}}R\left(\frac{1}{\xi}+m\xi\right)+D_{1}R\left(\xi+\frac{m}{\xi}\right)$$
$$-i\alpha_{1}\sum_{n=a}^{b}\left[R^{n}\sum_{r=0}^{n}\left[\binom{n}{r}\xi^{2r-n}\overline{X_{n,(n,r)}}\right]+\sum_{\alpha=0}^{\frac{n-(2+\delta_{0})}{2}}\left[R^{2\alpha+\delta_{0}}\sum_{r=0}^{2\alpha+\delta_{0}}\binom{2\alpha+\delta_{0}}{r}\xi^{2(r-\alpha)-\delta_{0}}\overline{X_{n,(\alpha,r)}}\right]\right]$$
$$+i\alpha_{2}\sum_{n=a}^{b}\left[\overline{C_{n}}R^{n}\sum_{r=0}^{n}\left[\binom{n}{r}\xi^{2r-n}m^{r}\right]+\sum_{\alpha=0}^{\frac{n-(2+\delta_{0})}{2}}\left[\overline{C_{n,\alpha}}R^{2\alpha+\delta_{0}}\sum_{r=0}^{2\alpha+\delta_{0}}\binom{2\alpha+\delta_{0}}{r}\xi^{2(r-\alpha)-\delta_{0}}m^{r+1}\right]\right]$$
$$+\sum_{n=a}^{b}[D_{n}R^{n}\sum_{r=0}^{n}[\binom{n}{r}\xi^{2r-n}m^{n-r}]$$

$$+ \sum_{\alpha=0}^{\frac{n-(2+\delta_{o})}{2}} [D_{n,\alpha}R^{2\alpha+\delta_{o}} \sum_{r=0}^{2\alpha+\delta_{o}} {\binom{2\alpha+\delta_{o}}{r}} \xi^{2(r-\alpha)-\delta_{o}}m^{2\alpha+\delta_{o}-r+1}]] + \overline{R_{(n,\xi)}}i(1-\alpha_{1}S) - \frac{i\beta_{1}R^{2}\left(\frac{1}{\xi}+m\xi\right)\left(1-\frac{m}{\xi^{2}}\right)}{\phi_{1}'(\xi)}, \qquad (3.18)$$

where, again, $|\xi| > 1$. Here, the unknown complex constants $C_1, C_n, C_{n,\alpha}, D_1, D_n$ and $D_{n,\alpha}$ are determined by the equations

$$A_{1}R\xi + \sum_{n=a}^{b} \left[A_{n}R^{n} \sum_{r=1}^{\frac{n-\delta_{0}}{2}} \binom{n}{r} \xi^{n-2r} m^{r} \right] = \xi \left(KC_{1}R + iS\overline{D_{1}}Rm \right) + \lim_{|\xi| \to \infty} \left(SR_{(n,\xi)} \right) \\ + \sum_{n=a}^{b} \left[R^{n} \sum_{r=1}^{\frac{n-\delta_{0}}{2}} \left[\binom{n}{r} \xi^{n-2r} X_{n,(n,r)} \right] + \sum_{\alpha=0}^{\frac{n-(2+\delta_{0})}{2}} \left[R^{2\alpha+\delta_{0}} \sum_{r=0}^{\alpha} \binom{2\alpha+\delta_{0}}{r} \xi^{2(\alpha-r)+\delta_{0}} X_{n,(\alpha,r)} \right] \right],$$
(3.19A)

$$\sum_{n=a}^{b} [A_n] \xi^n = \sum_{n=a}^{b} [X_{n,(n,r=0)}] \xi^n = \sum_{n=a}^{b} [KC_n + iS\overline{D_n}m^n] \xi^n, \quad (3.19B)$$

$$B_{1}R\xi + \sum_{n=a}^{b} \left[B_{n}R^{n} \sum_{r=\frac{n+\delta_{o}}{2}}^{n-1} \binom{n}{r} \xi^{2r-n} m^{n-r} \right] = \xi \left[-i\alpha_{1} \left(K\overline{C_{1}}Rm - iSD_{1}R \right) + i\alpha_{2}\overline{C_{1}}Rm + D_{1}R \right]$$
$$+ \sum_{n=a}^{b} \left[R^{n} \sum_{r=\frac{n+\delta_{o}}{2}}^{n-1} \binom{n}{r} \xi^{2r-n} \left(-i\alpha_{1}\overline{X_{n,(n,r)}} + i\alpha_{2}\overline{C_{n}}m^{r} + D_{n}m^{n-r} \right) \right] + \sum_{n=a}^{b} \left[\sum_{\alpha=0}^{n-(2+\delta_{o})} \left[R^{2\alpha+\delta_{o}} \sum_{r=\alpha+\delta_{o}}^{2\alpha+\delta_{o}} \binom{2\alpha+\delta_{o}}{r} \xi^{2(r-\alpha)-\delta_{o}} \left(-i\alpha_{1}\overline{X_{n,(\alpha,r)}} + i\alpha_{2}\overline{C_{n,\alpha}}m^{r+1} + D_{n,\alpha}m^{2\alpha+\delta_{o}-r+1} \right) \right] \right]$$

$$+\lim_{|\xi|\to\infty} \left(\overline{R_{(n,\xi)}}i\left(1-\alpha_1S\right)\right) - \lim_{|\xi|\to\infty} \left(\frac{i\beta_1 R^2\left(\frac{1}{\xi}+m\xi\right)\left(1-\frac{m}{\xi^2}\right)}{\phi_1'\left(\xi\right)}\right),\qquad(3.19C)$$

$$\sum_{n=a}^{b} [B_n] \xi^n = \sum_{n=a}^{b} \left[-i\alpha_1 \overline{X_{n,(n,r=n)}} + i\alpha_2 \overline{C_n} m^n + D_n \right] \xi^n,$$
(3.19D)

where, from Remarks. 3 and 4, we must have that

$$A_{1}, \ C_{1} \neq 0,$$

$$C_{1} + \sum_{n=a}^{b} \left[nC_{n}R^{n-1} \left(\xi + \frac{m}{\xi}\right)^{n-1} + m \sum_{\alpha=0}^{\frac{n-(2+\delta_{o})}{2}} \left[(2\alpha + \delta_{0}) C_{n,\alpha}R^{2\alpha + \delta_{o}-1} \left(\xi + \frac{m}{\xi}\right)^{2\alpha + \delta_{o}-1} \right] \right] \neq 0$$

 \Leftrightarrow

$$\overline{C_1} + \sum_{n=a}^{b} \left[n \overline{C_n} R^{n-1} \left(\overline{\xi} + \frac{m}{\overline{\xi}} \right)^{n-1} + m \sum_{\alpha=0}^{\frac{n-(2+\delta_o)}{2}} \left[(2\alpha + \delta_0) \overline{C_{n,\alpha}} R^{2\alpha + \delta_o - 1} \left(\overline{\xi} + \frac{m}{\overline{\xi}} \right)^{2\alpha + \delta_o - 1} \right] \neq 0,$$

$$\begin{aligned} \text{for } & |\xi| = 1, \text{ and} \\ & KC_1R + iS\overline{D_1}Rm - \frac{1}{\xi^2} \left(KC_1Rm + iS\overline{D_1}R \right) + SR'\left(n,\xi\right) \\ & \quad + \sum_{n=a}^{b} \left[R^n \sum_{r=0}^{n} \left(n - 2r\right) \binom{n}{r} \xi^{n-2r-1} X_{n,(n,r)} \right] \\ & \quad + \sum_{n=a}^{b} \left[\sum_{\alpha=0}^{\frac{n-(2+\delta_0)}{2}} \left[R^{2\alpha+\delta_0} \sum_{r=0}^{2\alpha+\delta_0} \left(2\left(\alpha-r\right) + \delta_0\right) \binom{2\alpha+\delta_0}{r} \xi^{2(\alpha-r)+\delta_0-1} X_{n,(\alpha,r)} \right] \right] \neq 0, \\ & \quad \forall \xi : |\xi| \ge 1. \end{aligned}$$

This solution is rather complicated since it accommodates an extremely general class of remote loadings. However, the computation of the complete solution from this more general framework is not difficult when considering particular cases. For example, in Section 3.5, we demonstrate the case in which the system is subjected to a 'simple' nonuniform remote loading characterized by stress functions $\phi_1(z) = A_1 z + A_3 z^3$, $\psi_1(z) = B_1 z + B_3 z^3$, $|z| \rightarrow \infty$, in the matrix. Before, we do this, however, we use the above solution to derive general expressions for the important mechanical quantities describing the system.

3.4 FURTHER DISCUSSION

As a consequence of the results obtained above, we can draw the following interesting conclusions.

3.4.1 Expressions for I and J Inside the Elliptic Inhomogeneity

From (2.1) and (2.6), R and J can be calculated as follow:

$$J = \det \mathbf{F} = -\operatorname{Im} \left[w_{,1} \overline{w}_{,2} \right] = y_{1,1} y_{2,2} - y_{1,2} y_{2,1}, \tag{3.20}$$

$$R = tr(\mathbf{F}\mathbf{F}^{T}) = |w_{,1}|^{2} + |w_{,2}|^{2} = y_{p,s}y_{p,s}, \dots p, s = 1, 2,$$
(3.21)

Hence, in the view of (2.1), Equation (2.5) yields

$$I = \sqrt{R + 2J} = \sqrt{tr(\mathbf{F}\mathbf{F}^{T}) + 2J},$$

$$= \sqrt{(y_{1,1} + y_{2,2})^{2} + (y_{1,2} - y_{2,1})^{2}},$$
(3.22)

where

$$(y_{1,1} + y_{2,2})^2 + (y_{1,2} - y_{2,1})^2 = 2(y_{1,1}y_{2,2} - y_{1,2}y_{2,1}) + y_{11}^2 + y_{22}^2 + y_{12}^2 + y_{21}^2.$$

Equation (2.25) and (3.16) give us that

$$w_{2}(z) = -i\alpha_{2} \left[C_{1}z + \sum_{n=a}^{b} \left(C_{n}z^{n} + m \sum_{\alpha=0}^{\frac{n-(2+\delta_{o})}{2}} C_{n,\alpha}z^{2\alpha+\delta_{o}} \right) \right]$$

$$+\overline{D}_{1}\overline{z} + \sum_{n=a}^{b} \left(\overline{D}_{n}\overline{z}^{n} + m \sum_{\alpha=0}^{\frac{n-(2+\delta_{o})}{2}} \overline{D}_{n,\alpha}\overline{z}^{2\alpha+\delta_{o}} \right)$$

$$-\frac{i\beta_{2}z}{\overline{C_{1}} + \sum_{n=a}^{b} \left(n\overline{C_{n}}\overline{z}^{n-1} + m \sum_{\alpha=0}^{\frac{n-(2+\delta_{o})}{2}} (2\alpha+\delta_{o}) \overline{C}_{n,\alpha}\overline{z}^{2\alpha+\delta_{o}-1} \right)}.$$
(3.23)

Consequently, from (2.14) and (2.21), or directly from (2.36) I has the following expression within the inhomogeneity

$$I = P(|\phi'_{2}(z)|) = 2\left(\alpha |\phi'_{2}(z)| + \frac{\beta}{|\phi'_{2}(z)|}\right).$$

Therefore, from (3.16), we have that

$$I = 2\left[\alpha_{2}\left|C_{1} + \sum_{n=a}^{b}\left[nC_{n}z^{n-1} + m\sum_{\alpha=0}^{\frac{n-(2+\delta_{o})}{2}}\left(2\alpha + \delta_{o}\right)C_{n,\alpha}z^{2\alpha+\delta_{o}-1}\right]\right| + \frac{\beta_{2}}{\left|C_{1} + \sum_{n=a}^{b}\left[nC_{n}z^{n-1} + m\sum_{\alpha=0}^{\frac{n-(2+\delta_{o})}{2}}\left(2\alpha + \delta_{o}\right)C_{n,\alpha}z^{2\alpha+\delta_{o}-1}\right]\right|}\right],$$
(3.24)

and from Remark 3.

$$|\phi_{2}'(z)| = \left| C_{1} + \sum_{n=a}^{b} \left[nC_{n}z^{n-1} + m\sum_{\alpha=0}^{\frac{n-(2+\delta_{o})}{2}} (2\alpha + \delta_{o}) C_{n,\alpha}z^{2\alpha + \delta_{o}-1} \right] \right| \neq 0,$$

$$I^{2} - 16\alpha_{2}\beta_{2} = 4\left[\alpha_{2} \left|\phi_{2}'(z)\right| - \frac{\beta_{2}}{\left|\phi_{2}'(z)\right|}\right]^{2} > 0.$$
(3.25)

which guarantees that F'(I), from (2.21) is well-defined.

3.4.2 Stress Distribution

From (2.18), the Piola stresses inside the inhomogeneity and at the remote boundary are given by

$$-\sigma_{21} + i\sigma_{11} = (\chi_2)_{,2}, \quad \sigma_{22} - i\sigma_{12} = (\chi_2)_{,1}, \quad (3.26)$$
$$-\sigma_{21}^{\infty} + i\sigma_{11}^{\infty} = (\chi_1)_{,2}, \quad \sigma_{22}^{\infty} - i\sigma_{12}^{\infty} = (\chi_1)_{,1}.$$

From Equation (2.26), (3.1) and (3.16)

$$\chi_1(z) = 2\mu_1 i [(\alpha_1 - 1)(A_1 z + \sum_{n=a}^b A_n z^n) + i(\overline{B_1}\overline{z} + \sum_{n=a}^b \overline{B_n}\overline{z}^n) + \frac{\beta_1 z}{\overline{A_1} + \sum_{n=a}^b n\overline{A_n}\overline{z}^{n-1}}],$$

$$\chi_2(z) = 2\mu_2 i[(\alpha_2 - 1)(C_1 z + \sum_{n=a}^b (C_n z^n + m \sum_{\alpha=0}^{\frac{n-(2+\delta_0)}{2}} C_{n,\alpha} z^{2\alpha+\delta_0}))$$
(3.27)

$$+i(\overline{D}_{1}\overline{z}+\sum_{n=a}^{b}(\overline{D}_{n}\overline{z}^{n}+m\sum_{\alpha=0}^{\frac{n-(2+\delta_{o})}{2}}\overline{D_{n,\alpha}}\overline{z}^{2\alpha+\delta_{o}}))$$
$$+\frac{\beta_{2}z}{\overline{C_{1}}+\sum_{n=a}^{b}(n\overline{C_{n}}\overline{z}^{n-1}+m\sum_{\alpha=0}^{\frac{n-(2+\delta_{o})}{2}}(2\alpha+\delta_{o})\overline{C_{n,\alpha}}\overline{z}^{2\alpha+\delta_{o}-1})].$$

so that the Piola stresses both inside the inhomogeneity and at the remote boundary can be calculated from (3.19A-D), (3.26-27).

In addition, from (2.12) the Cauchy stresses inside the inhomogeneity are given by

$$\tau_{11} + \tau_{22} = 2\mu_2 [\frac{IF'(I)}{J} - 2],$$

$$\tau_{11} - \tau_{22} + 2i\tau_{12} = 2\mu_2 \frac{F'(I)}{IJ} [(w_2)_{,2}^2 + (w_2)_{,1}^2]$$

In our case, from Equation (2.14) and (3.20-25)

$$F'(I) = |\phi'_{2}(z)| = \left| C_{1} + \sum_{n=a}^{b} \left[nC_{n}z^{n-1} + m\sum_{\alpha=0}^{\frac{n-(2+\delta_{o})}{2}} (2\alpha + \delta_{o}) C_{n,\alpha}z^{2\alpha + \delta_{o}-1} \right] \right| \neq 0.$$

from which the Cauchy stresses can be calculated.

3.5 EXAMPLE: The Case a = b = 3, $\delta_o = 1$: Quadratic Remote Loading

The general polynomial solutions mentioned above can accommodate many different forms of applied remote loading. Among these, we consider the particular class of remote loading, characterized by the stress functions $\phi_1(z) = A_1 z + A_3 z^3$, $\psi_1(z) = B_1 z + B_3 z^3$, $|z| \to \infty$.

3.5.1 Complete Solutions

From (3.1) and (3.16), the stress functions at the remote boundary and inside the inhomogeneity are respectively given by

$$\phi_{1}(z) = A_{1}z + A_{3}z^{3}, \quad \psi_{1}(z) = B_{1}z + B_{3}z^{3}, \quad |z| \to \infty,$$

$$\phi_{2}(z) = C_{1}z + C_{3}z^{3} + mzC_{3,0}, \quad \psi_{2}(z) = D_{1}z + D_{3}z^{3} + mzD_{3,0}.$$
(3.28)

From the general solution (3.19A-D), the unknown complex constants $C_1, C_3, C_{3,0}, D_1, D_3$ and $D_{3,0}$ are completely determined by the equations

$$A_1 R\xi + 3A_3 R^3 m\xi = \xi \left(KC_1 R + iS\overline{D_1}Rm \right) + \lim_{|\xi| \to \infty} \left(S \frac{\beta_2 R\left(\xi + \frac{m}{\xi}\right)}{\overline{C_1} + 3\overline{C_3}R^2\left(\frac{1}{\xi} + m\xi\right)^2 + m\overline{C_{3,0}}} \right)$$

$$+3R^{3}\xi X_{n=3,(n=3,r=1)} + R\xi X_{n=3,(\alpha=0,r=0)},$$
(3.29A)

$$B_{1}R\xi + 3B_{3}R^{3}\xi m = \xi \left[-i\alpha_{1} \left(K\overline{C_{1}}Rm - iSD_{1}R \right) + i\alpha_{2}\overline{C_{1}}Rm + D_{1}R \right] \\ + \lim_{|\xi| \to \infty} \left(i \left(1 - \alpha_{1}S \right) \frac{\beta_{2}R\left(\frac{1}{\xi} + m\xi\right)}{C_{1} + 3C_{3}R^{2}\left(\xi + \frac{m}{\xi}\right)^{2} + mC_{3,0}} \right)$$

$$-\lim_{|\xi|\to\infty}\left(\frac{i\beta_1 R^2 \left(\frac{1}{\xi}+m\xi\right) \left(1-\frac{m}{\xi^2}\right)}{\phi_1'\left(\xi\right)}\right)+3R^3\xi \left(-i\alpha_1 \overline{X_{n=3,(n=3,r=2)}}+i\alpha_2 \overline{C_3}m^2+D_3m\right)$$

$$+R\xi\left(-i\alpha_{1}\overline{X_{n=3,(\alpha=0,r=1)}}+i\alpha_{2}\overline{C_{3,0}}m^{2}+D_{3,0}m\right),$$
(3.29B)

$$A_3 = KC_3 + iS\overline{D_3}m^3, \quad B_3 = -i\alpha_1\overline{X_{3,(n=3,r=3)}} + i\alpha_2\overline{C_3}m^3 + D_3, \quad (3.29C)$$

where

$$\phi_1'(\xi) = KC_1R + iS\overline{D_1}Rm - \frac{1}{\xi^2} \left(KC_1Rm + iS\overline{D_1}R \right)$$
$$+S\left(\frac{\beta_2 R \left(\xi + \frac{m}{\xi}\right)}{\overline{C_1} + 3\overline{C_3}R^2 \left(\frac{1}{\xi} + m\xi\right)^2 + m\overline{C_{3,0}}}\right)'$$

$$+R^{3}\sum_{r=0}^{3}\left[\binom{3}{r}\left(3-2r\right)X_{n=3,(n=3,r)}\xi^{2(1-r)}\right]+R\sum_{r=0}^{1}\left[\binom{1}{r}\left(1-2r\right)\xi^{-2r}X_{n=3,(\alpha=0,r)}\right],$$

and A_1, A_2, B_1 and B_2 are prescribed complex constants. Assume $C_3, C_{3,0} \neq 0$, then from (3.10), we finally have that

$$A_{\gamma} = KC_{\gamma} + iS\overline{D_{\gamma}}m^{\gamma}, \quad B_{\gamma} = -i\alpha_1 \left[K\overline{C_{\gamma}}m^{\gamma} - iSD_{\gamma} \right] + i\alpha_2\overline{C_{\gamma}}m^{\gamma} + D_{\gamma},$$

 $\gamma = 1,3$ (no sum over repeated indices),

$$3A_3R^2m = 3R^2 \left[KC_3m + iS\overline{D_3}m^2 \right] + KC_{3,0}m + iS\overline{D_{3,0}}m^2,$$

$$3B_{3}R^{2}m = 3R^{2} \left[-i\alpha_{1} \left(K\overline{C_{3}}m^{2} - iSD_{3}m\right) + i\alpha_{2}\overline{C_{3}}m^{2} + D_{3}m\right]$$
$$-i\alpha_{1} \left(K\overline{C_{3,0}}m^{2} - iSD_{3,0}m\right) + i\alpha_{2}\overline{C_{3,0}}m^{2} + D_{3,0}m.$$

From Remarks. 3 and 4, we must have

$$C_{1} + 3C_{3}R^{2}\left(\xi + \frac{m}{\xi}\right)^{2} + mC_{3,0} \neq 0$$

$$\iff \overline{C_{1}} + 3\overline{C_{3}}R^{2}\left(\overline{\xi} + \frac{m}{\overline{\xi}}\right)^{2} + m\overline{C_{3,0}} \neq 0, \quad |\xi| = 1,$$

$$(3.30)$$

and, in the surrounding matrix

$$KC_{1}R + iS\overline{D_{1}}Rm - \frac{1}{\xi^{2}}\left(KC_{1}Rm + iS\overline{D_{1}}R\right) + S\left(\frac{\beta_{2}R\left(\xi + \frac{m}{\xi}\right)}{\overline{C_{1}} + 3\overline{C_{3}}R^{2}\left(\frac{1}{\xi} + m\xi\right)^{2} + m\overline{C_{3,0}}}\right)' + R^{3}\sum_{r=0}^{3}\left[\binom{3}{r}\left(3 - 2r\right)X_{n=3,(n=3,r)}\xi^{2(1-r)}\right] + R\sum_{r=0}^{1}\left[\binom{1}{r}\left(1 - 2r\right)\xi^{-2r}X_{n=3,(\alpha=0,r)}\right] \neq 0, \ \forall \xi : |\xi| \ge 1.$$
(3.31)

3.5.2 Determination of I and J Inside the Elliptic Inhomogeneity

From (3.23), the deformation functions now take the form

$$w_{2} = -i\alpha_{2}[(C_{1} + mC_{3,0})z + C_{3}z^{3}] + (\overline{D_{1}} + m\overline{D_{3,0}})\bar{z} + \overline{D_{3}}\bar{z}^{3} - \frac{i\beta_{2}z}{\overline{C_{1}} + m\overline{C_{3,0}} + 3\overline{C_{3}}\bar{z}^{2}},$$
(3.32)

and from (3.24)

$$I = 2\left(\alpha_2 \left|C_1 + mC_{3,0} + 3C_3 z^2\right| + \frac{\beta_2}{\left|C_1 + mC_{3,0} + 3C_3 z^2\right|}\right),\$$

$$|\phi'_{2}(z)| = |C_{1} + mC_{3,0} + 3C_{3}z^{2}| \neq 0$$
, (from Remark 3),

$$I^{2} - 16\alpha_{2}\beta_{2} = 4\left(\alpha_{2}\left|C_{1} + mC_{3,0} + 3C_{3}z^{2}\right| - \frac{\beta_{2}}{\left|C_{1} + mC_{3,0} + 3C_{3}z^{2}\right|}\right)^{2} \ge 0, \quad (3.33)$$

from which I can be calculated. Henceforth, F'(I), from (2.21) is well defined. Also J can now be determined from (3.20) together with (3.32).

$$J = -\operatorname{Im}\left[\left(w_{2}\right)_{,1}\left(\overline{w}_{2}\right)_{,2}\right],\,$$

where

$$w_2 = -i\alpha_2[(C_1 + mC_{3,0})z + C_3 z^3] + (\overline{D_1} + m\overline{D_{3,0}})\bar{z} + \overline{D_3}\bar{z}^3 - \frac{i\beta_2 z}{\overline{C_1} + m\overline{C_{3,0}} + 3\overline{C_3}\bar{z}^2}$$

3.5.3 Deformed Contour

From (3.32), we can easily plot an example of the deformed contour (see Figure 4). The fact that the corresponding complex constants $C_1, C_3, C_{3,0}, D_1, D_2$ and $D_{3,0}$ satisfy (3.30-31) guarantee that there is no overlapping.



Figure 4: Example of a Deformed Elliptic Inhomogeneity Subjected to a Remote Loading Characterized by (3.28). Here, we have taken $(A_1 = 0.75i, A_3 = -0.02i, B_1 = 2.4, B_3 = -0.01, \mu_1 = 0.27, \mu_2 = 0.3, \alpha_1 = \alpha_2 = \beta_1 = \beta_2 = 0.5)$

3.5.4 Admissibility Conditions

From (2.1), (3.20) and (3.32-33), inside the inhomogeneity,

$$I^2 > 16\alpha_2\beta_2, \quad J > 0.$$
 (3.34)

As in Ru (2002), (3.34) again ensure that there is no overlapping of the deformation field (interior to the elliptic inhomogeneity).

3.5.5 The Piola Stress Distribution

From (3.27), the Piola stresses in the matrix and in the inhomogeneity are described, respectively, by:

$$\chi_1(z) = 2\mu_1 i [(\alpha_1 - 1)(A_1 z + A_3 z^3) + i(\overline{B_1} \overline{z} + \overline{B_3} \overline{z}^3) + \frac{\beta_1 z}{\overline{A_1} + 3\overline{A_3} \overline{z}^2}],$$

$$\chi_{2}(z) = 2\mu_{2}i[(\alpha_{2}-1)((C_{1}+mC_{3,0})z+C_{3}z^{3}) + i((\overline{D_{1}}+m\overline{D_{3,0}})\overline{z}+\overline{D_{3}}\overline{z}^{3}) + \frac{\beta_{2}z}{\overline{C_{1}}+m\overline{C_{3,0}}+3\overline{C_{3}}\overline{z}^{2}}].$$

Thus, the corresponding Piola stresses can be found from (3.26) as follows:

$$\begin{aligned} -\sigma_{21} + i\sigma_{11} &= 2\mu_2 i [i(\alpha_2 - 1)(C_1 + mC_{3,0} + 3C_3 z^2) + \overline{D_1} + m\overline{D_{3,0}} + 3\overline{D_3}\overline{z}^2 \\ &+ i\beta_2 \left(\frac{\overline{C_1} + m\overline{C_{3,0}} + 3\overline{C_3}\overline{z}(\overline{z} + 2z)}{(\overline{C_1} + m\overline{C_{3,0}} + 3\overline{C_3}\overline{z}^2)^2} \right)], \end{aligned}$$

$$\begin{split} \sigma_{22} - i\sigma_{12} &= 2\mu_2 i [(\alpha_2 - 1)(C_1 + mC_{3,0} + 3C_3 z^2) + i(\overline{D_1} + m\overline{D_{3,0}} + 3\overline{D_3}\overline{z}^2) \\ &+ \beta_2 \left(\frac{\overline{C_1} + m\overline{C_{3,0}} + 3\overline{C_3}\overline{z}(\overline{z} - 2z)}{(\overline{C_1} + m\overline{C_{3,0}} + 3\overline{C_3}\overline{z}^2)^2} \right)], \end{split}$$

$$\begin{aligned} -\sigma_{21}^{\infty} + i\sigma_{11}^{\infty} &= 2\mu_1 i [i(\alpha_1 - 1)(A_1 + 3A_3 z^2) + \overline{B_1} + 3\overline{B_3}\overline{z}^2 \\ &+ i\beta_1 \lim_{|z| \to \infty} \left(\frac{\overline{A_1} + 3\overline{A_3}\overline{z}(\overline{z} + 2z)}{(\overline{A_1} + 3\overline{A_3}\overline{z}^2)^2} \right)], \end{aligned}$$

$$\begin{split} \sigma_{22}^{\infty} - i\sigma_{12}^{\infty} &= 2\mu_1 i [(\alpha_1 - 1)(A_1 + 3A_3 z^2) + i(\overline{B_1} + 3\overline{B_3}\overline{z}^2) \\ &+ \beta_1 \lim_{|z| \to \infty} \left(\frac{\overline{A_1} + 3\overline{A_3}\overline{z}(\overline{z} - 2z)}{(\overline{A_1} + 3\overline{A_3}\overline{z}^2)^2} \right)]. \end{split}$$

For example, the Piola stresses at the origin ($z = x_1 + ix_2$, $x_1 = x_2 = 0$, center of elliptic inhomogeneity) can be calculated as

$$-\sigma_{21} + i\sigma_{11} = 2\mu_2 i [i(\alpha_2 - 1)(C_1 + mC_{3,0}) + \overline{D_1} + m\overline{D_{3,0}} + i\beta_2 \frac{1}{\overline{C_1} + m\overline{C_{3,0}}}],$$

$$\sigma_{22} - i\sigma_{12} = 2\mu_2 i [(\alpha_2 - 1) (C_1 + mC_{3,0}) + i (\overline{D_1} + m\overline{D_{3,0}}) + \beta_2 \frac{1}{\overline{C_1} + m\overline{C_{3,0}}}].$$

and, clearly, the corresponding deformation is , from (3.32), $w_{2}(z) = 0$.

Remark 5. We note from (3.28) and (3.29A-C) that if we set $A_3 = B_3 = C_3 = C_{3,0} = D_3 = D_{3,0} = 0$, we obtain exactly the corresponding solutions for the simpler case

presented in Ru et al (2005) when the elliptic inhomogeneity-matrix system is subjected to uniform remote loading.

3.6 THE CASE OF THE CIRCULAR INHOMOGENEITY

As we previously mentioned, the solution derived above is sufficiently general to accommodate a wide class of inhomogeneity problems. In this section, from the general solution presented in the previous sections, we obtain the complete set of solutions for a perfectly bonded *circular* inhomogeneity-matrix system (a particular class of an elliptic inhomogeneity) subjected to the same class of remote stresses (Kim & Schiavone, 2007b).

3.6.1 Complete Solution in the Case of the Circular Inhomogeneity

By setting m = 0, the mapping function (3.2) becomes:

$$z = \omega(\xi) = R\xi, \quad R > 0, \quad \xi = \rho e^{i\theta}, \quad \rho = 1,$$
 (3.35)

which implies that we have a circle whose radius is R in the z-plane as a special case of an ellipse. In other words, we can actually derive the complete solution for the circular inhomogeneity problems from the general solution presented in the previous section, by applying the the following condition:

$$z = \omega\left(\xi\right) = R\left(\xi + \frac{m}{\xi}\right), \quad R > 0, \quad m = 0,$$

(a circular mapping function: a particular case of the ellipse).

From (3.16) together with (3.2), we obtain, for m = 0,

$$\phi_{2}(z) = C_{1}R\xi + \sum_{n=a}^{b} C_{n}R\xi^{n}$$

$$\psi_{2}(z) = D_{1}R\xi + \sum_{n=a}^{b} D_{n}R\xi^{n}, \quad R\xi \in S_{2}, \ a \ge 2, b \ge a,$$
(3.36)

and from (3.17-18), we have that, for m = 0,

$$\phi_{1}\left(\xi\right) = KC_{1}R\xi + \frac{R}{\xi}iS\overline{D_{1}} + \sum_{n=a}^{b}\left(R^{n}\xi^{n}KC_{n}\right) + \sum_{n=a}^{b}\left(\xi^{-n}iS\overline{D_{n}}\right) + SR_{(n,\xi)},$$
$$, |\xi| > 1, \quad \left(\because \binom{n}{0} = \binom{n}{n} = 1\right), \tag{3.37}$$

$$\psi_1\left(\xi\right) = -i\alpha_1 \left[K\overline{C_1}\frac{R}{\xi} - iSD_1R\xi \right] + i\alpha_2\overline{C_1}\frac{R}{\xi} + D_1R\xi$$

$$-i\alpha_1 \sum_{n=a}^{b} \left[R^n \xi^{-n} \overline{KC_n} + R^n \xi^n \overline{iSD_n} \right] + i\alpha_2 \sum_{n=a}^{b} \left(\overline{C_n} R^n \xi^{-n} \right) + \sum_{n=a}^{b} \left(D_n R^n \xi^n \right) \quad (3.38)$$

$$+\overline{R_{(n,\xi)}}i\left(1-\alpha_1S\right)-\frac{i\beta_1R^2\frac{1}{\xi}}{\phi_1'\left(\xi\right)},$$

where

$$R_{(n,\xi)} = \frac{\beta_2 R\xi}{\overline{C_1} + \sum_{n=a}^{b} \left[n \overline{C_n} \left(\frac{R}{\xi} \right)^{n-1} \right]},$$

and

$$X_{n,(n,r)} = KC_n m^r + iS\overline{D_n} m^{n-r} = \begin{cases} KC_n, & r = 0, \\ iS\overline{D_n}, & r = n, \end{cases}$$
$$, X_{n,(\alpha,r)} = KC_{n,\alpha} m^{r+1} + iS\overline{D_{n,\alpha}} m^{2\alpha+\delta_o-r+1} = 0.$$

Finally, from (3.19A-D), the unknown complex constants C_1 , C_n , D_1 , and D_n are determined, for m = 0

$$A_1 R\xi = KC_1 R\xi + \frac{S\beta_2 R\xi}{\overline{C_1}}, \quad \sum_{n=a}^{b} [A_n] R\xi^n = \sum_{n=a}^{b} [KC_n] R\xi^n, \quad (3.39A)$$

$$\therefore \quad X_{n,(n,r=0)} = KC_n + iS\overline{D_n}m^n = KC_n,$$

$$B_1 R\xi = R\xi D_1 (1 - \alpha_1 S), \quad \sum_{n=a}^{b} [B_n] R\xi^n = \sum_{n=a}^{b} [D_n (1 - \alpha_1 S)] R\xi^n, \quad (3.39B)$$

$$\therefore \quad \overline{X_{n,(n,r=n)}} = K \overline{C}_n m^n - iS D_n = -iS D_n.$$

where

$$K = \frac{\mu_2}{\mu_1} + \alpha_2 \left(1 - \frac{\mu_2}{\mu_1} \right), \quad S = \left(1 - \frac{\mu_2}{\mu_1} \right).$$

Hence, from (3.36-3.39(A, B)), we can derive the complete set of solutions of the inhomogeneitymatrix system subjected to remote loading described by (3.1). In fact, by letting $R \equiv R_c$ and in the view of (3.35), Equations (3.36-3.39(A, B)) immediately reduced to the corresponding set of solutions in Kim & Schiavone (2007b). We also mention here that the solution for the circular inhomogeneity-matrix system obtained in this section further reduces to the results presented in Ru et al (2005) and Ogden & Isherwood (1978) provided that the same conditions are considered.

SUMMARY

The complete solution for the circular inhomogeneity cases is given by

$$\phi_2(z) = C_1 z + \sum_{n=a}^{b} C_n z^n, \quad \psi_2(z) = D_1 z + \sum_{n=a}^{b} D_n z^n, \quad |z| < R_c, \quad a \ge 2, b \ge a, \quad (3.40)$$

$$\phi_1(z) = X_1 z + \sum_{n=a}^{b} (X_n z^n) + \frac{X_{-1}}{z} + \sum_{n=a}^{b} (\frac{X_{-n}}{z^n}) + R(z), \quad |z| > R_c, \quad (3.41)$$

and

$$\psi_{1}(z) = -i\alpha_{1}\left[\frac{\overline{X_{1}}R_{c}^{2}}{z} + \sum_{n=a}^{b}\left(\frac{\overline{X_{n}}R_{c}^{2n}}{z^{n}}\right) + \frac{\overline{X_{-1}}}{R_{c}^{2}}z + \sum_{n=a}^{b}\left(\frac{\overline{X_{-n}}}{R_{c}^{2n}}z^{n}\right) + \frac{S\beta_{2}R_{c}^{2}}{z\left[\sum_{n=a}^{b}(nC_{n}z^{n-1}) + C_{1}\right]}\right] - \frac{i\beta_{1}R_{c}^{2}}{z\left[X_{1} + \sum_{n=a}^{b}(nX_{n}z^{n-1}) - \frac{X_{-1}}{z^{2}} - \sum_{n=a}^{b}\left(\frac{nX_{-n}}{z^{n+1}}\right) + R'(z)\right]}$$

$$+i\alpha_{2}\left[\frac{\overline{C_{1}}R_{c}^{2}}{z}+\sum_{n=a}^{b}\left(\frac{\overline{C_{n}}R_{c}^{2n}}{z^{n}}\right)\right]+\frac{i\beta_{2}R_{c}^{2}}{z(C_{1}+\sum_{n=a}^{b}nC_{n}z^{n-1})}+D_{1}z+\sum_{n=a}^{b}D_{n}z^{n},\quad(3.42)$$

where , $|z| > R_c$. Here, the unknown complex constants C_1, C_n, D_1 and D_n are determined by the equations

$$A_1 = \left[\frac{\mu_2}{\mu_1} + \alpha_2(1 - \frac{\mu_2}{\mu_1})\right]C_1 + \left(1 - \frac{\mu_2}{\mu_1}\right)\frac{\beta_2}{\overline{C_1}}, \quad B_1 = D_1\left[1 - \alpha_1(1 - \frac{\mu_2}{\mu_1})\right],$$

$$\sum_{n=a}^{b} A_n z^n = \sum_{n=a}^{b} \left[\left(\frac{\mu_2}{\mu_1} + \alpha_2 (1 - \frac{\mu_2}{\mu_1}) \right) C_n \right] z^n, \quad \sum_{n=a}^{b} B_n z^n = \sum_{n=a}^{b} \left[D_n (1 - \alpha_1 (1 - \frac{\mu_2}{\mu_1})) \right] z^n, \tag{3.43}$$

where, from Remarks. 3 and 4, we must have that for $R \equiv R_c$, $z = R\xi$,

 $A_1, C_1 \neq 0,$

$$C_{1} + \sum_{n=a}^{b} nC_{n} z^{n-1} \neq 0 \iff \overline{C_{1}} + \sum_{n=a}^{b} n\overline{C_{n}} z^{n-1} \neq 0, \quad \forall z : |z| \le R_{c},$$
$$X_{1} + \sum_{n=a}^{b} (nX_{n} z^{n-1}) - \frac{X_{-1}}{z^{2}} - \sum_{n=a}^{b} (\frac{nX_{-n}}{z^{n+1}}) + R'(z) \neq 0, \quad \forall z : |z| \ge R_{c}.$$

3.6.2 Expressions for I and J Inside the Circular Inhomogeneity

From (3.23) we have for m = 0

$$w_{2} = -i\alpha_{2}[C_{1}z + \sum_{n=a}^{b} C_{n}z^{n}] + \overline{D}_{1}\overline{z} + \sum_{n=a}^{b} \overline{D_{n}}\overline{z}^{n} - \frac{i\beta_{2}z}{\overline{C_{1}} + \sum_{n=a}^{b} n\overline{C_{n}}\overline{z^{n-1}}}, \quad (3.44)$$

and from (3.24)

$$I = 2\left(\alpha_2 \left| C_1 + \sum_{n=a}^{b} nC_n z^{n-1} \right| + \frac{\beta_2}{\left| C_1 + \sum_{n=a}^{b} nC_n z^{n-1} \right|} \right).$$
(3.45)

Finally, from Remark 3

$$|\phi'_{2}(z)| = \left|C_{1} + \sum_{n=a}^{b} nC_{n}z^{n-1}\right| \neq 0.$$

$$I^{2} - 16\alpha_{2}\beta_{2} = 4\left(\alpha_{2}\left|C_{1} + \sum_{n=a}^{b} nC_{n}z^{n-1}\right| - \frac{\beta_{2}}{\left|C_{1} + \sum_{n=a}^{b} nC_{n}z^{n-1}\right|}\right)^{2} > 0, \quad (3.46)$$

which guarantees that F'(I), from (2.21) is well-defined.

3.6.3 Stress Distribution

From (3.27), we have that, for m = 0

$$\chi_1(z) = 2\mu_1 i [(\alpha_1 - 1)(A_1 z + \sum_{n=a}^b A_n z^n) + i(\overline{B_1}\overline{z} + \sum_{n=a}^b \overline{B_n}\overline{z}^n) + \frac{\beta_1 z}{\overline{A_1} + \sum_{n=a}^b n\overline{A_n}\overline{z^{n-1}}}],$$

$$\chi_{2}(z) = 2\mu_{2}i[(\alpha_{2}-1)(C_{1}z + \sum_{n=a}^{b}C_{n}z^{n}) + i(\overline{D_{1}}\bar{z} + \sum_{n=a}^{b}\overline{D_{n}}\bar{z}^{n}) + \frac{\beta_{2}z}{\overline{C_{1}} + \sum_{n=a}^{b}n\overline{C_{n}}z^{n-1}}],$$
(3.47)

so that the Piola stresses both inside the inhomogeneity and at the remote boundary can be calculated from (3.43), (3.26) and (3.47).

In addition, from Equation (2.14), (3.21-22) and (3.44-46), we have that, for m = 0

$$F'(I) = |\phi'_2(z)| = \left|C_1 + \sum_{n=a}^b nC_n z^{n-1}\right| \neq 0,$$

from with the Cauchy stresses inside the inhomogeneity can be evaluated by (2.12)

3.6.4 Example: The Case a = b = 2.

Similar to the previous section, here, we consider the particular class of remote stresses, characterized by the stress functions $\phi_1(z) = A_1 z + A_2 z^2$, $\psi_1(z) = B_1 z + B_2 z^2$, where A_1, A_2, B_1 and B_2 are prescribed complex constants.

Complete Solutions

From (3.1) and (3.40), the stress functions at the remote boundary and inside the inhomogeneity, respectively, are given by

$$\phi_{1}(z) = A_{1}z + A_{2}z^{2}, \quad \psi_{1}(z) = B_{1}z + B_{2}z^{2}, \quad |z| \to \infty,$$

$$\phi_{2}(z) = C_{1}z + C_{2}z^{2}, \quad \psi_{2}(z) = D_{1}z + D_{2}z^{2}, \quad |z| < R_{c}$$
(3.48)

From the general solution (3.43), the unknown complex constants C_1, C_2, D_1 , and D_2 are completely determined by the equations

$$A_{1} = \left[\frac{\mu_{2}}{\mu_{1}} + \alpha_{2}\left(1 - \frac{\mu_{2}}{\mu_{1}}\right)\right]C_{1} + \left(1 - \frac{\mu_{2}}{\mu_{1}}\right)\frac{\beta_{2}}{\overline{C_{1}}}, \quad B_{1} = D_{1}\left[1 - \alpha_{1}\left(1 - \frac{\mu_{2}}{\mu_{1}}\right)\right], \quad (3.49)$$

$$A_{2} = \left[\frac{\mu_{2}}{\mu_{1}} + \alpha_{2}\left(1 - \frac{\mu_{2}}{\mu_{1}}\right)\right]C_{2}, \quad B_{2} = D_{2}\left[1 - \alpha_{1}\left(1 - \frac{\mu_{2}}{\mu_{1}}\right)\right].$$

From Remarks. 3 and 4, we must have

$$C_1 + 2C_2 z \neq 0 \iff \overline{C_1} + 2\overline{C_2}\overline{z} \neq 0, \quad \forall z : |z| \le R_c, \tag{3.50}$$

and in the surrounding matrix

$$KC_1 + 2KC_2z - \frac{iS\overline{D_1}R_c^2}{z^2} - \frac{2iS\overline{D_2}R_c^4}{z^3} + S\beta_2\left(\frac{4\overline{C_2}R_c^2}{(\frac{z}{\overline{C_2}R_c^2} + \overline{C_1})}{(\frac{2\overline{C_2}R_c^2}{z} + \overline{C_1})^2}\right) \neq 0, \quad \forall z : |z| \ge R_c.$$
(3.51)

Determination of I and J Inside the inhomogeneity

From (3.44), the deformation function inside the inhomogeneity now takes the form

$$w_2 = -i\alpha_2[C_1z + C_2z^2] + \overline{D_1}\overline{z} + \overline{D_2}\overline{z}^2 - \frac{i\beta_2z}{\overline{C_1} + 2\overline{C_2}\overline{z}},$$
(3.52)

and from (3.45)

$$I = 2\left(\alpha_2 |C_1 + 2C_2 z| + \frac{\beta_2}{|C_1 + 2C_2 z|}\right),$$

$$|\phi_2'(z)| = |C_1 + 2C_2 z| \neq 0, \text{(from Remark. 2)},$$

$$I^{2} - 16\alpha_{2}\beta_{2} = 4\left(\alpha_{2}\left|C_{1} + 2C_{2}z\right| - \frac{\beta_{2}}{\left|C_{1} + 2C_{2}z^{1}\right|}\right)^{2} \ge 0,$$
(3.53)

from which I can be calculated. Henceforth, F'(I), from (2.21) again is well defined. Also J can now be determined from (3.20) together with (3.52).

$$J = -\operatorname{Im}\left[\left(w_2\right)_{,1}\left(\overline{w}_2\right)_{,2}\right],\,$$

where

$$w_2 = -i\alpha_2[C_1z + C_2z^2] + \overline{D_1}\overline{z} + \overline{D_2}\overline{z}^2 - \frac{i\beta_2z}{\overline{C_1} + 2\overline{C_2}\overline{z}}.$$

Deformed contour

From (41), we can easily plot the deformed contour (see Figure 5). The fact that the corresponding complex constants C_1, C_2, D_1 and D_2 satisfy (3.50) and (3.51) guarantee that there is no overlapping.



Figure 5: Example of a Deformed Circular Inhomogeneity Subjected to a Remote Loading Characterized by (3.48). Here, we have taken ($A_1 = 3.5$, $A_2 = 1$, $B_1 = 0.5$, $B_2 = 0.25$, $\mu_1 = 0.27$, $\mu_2 = 0.3$, $\alpha_1 = \alpha_2 = \beta_1 = \beta_2 = 0.5$)

Admissibility conditions

From (3.20) and (3.52-53), inside inhomogeneity,

$$I^2 > 16\alpha_2\beta_2, \quad J > 0.$$
 (3.54)

Again, as in Ru (2002), (3.54) ensure that there is no overlapping of the deformation field (interior to the circular inhomogeneity).

The Piola Stress Distribution

From (3.47), the Piola stress in the inhomogeneity and the matrix is described, respectively, by:

$$\chi_1(z) = 2\mu_1 i [(\alpha_1 - 1)(A_1 z + A_2 z^2) + i(\overline{B_1} \overline{z} + \overline{B_2} \overline{z}^2) + \frac{\beta_1 z}{\overline{A_1} + 2\overline{A_2} \overline{z}}],$$

$$\chi_2(z) = 2\mu_2 i [(\alpha_2 - 1)(C_1 z + C_2 z^2) + i(\overline{D_1} \bar{z} + \overline{D_2} \bar{z}^2) + \frac{\beta_2 z}{\overline{C_1} + 2\overline{C_2} \bar{z}}],$$

Thus, the corresponding Piola stress can be found from (3.26) as follows:

$$-\sigma_{21} + i\sigma_{11} = 2\mu_2 i [i(\alpha_2 - 1)(C_1 + 2C_2 z) + \overline{D_1} + 2\overline{D_2}\overline{z} + i\beta_2 \left(\frac{\overline{C_1} + 2\overline{C_2}(\overline{z} + z)}{(\overline{C_1} + 2\overline{C_2}\overline{z})^2}\right)],$$

$$\sigma_{22} - i\sigma_{12} = 2\mu_2 i [(\alpha_2 - 1)(C_1 + 2C_2 z) + i(\overline{D_1} + 2\overline{D_2}\overline{z}) + \beta_2 \left(\frac{\overline{C_1} + 2\overline{C_2}(\overline{z} - z)}{(\overline{C_1} + 2\overline{C_2}\overline{z})^2}\right)],$$

$$-\sigma_{21}^{\infty} + i\sigma_{11}^{\infty} = 2\mu_1 i [i(\alpha_1 - 1)(A_1 + 2A_2z) + \overline{B_1} + 2\overline{B_2}\overline{z} + i\beta_1 \lim_{|z| \to \infty} \left(\frac{\overline{A_1} + 2\overline{A_2}(\overline{z} + z)}{(\overline{A_1} + 2\overline{A_2}\overline{z})^2}\right)],$$

$$\sigma_{22}^{\infty} - i\sigma_{12}^{\infty} = 2\mu_1 i [(\alpha_1 - 1)(A_1 + 2A_2z) + i(\overline{B_1} + 2\overline{B_2}\overline{z}) + \beta_1 \lim_{|z| \to \infty} \left(\frac{\overline{A_1} + 2\overline{A_2}(\overline{z} - z)}{(\overline{A_1} + 2\overline{A_2}\overline{z})^2}\right)]$$

Note that the Piola stress at the origin ($z = x_1 + ix_2$, $x_1 = x_2 = 0$, center of circular inhomogeneity) can be calculated as

$$-\sigma_{21} + i\sigma_{11} = 2\mu_2 i [i(\alpha_2 - 1)C_1 + \overline{D_1} + i\beta_2 \frac{1}{\overline{C_1}}],$$

$$\sigma_{22} - i\sigma_{12} = 2\mu_2 i [(\alpha_2 - 1)C_1 + i\overline{D_1} + \beta_2 \frac{1}{\overline{C_1}}]$$

and, clearly, the corresponding deformation is, from (3.52), $w_2 = 0$.

3.7 CONCLUSIONS

In this Chapter, we consider an inhomogeneity-matrix system from a particular class of compressible hyperelastic materials of harmonic-type undergoing finite plane deformations. We obtain the complete solution for a perfectly bonded elliptic inhomogeneity when the system is subjected to general classes of remote Piola stresses characterized by stress functions described by general polynomials of degree $n \ge 1$. In addition, from the general solution, we also derived the complete set of solutions for a perfectly bonded *circular* inhomogeneity-matrix system (a particular class of an elliptic inhomogeneity) subjected to the same class of remote stresses. The solutions presented here are extremely important in that they essentially lead to solutions of problems in which the inhomogeneity-matrix system is subjected to a wide class of remote loading conditions and so can accommodate a correspondingly large set of physically relevant problems. We note, in particular, the generality of our results and that existing results in the literature are obtained as special cases of the solutions derived here. For example, the results found in Ru et al (2005) for the el-

liptic inhomogeneity-matrix system subjected to uniform remote loading are obtained here as special cases (see Appendix). In addition, concerning for the case of the circular inhomogeneity problems, the general solutions derived here further reduce to the corresponding results presented in Ogden & Isherwood (1978) as far as the same conditions are applied.
Chapter 4 ESHELBY'S CONJECTURE

4.1 INTRODUCTION

It is well-known that 'peak' in the stress distribution inside an inhomogeneity are responsible for the failure of the corresponding fiber and ultimately the entire composite. It is of interest, therefore, to design an inhomogeneity in which the stress distribution remains uniform. One of the most celebrated results in solid mechanics is concerned specifically with this problem. This result is known as Eshelby's conjecture.

Eshelby's conjecture in linear plane and anti-plane elasticity states that the stress distribution inside an inhomogeneity, perfectly bonded to an infinite isotropic elastic medium, subjected at infinity to a uniform stress field, is uniform if and only if the inhomogeneity is elliptical (Sendeckyj, 1970; Ru & Schiavone, 1996). In Ru et al (2005), the authors examined one side of Eshelby's conjecture for the plane deformations of the same class of compressible hyperelastic materials of harmonic-type considered in this thesis. More specifically, it was shown that, under the analogous constraints of perfect bonding and uniform remote Piola stress, if the Piola stress within the inhomogeneity is uniform, the inhomogeneity is necessarily elliptical except for the special case when the complex potentials at the remote boundary and inside the inhomogeneity takes a particular restricted form:

$$\phi_1(z) = Az, \ \psi_1(z) = Bz, \ |z| \to \infty,$$

$$\phi_2(z) = Cz, \psi_2(z) = Bz$$
, inside the inhomogeneity,

where

$$A = C\frac{\mu_2}{\mu_1} + \left(1 - \frac{\mu_2}{\mu_1}\right) \left[\alpha_2 C + \frac{\beta_2}{\overline{C}}\right],$$
$$\left(1 - \frac{\mu_2}{\mu_1}\right) B = 0, \quad \alpha_1 \overline{A} + \frac{\beta_1}{\overline{A}} = \alpha_2 \overline{C} + \frac{\beta_2}{\overline{C}}),$$

in which case the shape of the inhomogeneity can be taken as arbitrary. This latter result is extremely interesting in that it does not arise in the analogous results from linear elasticity and can be attributed to the particular class of materials being considered here. It is of interest, therefore, to examine if the converse result (i.e. the 'other side' of Eshelby's conjecture) holds true for this class of materials. Namely, if the Piola stress distribution within an elliptic inhomogeneity is necessarily uniform when the same inhomogeneity-matrix system is subjected to finite plane deformations and uniform remote loading (Kim et al, 2007).

To examine Eshelby's conjecture, we begin by assuming that, at infinity, the Piola stress is uniform so that

$$\phi_1(z) = Az + O_{(1)}, \quad \psi_1(z) = Bz + O_{(2)}, \quad |z| \to \infty,$$

where A, B, $O_{(1)}$ and $O_{(2)}$ are prescribed complex constants.

4.2 COMPLEX VARIABLE FORMULATION

As a consequence of (2.27-28) and the remote loading condition, we seek analytic functions $\phi_{\gamma}(z)$ and $\psi_{\gamma}(z)$, $\gamma = 1, 2$ in S_1 and S_2 , respectively, such that

$$\phi_1(z) = K\phi_2(z) + iS\overline{\psi_2(z)} + \frac{S\beta_2 z}{\overline{\phi_2'(z)}}, \quad z \in \Gamma,$$
(4.1)

$$\psi_{1}(z) = i\left(-\alpha_{1}\overline{\phi_{1}(z)} - \frac{\beta_{1}\overline{z}}{\phi_{1}'(z)} + \alpha_{2}\overline{\phi_{2}(z)} + \frac{\beta_{2}\overline{z}}{\phi_{2}'(z)}\right) + \psi_{2}(z), \quad z \in \Gamma,$$

$$\phi_1(z) = Az + O_{(1)}, \quad \psi_1(z) = Bz + O_{(2)}, \quad |z| \to \infty,$$
(4.2)

where

$$K = \frac{\mu_2}{\mu_1} + \alpha_2 \left(1 - \frac{\mu_2}{\mu_1} \right), \quad S = 1 - \frac{\mu_2}{\mu_1},$$

For the class of harmornic materials considered here, we have from (2.21-22):

$$1 > \alpha_{\gamma} \ge 1/2, \ \beta_{\gamma} > 0, \ \alpha_{\gamma} + \beta_{\gamma} = 1, \ \mu_{\gamma} > 0,$$

$$\gamma = 1, 2 \text{ (no sum over repeated indices).}$$

$$(4.3)$$

It is well-known that the finite region bounded by an ellipse can be mapped conformally into a circle, but the corresponding transformation is complicated and inconvenient Muskhelishvili (1953). Instead, consider the enclosed region S_2 to be cut along the segment $D = \{(x_1, 0) : -2R \le x_1 \le 2R\}$ connecting the foci. This cut may be thought of as an ellipse, which is confocal with Γ but whose minor axis is zero. Hence the cut region in S_2 may be thought of as the limiting case of a region between two confocal ellipses. In fact, Muskhelishvili (1953),

$$z = \omega\left(\xi\right) = R\left(\xi + \frac{1}{\xi}\right), \quad R > 0, \tag{4.4}$$

will map the ellipse

$$x_1 = R\left(\rho + \frac{1}{\rho}\right)\cos\theta, \quad x_2 = R\left(\rho - \frac{1}{\rho}\right)\sin\theta,$$
 (4.5)

where R, $\rho > 0$ and $\theta \in [0, 2\pi]$ and its exterior region in the complex z - plane, onto and outside, respectively, the circle of radius ρ in the complex $\xi - plane$. Hence, using the transformation (4.4), the region S_1 is transformed into the exterior of the circle with radius R_* and the region $S_2 \setminus D$ into the ring $1 < |\xi| < R_*$ in the $\xi - plane$. The segment D and the boundary Γ are mapped onto the circles $|\xi| = 1$ and $|\xi| = R_*$, respectively, where,

$$R_* = \frac{a + \sqrt{a^2 - 4R^2}}{2R} > 1, \quad a = R\left(R_* + \frac{1}{R_*}\right).$$



Figure 6: The Conformal Mapping from z - plane to $\xi - plane$

Since $\phi_2(z)$ is analytic in S_2 , we must have

$$\phi_2(z) = \phi_2(\overline{z}), \quad \psi_2(z) = \psi_2(\overline{z}) \quad z \in D.$$
(4.6)

For convenience, we write $\phi_{\gamma}(\omega(\xi)) = \phi_{\gamma}(\xi)$, $\gamma = 1, 2$ and similarly for the functions ψ_{γ} so that, in the $\xi - plane$, condition (4.6) becomes

$$\phi_2\left(\xi\right) = \phi_2\left(\overline{\xi}\right), \quad \psi_2\left(\xi\right) = \psi_2\left(\overline{\xi}\right), \quad \forall \ \xi : |\xi| = 1.$$

Thus, in the $\xi - plane$, we seek the solution $\phi_{\gamma}(\xi)$, $\psi_{\gamma}(\xi)$, $\gamma = 1, 2$ in the regions $|\xi| > R_*$ and $1 < |\xi| < R_*$, respectively, such that

$$\phi_1(\xi) = K\phi_2(\xi) + iS\overline{\psi_2(\xi)} + \frac{S\beta_2\omega(\xi)\overline{\omega'(\xi)}}{\overline{\phi_2'(\xi)}}, \quad |\xi| = R_*,$$

$$(4.7)$$

$$\psi_{1}\left(\xi\right) = -i\alpha_{1}\overline{\phi_{1}\left(\xi\right)} - \frac{i\beta_{1}\overline{\omega\left(\xi\right)}}{\phi_{1}'\left(\xi\right)} + i\alpha_{2}\overline{\phi_{2}'\left(\xi\right)} + \frac{i\beta_{2}\overline{\omega\left(\xi\right)}}{\phi_{2}'\left(\xi\right)} + \psi_{2}\left(\xi\right), \quad |\xi| = R_{*},$$

$$\therefore \quad \phi_{\gamma}'(z) = \frac{d\phi(z)}{dz} = \frac{d\phi(\xi)}{d\xi} \frac{d\xi}{dz} = \frac{\phi'(\xi)}{\omega'(\xi)}, \quad \overline{\phi_{\gamma}'(z)} = \overline{\left(\frac{\phi'(\xi)}{\omega'(\xi)}\right)} = \overline{\frac{\phi'(\xi)}{\omega'(\xi)}},$$

$$\phi_2\left(\xi\right) = \phi_2\left(\overline{\xi}\right), \quad \psi_2\left(\xi\right) = \psi_2\left(\overline{\xi}\right), \quad \forall \ \xi : |\xi| = 1, \tag{4.8}$$

$$\phi_1(\xi) = AR\xi + O_{(1)}, \quad \psi_1(\xi) = BR\xi + O_{(2)}, \quad |\xi| \to \infty.$$
(4.9)

Since $\phi_2(\xi)$ and $\psi_2(\xi)$ are analytic in the ring $1 < |\xi| < R_*$, they have Laurent series representations

$$\phi_2\left(\xi\right) = \sum_{n=-\infty}^{\infty} C_n \xi^n, \quad \psi_2\left(\xi\right) = \sum_{n=-\infty}^{\infty} D_n \xi^n, \quad C_n, D_n \in \mathbb{C}.$$

Letting $\xi = \gamma e^{i\theta}$, from (3.8), we obtain Muskhelishvili (1953)

$$C_n = C_{-n}, \quad D_n = D_{-n},$$
 (4.10)

$$\phi_2(\xi) = C_o + \sum_{n=1}^{n=\infty} C_n\left(\xi^n + \frac{1}{\xi^n}\right), \quad \psi_2(\xi) = D_o + \sum_{n=1}^{n=\infty} D_n\left(\xi^n + \frac{1}{\xi^n}\right), \quad 1 < |\xi| < R_*.$$

Remark 6. Since, from Remark 1

$$\left|\phi_{\gamma}'\left(z\right)\right| = F'\left(I\right) \neq 0, \quad A_{1}, \ C_{1} \neq 0,$$

and

$$\omega'\left(\xi\right) \neq 0, \quad |\xi| > 1,$$

it is clear that we must have

$$\phi_2'\left(\xi\right) \neq 0 \iff \overline{\phi_2'\left(\xi\right)} \neq 0, \quad |\xi| > 1.$$

In other words, from (4.10), we must have that

$$\sum_{n=1}^{n=\infty} nC_n \left(\xi^{n-1} - \frac{1}{\xi^{n+1}}\right) \neq 0$$

$$\iff \overline{\sum_{n=1}^{n=\infty} nC_n \left(\xi^{n-1} - \frac{1}{\xi^{n+1}}\right)} \neq 0, \quad \forall \xi : |\xi| = R_*.$$

It follows from (4.10) and the interface condition (4.7) that we now have, for $|\xi| = R_*$

$$\phi_1(\xi) = K \left[C_o + \sum_{n=1}^{n=\infty} C_n \left(\xi^n + \frac{1}{\xi^n} \right) \right] + iS \left[D_o + \sum_{n=1}^{n=\infty} D_n \left(\xi^n + \frac{1}{\xi^n} \right) \right] \\ + \frac{S\beta_2 R \left(\xi + \frac{1}{\xi} \right) \overline{R \left(1 - \frac{1}{\xi^2} \right)}}{\sum_{n=1}^{n=\infty} n C_n \left(\xi^{n-1} - \frac{1}{\xi^{n+1}} \right)}.$$

Since $\xi \overline{\xi} = |\xi|^2 = R_*^2$ at the interface,

$$\phi_{1}(\xi) = K \left[C_{o} + \sum_{n=1}^{n=\infty} C_{n} \left(\xi^{n} + \frac{1}{\xi^{n}} \right) \right] + iS \left[\overline{D_{o}} + \sum_{n=1}^{n=\infty} \overline{D_{n}} \left(\frac{R_{*}^{2n}}{\xi^{n}} + \frac{\xi^{n}}{R_{*}^{2n}} \right) \right] + \frac{S\beta_{2}R^{2} \left(\xi + \frac{1}{\xi} \right) \left(1 - \frac{\xi^{2}}{R_{*}^{4}} \right)}{\sum_{n=1}^{n=\infty} n \overline{C_{n}} \left(\frac{R_{*}^{2(n-1)}}{\xi^{n-1}} - \frac{\xi^{n+1}}{R_{*}^{2(n+1)}} \right)}.$$
(4.11)

If we adopt ϕ_1 from (4.11) for $|\xi| > R_*$, at the remote boundary, we have

$$\phi_1(\xi) = K \left[C_o + \sum_{n=1}^{n=\infty} C_n \xi^n \right] + iS \left[\overline{D_o} + \sum_{n=1}^{n=\infty} \overline{D_n} \frac{\xi^n}{R_*^{2n}} \right] + \lim_{|\xi| \to \infty} \left(SR_{(n,\xi)} \right), \quad |\xi| \to \infty,$$
(4.12)

where

$$R_{(n,\xi)} = \frac{\beta_2 R^2 \left(\xi + \frac{1}{\xi}\right) \left(1 - \frac{\xi^2}{R_*^4}\right)}{\sum_{n=1}^{n=\infty} n \overline{C_n} \left(\frac{R_*^{2(n-1)}}{\xi^{n-1}} - \frac{\xi^{n+1}}{R_*^{2(n+1)}}\right)}$$

Also from (4.9),

$$\phi_1(\xi) = AR\xi + O_{(1)}, \quad |\xi| \to \infty.$$
 (4.13)

Therefore, it follows from (4.12-13) that

$$AR\xi + O_{(1)} = K \left[C_o + \sum_{n=1}^{n=\infty} C_n \xi^n \right] + iS \left[\overline{D_o} + \sum_{n=1}^{n=\infty} \overline{D_n} \frac{\xi^n}{R_*^{2n}} \right] + \lim_{|\xi| \to \infty} \left(SR_{(n,\xi)} \right), \quad |\xi| \to \infty.$$
(4.14)

Equating coefficients of ξ , we require from (4.14) that

$$AR\xi + O_{(1)} = K \left[C_o + C_1 \xi \right] + iS \left[\overline{D_o} + \overline{D_1} \frac{\xi}{R_*^2} \right] + \lim_{|\xi| \to \infty} \left(SR_{(n,\xi)} \right), \tag{4.15}$$

$$\sum_{n=2}^{n=\infty} \left(KC_n + \frac{iS\overline{D_n}}{R_*^{2n}} \right) \xi^n = 0, \quad |\xi| \to \infty.$$
(4.16)

From (4.11-12) and (4.16), for

$$\phi_{1}\left(\xi\right) = KC_{o} + iS\overline{D_{o}} + \left(KC_{1} + \frac{iS\overline{D_{1}}}{R_{*}^{2}}\right)\xi + \sum_{n=1}^{\infty} \left[\left(KC_{n} + iS\overline{D_{n}}R_{*}^{2n}\right)\frac{1}{\xi^{n}}\right] + SR_{(n,\xi)}, \quad |\xi| > R_{*}$$

$$(4.17)$$

Remark 7. As a direct consequence of Remark 6, we have that

$$KC_{1} + \frac{iS\overline{D_{1}}}{R_{*}^{2}} + \sum_{n=1}^{\infty} \left[-n \left(KC_{n} + iS\overline{D_{n}}R_{*}^{2n} \right) \frac{1}{\xi^{n+1}} \right] + SR'(n,\xi) \neq 0$$
$$\forall \xi : |\xi| > R_{*}.$$

Similarly, from the interface condition (4.7) together with (4.10) and (3.12), it follows for $|\xi| = R_*,$

$$\psi_1(\xi) = -i\alpha_1 \left[K\overline{C_o} - iSD_o + \left(K\overline{C_1} - \frac{iSD_1}{R_*^2} \right) \frac{R_*^2}{\xi} + \sum_{n=1}^{\infty} \left[\left(K\overline{C_n} - iSD_n R_*^{2n} \right) \frac{\xi^n}{R_*^{2n}} \right] + S\overline{R_{(n,\xi)}} \right]$$

$$-\frac{i\beta_{1}R^{2}\left(\frac{R_{*}^{2}}{\xi}+\frac{\xi}{R_{*}^{2}}\right)\left(1-\frac{1}{\xi^{2}}\right)}{KC_{1}+\frac{iS\overline{D_{1}}}{R_{*}^{2}}+\sum_{n=1}^{\infty}\left[-n\left(KC_{n}+iS\overline{D_{n}}R_{*}^{2n}\right)\frac{1}{\xi^{n+1}}\right]+SR'\left(n,\xi\right)}+D_{o}$$
(4.18)

$$+i\alpha_{2}\left[\overline{C_{o}}+\sum_{n=1}^{\infty}\overline{C_{n}}\left(\frac{R_{*}^{2n}}{\xi^{n}}+\frac{\xi^{n}}{R_{*}^{2n}}\right)\right]+\frac{i\beta_{2}R^{2}\left(\frac{R_{*}^{2}}{\xi}+\frac{\xi}{R_{*}^{2}}\right)\left(1-\frac{1}{\xi^{2}}\right)}{\sum_{n=1}^{\infty}nC_{n}\left(\xi^{n-1}-\frac{1}{\xi^{n+1}}\right)}+\sum_{n=1}^{\infty}D_{n}\left(\xi^{n}+\frac{1}{\xi^{n}}\right),$$

where from (4.12)

$$\overline{R_{(n,\xi)}} = \frac{\beta_2 R^2 \left(\frac{R_*^2}{\xi} + \frac{\xi}{R_*^2}\right) \left(1 - \frac{1}{\xi^2}\right)}{\sum_{n=1}^{\infty} n C_n \left(\xi^{n-1} - \frac{1}{\xi^{n+1}}\right)}.$$

Thus, at the remote boundary, we have from (4.18)

$$\psi_1(\xi) = -i\alpha_1 \left[K\overline{C_o} - iSD_o + \sum_{n=1}^{\infty} \left[\left(\frac{K\overline{C_n}}{R_*^{2n}} - iSD_n \right) \xi^n \right] + \lim_{|\xi| \to \infty} \left(S\overline{R_{(n,\xi)}} \right) \right] - \lim_{|\xi| \to \infty} \left(\mathcal{Q}_{(n,\xi)} \right)$$

$$+i\alpha_2\left[\overline{C_o} + \sum_{n=1}^{\infty} \frac{\overline{C_n}}{R_*^{2n}} \xi^n\right] + \lim_{|\xi| \to \infty} \left(i\overline{R_{(n,\xi)}}\right) + D_o + \sum_{n=1}^{\infty} D_n \xi^n, \quad |\xi| \to \infty,$$
(4.19)

where

$$Q_{(n,\xi)} = \frac{i\beta_1 R^2 \left(\frac{R_\star^2}{\xi} + \frac{\xi}{R_\star^2}\right) \left(1 - \frac{1}{\xi^2}\right)}{KC_1 + \frac{iS\overline{D_1}}{R_\star^2} - \sum_{n=1}^{\infty} \left[n \left(KC_n + iS\overline{D_n}R_\star^{2n}\right)\frac{1}{\xi^{n+1}}\right] + SR'(n,\xi)}$$

From (4.9), ψ_1 must also satisfy the asymptotic condition

$$\psi_1(\xi) = BR\xi + O_{(2)}, \quad |\xi| \to \infty.$$

As in (4.14), we require from (4.19) that

$$BR\xi + O_{(2)} = -i\alpha_1 \left[K\overline{C_o} - iSD_o + \sum_{n=1}^{\infty} \left[\left(\frac{K\overline{C_n}}{R_*^{2n}} - iSD_n \right) \xi^n \right] + \lim_{|\xi| \to \infty} \left(S\overline{R_{(n,\xi)}} \right) \right]$$
$$- \lim_{|\xi| \to \infty} \left(\mathcal{Q}_{(n,\xi)} \right) + i\alpha_2 \left[\overline{C_o} + \sum_{n=1}^{\infty} \frac{\overline{C_n}}{R_*^{2n}} \xi^n \right] + \lim_{|\xi| \to \infty} \left(i\overline{R_{(n,\xi)}} \right) + D_o + \sum_{n=1}^{\infty} D_n \xi^n, \quad |\xi| \to \infty.$$

Therefore, equating coefficients of ξ , we have

$$BR\xi + O_{(2)} = -i\alpha_1 \left[K\overline{C_o} - iSD_o + \left(\frac{K\overline{C_1}}{R_*^2} - iSD_1 \right) \xi \right]$$

$$+\lim_{|\xi|\to\infty} \left[i\overline{R_{(n,\xi)}}\left(1-\alpha_1S\right)\right] + -\lim_{|\xi|\to\infty} \left(\mathcal{Q}_{(n,\xi)}\right) + i\alpha_2\left[\overline{C_o} + \overline{\frac{C_1}{R_*^2}}\xi\right] + D_o + D_1\xi, \quad (4.20)$$

$$\sum_{n=2}^{\infty} \left[\left\{ \frac{i\overline{C_n}}{R_*^{2n}} \left(-\alpha_1 K + \alpha_2 \right) + D_n \left(1 - \alpha_1 S \right) \right\} \xi^n \right] = 0.$$
(4.21)

Recall (4.16). Together with (4.21), since $\xi = \rho e^{i\theta}$, $\rho > 0$, we derive the following equations for $C_n, D_n, n = 2, 3...$

$$KC_n + \frac{iS\overline{D_n}}{R_*^{2n}} = 0, (4.22)$$

$$\frac{i(\alpha_1 K - \alpha_2) C_n}{R_*^{2n}} + \overline{D_n} (1 - \alpha_1 S) = 0.$$
(4.23)

Also from (4.3), for the class of harmornic materials discussed here

$$K, \ (1 - \alpha_1 S) \neq 0. \tag{4.24}$$

We consider the following cases:

4.2.1 Case 1 : When $S\overline{D_n} = 0$

From (4.22) that we must have $KC_n = 0, n \ge 2$, and hence from (4.24)

$$C_n = 0, \quad n \ge 2,$$

Then, it follows from (4.23-4.24)

$$D_n = 0, n \ge 2.$$

4.2.2 Case 2 : When $(\alpha_1 K - \alpha_2) C_n = 0$

From (4.23) together with (4.24)

$$D_n = 0, \quad n \ge 2,$$

Then, (4.22) and again (4.24) yield that

$$C_n = 0, \quad n \ge 2.$$

4.2.3 Case 3 : When $S\overline{D_n} = (\alpha_1 K - \alpha_2) C_n = 0$

In this case Equations (4.22-24) yield only that

$$C_n = D_n = 0, \quad n \ge 2.$$

4.2.4 Case 4 : When $S\overline{D_n}$, $(\alpha_1 K - \alpha_2) C_n \neq 0$

For the case when both $S\overline{D_n}$ and $(\alpha_1 K - \alpha_2) C_n$ are not equal to zero, we start with the assumption that

$$C_n, D_n \neq 0.$$

By letting $C_n = \operatorname{Re}(C_n) + i \operatorname{Im}(C_n)$, $D_n = \operatorname{Re}(D) + i \operatorname{Im}(D_n)$, we solve Equations (4.22-23) and obtain

$$\left[K\left(1-\alpha_{1}S\right)+\frac{\left(\alpha_{1}K-\alpha_{2}\right)S}{R_{*}^{4n}}\right]C_{n} = 0, \qquad (4.25)$$

$$\left[K\left(1-\alpha_1 S\right)+\frac{\left(\alpha_1 K-\alpha_2\right)S}{R_*^{4n}}\right]D_n = 0.$$

Suppose that $C_n, D_n \neq 0$ then we must have

$$\left[K(1 - \alpha_1 S) + \frac{(\alpha_1 K - \alpha_2) S}{R_*^{4n}}\right] = 0,$$
(4.26)

otherwise, we have the same result as in Cases. 1-3, that is, $C_n = D_n = 0$, $n \ge 2$. From (4.2), rewriting (4.26) in terms of α_{γ} and $\mu_{\gamma} > 0$ $\gamma = 1, 2$, we find that (4.26) reduces to

$$\left(\frac{\mu_2}{\mu_1}\right)^2 \alpha_1 \left(\alpha_2 - 1\right) \left(\frac{1}{R_*^{4n}} - 1\right) + \left(\frac{\mu_2}{\mu_1}\right) \left[1 + (\alpha_1 + \alpha_2 - 2\alpha_1\alpha_2) \left(\frac{1}{R_*^{4n}} - 1\right)\right] + \alpha_2 \left(\alpha_1 - 1\right) \left(\frac{1}{R_*^{4n}} - 1\right) = 0$$

The two roots $\left(\frac{\mu_2}{\mu_1}\right)_1$ and $\left(\frac{\mu_2}{\mu_1}\right)_2$ are easily found to be:

$$(\frac{\mu_2}{\mu_1})_1 = \frac{-\mathcal{B} + \sqrt{\mathcal{B}^2 - 4\mathcal{AC}}}{2\mathcal{A}},$$
 (4.27)

$$(\frac{\mu_2}{\mu_1})_2 = \frac{-\mathcal{B} - \sqrt{\mathcal{B}^2 - 4\mathcal{A}C}}{2\mathcal{A}},$$

where

$$\mathcal{A} = \alpha_1 (\alpha_2 - 1) \left(\frac{1}{R_*^{4n}} - 1 \right), \ \mathcal{B} = 1 + (\alpha_1 + \alpha_2 - 2\alpha_1 \alpha_2) \left(\frac{1}{R_*^{4n}} - 1 \right),$$

$$\mathcal{C} = \alpha_2 \left(\alpha_1 - 1 \right) \left(\frac{1}{R_*^{4n}} - 1 \right).$$

It is not difficult to show that (4.27) means that, (see Figures. 7 and 8)

$$(\frac{\mu_2}{\mu_1})_1 < 0, \quad (\frac{\mu_2}{\mu_1})_2 < 0.$$



Figure 7: First Root (equation 4.27): $(\frac{\mu_2}{\mu_1})_1$, where $(1 > \alpha_{\gamma} \ge 1/2, \gamma = 1, 2)$



Figure 8: Second Root (equation 4.27): $(\frac{\mu_2}{\mu_1})_2$, where $(1 > \alpha_{\gamma} \ge 1/2, \gamma = 1, 2)$

This contradicts the assumptions made on the elastic constants for this class of materials, namely Ru (2002):

$$\begin{split} 1 &> & \alpha_{\gamma} \geq 1/2, \ \beta_{\gamma} > 0, \ \alpha_{\gamma} + \beta_{\gamma} = 1, \ \mu_{\gamma} > 0, \\ \gamma &= & 1,2 \ \text{(no sum over repeated indices).} \end{split}$$

Consequently, (4.26) cannot be true and

$$\left[K(1 - \alpha_1 S) + \frac{(\alpha_1 K - \alpha_2) S}{R_*^{2n}}\right] \neq 0.$$
(4.28)

From (4.25), (4.28) now implies that $C_n = D_n = 0$, $n \ge 2$.

In conclusion, from (4.10), in the ξ – plane

$$\phi_2(\xi) = C_o + C_1\left(\xi + \frac{1}{\xi}\right), \quad \psi_2(\xi) = D_o + D_1\left(\xi + \frac{1}{\xi}\right), \quad 1 < |\xi| < R_*,$$

:
$$\sum_{n=1}^{\infty} C_n = C_1$$
, $\sum_{n=1}^{\infty} D_n = D_1$, (: $C_n = D_n = 0$, $n \ge 2$),

and through the mapping function (4.4), we have that in the z - plane

$$\phi_2(z) = C_o + \frac{C_1}{R}z, \quad \psi_2(z) = D_o + \frac{D_1}{R}z, \quad z \in S_2,$$

$$\therefore \quad \frac{z}{R} = \left(\xi + \frac{1}{\xi}\right),$$

which implies from (2.18) and (2.28) that the state of (Piola) stress inside the inhomogeneity is necessarily uniform. In fact, from (2.26), we derive the Piola stress function χ inside the inhomogeneity:

$$\chi_2(z) = 2\mu_2 i[(\alpha_2 - 1)\left(C_o + \frac{C_1}{R}z\right) + i\overline{\left(D_o + \frac{D_1}{R}z\right)} + \frac{\beta_2 Rz}{\overline{C_1}}],$$

and from (2.18), the Piola stress inside inhomogeneity can be calculated as

$$-\sigma_{21} + i\sigma_{11} = (\chi_2)_{,2} = 2\mu_2 i [(\alpha_2 - 1)\frac{iC_1}{R} + \frac{\overline{D_1}}{R} + \frac{i\beta_2 R}{\overline{C_1}}],$$

$$\sigma_{22} - i\sigma_{12} = (\chi_2)_{,1} = 2\mu_2 i [(\alpha_2 - 1)\frac{C_1}{R} + \frac{i\overline{D_1}}{R} + \frac{\beta_2 R}{\overline{C_1}}],$$

which, clearly, is uniform.

We have just proved the following result concerning necessity in Eshelby's conjecture for this class of harmonic materials:

Theorem 1. For this class of harmonic materials, if an inhomogeneity-matrix system is subjected to finite plane deformations and uniform remote loading, the Piola stress distribution within an elliptic inhomogeneity is necessarily uniform.

4.3 SOLUTION FOR THE UNIFORM INTERIOR STRESS CASE

We can, in fact, obtain the complete solution corresponding to the elliptic inhomogeneitymatrix system under the assumption of uniform remote loading. In fact, from the results obtained in the previous section, the analytic functions ϕ_2 and ψ_2 take the form

$$\phi_2(z) = C_o + \frac{C_1}{R}z, \quad \psi_2(z) = D_o + \frac{D_1}{R}z,$$
(4.29)

where $C_o, C_1, D_o, D_1 \in \mathbb{C}$. From (4.4) and (4.7), on $|\xi| = R_{*}$,

$$\phi_1(\xi) = K \left[C_o + C_1 \left(\xi + \frac{1}{\xi} \right) \right] + iS \left[\overline{D_o} + \overline{D_1} \left(\frac{R_*^2}{\xi} + \frac{\xi}{R_*^2} \right) \right] + \frac{S\beta_2 R^2 \left(\xi + \frac{1}{\xi} \right)}{\overline{C_1}},$$
(4.30)

where, from Remark 1, $C_1, \overline{C_1} \neq 0$. Also from (4.9)

$$\phi_1(\xi) = AR\xi + O_{(1)}, \quad |\xi| \to \infty.$$
 (4.31)

Then, without loss of generality, the Laurent series expansion for ϕ_1 takes the form:

$$\phi_1(\xi) = \sum_{n=-\infty}^{n=1} X_n \xi^n, \quad |\xi| > R_{*}.$$
(4.32)

It follows from (4.30-32) that

$$\phi_1(\xi) = X_o + K_{-1}\frac{1}{\xi} + X_1\xi, \quad K_{-2} = K_{-3} = K_{-4} = \dots = 0, \quad |\xi| > R_*, \quad (4.33)$$

79

where

$$X_o = O_{(1)} = KC_o + iS\overline{D_o},$$

$$X_1 = AR = KC_1 + \frac{iS\overline{D_1}}{R_*^2} + \frac{S\beta_2 R^2}{\overline{C_1}},$$

$$K_{-1} = KC_1 + iS\overline{D_1}R_*^2 + \frac{S\beta_2R^2}{\overline{C_1}}.$$

In the view of Remark 7, we must have from (4.33) that

$$K_1 \neq \frac{X_{-1}}{\xi^2}, \quad \forall \xi : |\xi| > R_*.$$

From the interface condition (4.7), (4.29) and (4.33), it follows that for $|\xi| = R_*$,

$$\psi_{1}(\xi) = -i\alpha_{1} \left[\overline{X_{o}} + \overline{X_{-1}} \frac{\xi}{R_{*}^{2}} + \overline{X_{1}} \frac{R_{*}^{2}}{\xi} \right] - \frac{i\beta_{1}R^{2} \left(\frac{R_{*}^{2}}{\xi} + \frac{\xi}{R_{*}^{2}}\right) \left(1 - \frac{1}{\xi^{2}}\right)}{X_{1} - \frac{X_{-1}}{\xi^{2}}}$$
(4.34)

$$+i\alpha_2\left[\overline{C_o}+\overline{C_1}\left(\frac{R_*^2}{\xi}+\frac{\xi}{R_*^2}\right)\right]+\frac{i\beta_2R^2\left(\frac{R_*^2}{\xi}+\frac{\xi}{R_*^2}\right)}{C_1}+D_o+D_1\left(\xi+\frac{1}{\xi}\right).$$

From (4.9), ψ_1 must also satisfy the asymptotic condition

$$\psi_1(\xi) = BR\xi + O_{(2)}, \quad |\xi| \to \infty.$$

Hence, if we adopt (4.34) for the region $|\xi| > R_*$, we require that

$$-i\alpha_1 \left[\overline{X_o} + \overline{X_{-1}} \frac{\xi}{R_*^2} \right] - \frac{i\beta_1 R^2}{X_1 R_*^2} \xi + i\alpha_2 \left[\overline{C_o} + \overline{\frac{C_1}{R_*^2}} \xi \right] + \frac{i\beta_2 R^2}{C_1 R_*^2} \xi + D_o + D_1 \xi = BR\xi + O_{(2)}, \quad |\xi| \rightarrow \infty$$

Thus, since $X_1 = AR$, we require

$$O_{(2)} = -i\alpha_1 \overline{X_o} + i\alpha_2 \overline{C_o} + D_o$$

$$-\frac{i\alpha_1\overline{K_{-1}}}{R_*^2} - \frac{i\beta_1R}{AR_*^2} + \frac{i\alpha_2\overline{C_1}}{R_*^2} + \frac{i\beta_2R^2}{C_1R_*^2} + D_1 = BR.$$

SUMMARY

The complete solution corresponding to the elliptic inhomogeneity-matrix system subjected to uniform remote loading is given by

$$\phi_2(z) = C_o + \frac{C_1}{R}z, \quad \psi_2(z) = D_o + \frac{D_1}{R}z, \quad |z| \in S_2,$$
 (4.35)

$$\phi_1(\xi) = K_o + K_{-1}\frac{1}{\xi} + K_1\xi, \quad K_o = O_{(1)}, \quad K_1 = AR, \quad |\xi| > R_*$$
(4.36)

and

$$\psi_{1}(\xi) = -i\alpha_{1} \left[\overline{K_{o}} + \overline{K_{-1}} \frac{\xi}{R_{*}^{2}} + \overline{K_{1}} \frac{R_{*}^{2}}{\xi} \right] - \frac{i\beta_{1}R^{2} \left(\frac{R_{*}^{2}}{\xi} + \frac{\xi}{R_{*}^{2}} \right) \left(1 - \frac{1}{\xi^{2}} \right)}{K_{1} - \frac{K_{-1}}{\xi^{2}}}$$

$$+ i\alpha_{2} \left[\overline{C_{o}} + \overline{C_{1}} \left(\frac{R_{*}^{2}}{\xi} + \frac{\xi}{R_{*}^{2}} \right) \right] + \frac{i\beta_{2}R^{2} \left(\frac{R_{*}^{2}}{\xi} + \frac{\xi}{R_{*}^{2}} \right)}{C_{1}} + D_{o} + D_{1} \left(\xi + \frac{1}{\xi} \right).$$

$$(4.37)$$

where $|\xi| > R_*$. Here, the complex constants C_o, C_1, D_o and D_1 are determined by the equations

$$\begin{aligned} KC_o + iS\overline{D_o}, &= K_o = O_{(1)}, \\ -i\alpha_1\overline{K_o} + i\alpha_2\overline{C_o} + D_o &= O_{(2)}, \end{aligned}$$

$$AR = KC_1 + \frac{iS\overline{D_1}}{R_*^2} + \frac{S\beta_2 R^2}{\overline{C_1}},$$
 (4.38)

$$X_{-1} = KC_1 + iS\overline{D_1}R_*^2 + \frac{S\beta_2R^2}{\overline{C_1}},$$

$$BR = -\frac{i\alpha_1 \overline{K_{-1}}}{R_*^2} - \frac{i\beta_1 R}{AR_*^2} + \frac{i\alpha_2 \overline{C_1}}{R_*^2} + \frac{i\beta_2 R^2}{C_1 R_*^2} + D_1.$$

where from Remarks. 6 and 7

$$\begin{array}{rcl} C_1 & \neq & 0 \Longleftrightarrow \overline{C_1} \neq 0, \\ \\ X_1 & \neq & \frac{X_{-1}}{\xi^2}, \ \forall \xi : |\xi| > R_* \end{array}$$

We note that the results obtained in Ru et al (2005) arise as a special case of those obtained here (see Appendix).

4.4 FURTHER DISCUSSION

From the results obtained in Section 4.3, we easily obtain the following important properties of this inhomogeneity-matrix system. Since, these properties will provide the actual response of the elliptic inhomogeneity-matrix system under the prescribed remote loading.

4.4.1 Expressions for I and J Inside the Inhomogeneity

From (2.1) and (2.6), R and J can be calculated as follow:

$$J = \det \mathbf{F} = -\operatorname{Im} \left[w_{,1} \overline{w}_{,2} \right] = y_{1,1} y_{2,2} - y_{1,2} y_{2,1}, \tag{4.39}$$

$$R = tr(\mathbf{F}\mathbf{F}^{T}) = |w_{,1}|^{2} + |w_{,2}|^{2} = y_{p,s}y_{p,s}, \dots p, s = 1, 2,$$
(4.40)

Hence, in the view of (2.1), Equation (2.5) yields

$$I = \sqrt{R + 2J} = \sqrt{tr(\mathbf{F}\mathbf{F}^{T}) + 2J},$$

$$= \sqrt{(y_{1,1} + y_{2,2})^{2} + (y_{1,2} - y_{2,1})^{2}},$$
(4.41)

where

$$(y_{1,1} + y_{2,2})^2 + (y_{1,2} - y_{2,1})^2 = 2(y_{1,1}y_{2,2} - y_{1,2}y_{2,1}) + y_{11}^2 + y_{22}^2 + y_{12}^2 + y_{21}^2.$$

Equation (2.25) and (4.35) give us that

$$w_2 = -i\alpha_2[C_o + \frac{C_1}{R}z] + \overline{D_o} + \frac{\overline{D_1}}{R}\overline{z} - \frac{i\beta_2Rz}{\overline{C_1}}.$$
(4.42)

Consequently, I takes the following form within the inhomogeneity

$$I = 2\left(\alpha_2 \left|\frac{C_1}{R}\right| + \frac{\beta_2}{\left|\frac{C_1}{R}\right|}\right),\tag{4.43}$$

and from Remark 6.

$$|\phi'_{2}(z)| = \left|\frac{C_{1}}{R}\right| \neq 0, \quad \forall \xi : |\xi| > 1,$$

$$I^{2} - 16\alpha_{2}\beta_{2} = 4\left(\alpha_{2}\left|\frac{C_{1}}{R}\right| - \frac{\beta_{2}}{\left|\frac{C_{1}}{R}\right|}\right)^{2} > 0,$$
(4.44)

which guarantees that F'(I), from (2.21) is well-defined.

4.4.2 Stress Distribution Inside Inhomogeneity

From (2.18), the Piola stresses inside the inhomogeneity and at the remote boundary are given, respectively, by

$$-\sigma_{21} + i\sigma_{11} = (\chi_2)_{,2}, \quad \sigma_{22} - i\sigma_{12} = (\chi_2)_{,1}, \quad (4.45)$$
$$-\sigma_{21}^{\infty} + i\sigma_{11}^{\infty} = (\chi_1)_{,2}, \quad \sigma_{22}^{\infty} - i\sigma_{12}^{\infty} = (\chi_1)_{,1}.$$

From Equation (2.26), (4.2) and (4.29) (or (4.35))

$$\chi_1(z) = 2\mu_1 i[(\alpha_1 - 1)(Az + O_{(1)}) + i(\overline{B}\overline{z} + \overline{O_{(2)}}) + \frac{\beta_1 z}{\overline{A}}],$$

$$\chi_2(z) = 2\mu_2 i[(\alpha_2 - 1)(\frac{C_1}{R}z + C_o) + i(\frac{\overline{D_1}}{R}\overline{z} + \overline{D_o}) + \frac{\beta_2 zR}{\overline{C_1}}].$$

Thus, the corresponding uniform Piola stress can be found from (4.45) as follows:

$$\begin{aligned} -\sigma_{21} + i\sigma_{11} &= 2\mu_2 i [i(\alpha_2 - 1)\frac{C_1}{R} + \frac{\overline{D_1}}{R} + \frac{i\beta_2 R}{\overline{C_1}}], \\ \sigma_{22} - i\sigma_{12} &= 2\mu_2 i [(\alpha_2 - 1)\frac{C_1}{R} + \frac{i\overline{D_1}}{R} + \frac{\beta_2 R}{\overline{C_1}}], \\ -\sigma_{21}^{\infty} + i\sigma_{11}^{\infty} &= 2\mu_1 i [i(\alpha_1 - 1)A + \overline{B} + \frac{i\beta_1}{\overline{A}}], \\ \sigma_{22}^{\infty} - i\sigma_{12}^{\infty} &= 2\mu_1 i [(\alpha_1 - 1)A + i\overline{B} + \frac{\beta_1}{\overline{A}}]. \end{aligned}$$

4.4.3 Deformed Contour

From the expression (4.42) of the deformation inside the inhomogeneity, it is clear that the curve Γ remains elliptical after deformation. Hence the deformed curve does not admit overlapping (see Figure 9).



Figure 9: Deformed Contour of the Elliptic Inhomogeneity. Here, we have taken $(A = 0.21i, B = -0.33, O_{(1)} = O_{(2)} = 0, \mu_1 = 0.27, \mu_2 = 0.3, \alpha_1 = \alpha_2 = \beta_1 = \beta_2 = 0.5)$

4.4.4 Admissibility Conditions

From (4.40) and (4.42-44), inside the inhomogeneity,

$$I^2 > 16\alpha_2\beta_2, \quad J > 0.$$
 (4.46)

As in Ru (2002), (4.46) again ensures that there is no overlapping of the deformation field (interior to the inhomogeneity). More precisely, (4.46) guaranties that the strain energy density function F'(I) from (2.21) remains a positive real valued function. In addition, J > 0 implies that relative orientation of the line elements is preserved under deformation (see, Gurtin, 1981; Chadwick, 1999; Fung, 1977). Henceforth, there is no overlapping after deformation.

4.5 CONCLUSIONS

We have show that, under the analogous constraints of perfect bonding and uniform remote (Piola) stress if the inhomogeneity-matrix system is subjected to finite plane deformations, the Piola stress distribution within the elliptic inhomogeneity is necessarily uniform. In addition, we have obtained the complete solution for this system for any uniform remote stress distribution.

Chapter 5 DESIGNING AN INHOMOGENEITY WITH UNIFORM INTERIOR STRESS

5.1 INTRODUCTION

Following our investigation of Eshelby's conjecture, it is of interest to examine whether we can design the remote loading and the shape of the inhomogeneity (fiber) to achieve a state of uniform stress inside the inhomogeneity.

In this Chapter, we address this issue and consider an inhomogeneity-matrix system in which the inhomogeneity is perfectly bonded to the surrounding matrix and subjected to some arbitrary nonuniform remote loading. In fact, we show that if the Piola stress distribution inside the inhomogeneity is uniform and the system is subjected to linear remote loading, the inhomogeneity is necessarily hypotrochoidal, with no exceptions (Kim & Schiavone, 2007c). We also obtain the complete solution of the inhomogeneity-matrix system in this case. In addition, we consider the more general case when the system is subjected to nonuniform remote loading characterized by stress functions of the form of n^{th} degree polynomials in the complex variable z describing the matrix. In this case, by finding the complete solution of the system, we show that we can again obtain uniform stress inside a hypotrochoidal inhomogeneity with n + 1 cusps. Finally, we illustrate our results with an example.

In the view of Remarks. 1 and 2, we first consider the simpler problem corresponding to the case in which the inhomogeneity-matrix system is subjected to linear remote loading characterized by analytic functions ϕ_1 and ψ_1 of the form:

$$\phi_1(z) = A_1 z + A_2 z^2, \quad \psi_1(z) = B_1 z + B_2 z^2, \quad |z| \to \infty, \tag{5.1}$$

where A_1, A_2, B_1 and B_2 are prescribed complex constants.

5.2 UNIFORM STRESS DISTRIBUTION INSIDE THE INHOMOGENEITY

Suppose, that the state of Piola stress inside the inhomogeneity is uniform, characterized by

$$\phi_2(z) = Cz, \quad \psi_2(z) = Dz, \quad z \in S_2,$$
(5.2)

where C and D are two complex constants to be determined.

First, we note that the interface conditions (2.27-28) can be written in the form

$$\phi_{1}\left(z\right) = K\phi_{2}\left(z\right) + i\mathscr{B}\overline{\psi_{2}\left(z\right)} + \frac{\mathscr{B}\beta_{2}z}{\overline{\phi_{2}'\left(z\right)}},$$

$$\psi_{1}(z) = i\left(-\alpha_{1}\overline{\phi_{1}(z)} - \frac{\beta_{1}\overline{z}}{\phi_{1}'(z)} + \alpha_{2}\overline{\phi_{2}(z)} + \frac{\beta_{2}\overline{z}}{\phi_{2}'(z)}\right) + \psi_{2}(z), \quad z \in \Gamma,$$
(5.3)

where the constants K and S are defined by $K = \frac{\mu_2}{\mu_1} + \alpha_2 \left(1 - \frac{\mu_2}{\mu_1}\right)$ and $S = \left(1 - \frac{\mu_2}{\mu_1}\right)$.

From (5.2) and the first of (5.3), we now have that

$$\phi_1(z) = KCz + i\mathscr{S}\overline{D}\overline{z} + \frac{\mathscr{S}\beta_2 z}{\overline{C}}, \quad z \in \Gamma.$$
(5.4)

Without loss of generality, we assume that there is a mapping function (Muskhelishvili, 1953)

$$z = \omega(\xi) = R\left(\xi + \frac{m_1}{\xi} + \frac{m_2}{\xi^2} + \cdots\right), \quad R \neq 0,$$
 (5.5)

which maps the region S_1 onto the exterior of the unit circle in the ξ – plane. Here, for simplicity, we have chosen the origin of the coordinate system such that no constant term appears on the right-hand side of (5.5). For convenience, we wirte $\phi_{\gamma}(z) = \phi_{\gamma}(\omega(\xi)) = \phi_{\gamma}(\xi)$, $\gamma = 1, 2$ and similarly for the function ψ_{γ} . From (5.1) and (5.5), the remote loading condition in the ξ – plane is taken to be

$$\phi_1(\xi) = A_1 R \xi + A_2 R^2 \left(\xi^2 + 2m_1\right), \quad |\xi| \to \infty.$$
(5.6)

From (5.4-5), in the ξ – *plane*, the interface condition becomes:

$$\phi_1(\xi) = \left(KC + \frac{\$\beta_2}{\overline{C}}\right)\omega(\xi) + i\$\overline{D}\overline{\omega(\xi)}, \quad |\xi| = 1.$$
(5.7)

In addition, expanding $\phi_1(\xi)$ in Laurent series outside the unit circle $|\xi| = 1$, and using (5.6), we obtain

$$\phi_1(\xi) = X_2\xi^2 + X_1\xi + X_o + X_{-1}\frac{1}{\xi} + X_{-2}\frac{1}{\xi^2} + \cdots, \qquad (5.8)$$

$$X_1 = A_1 R \neq 0, \ X_2 = A_2 R^2 \neq 0, \ X_o = 2A_2 R^2 m_1 \quad |\xi| > 1,$$

$$(:: A_1 R\xi + A_2 R^2 (\xi^2 + 2m_1) = X_2 \xi^2 + X_1 \xi + X_o, \quad |\xi| \to \infty).$$

Since $\xi \overline{\xi} = |\xi|^2 = 1$ at the interface, to satisfy the boundary condition (5.7), it is clear that, from (5.8), we must have, for $|\xi| = 1$,

$$\phi_1(\xi) = X_2\xi^2 + X_1\xi + X_o + X_{-1}\frac{1}{\xi} + X_{-2}\frac{1}{\xi^2} + \cdots$$
(5.9)

$$= \left(KC + \frac{\$\beta_2}{\overline{C}}\right) R\left(\xi + \frac{m_1}{\xi} + \frac{m_2}{\xi^2} + \cdots\right) + i\$\overline{D}R\left(\frac{1}{\xi} + m_1\xi + m_2\xi^2 + \cdots\right).$$

We consider the following cases:

5.2.1 Case 1: When SD = 0, $KC + \frac{S\beta_2}{\overline{C}} \neq 0$

In this case (5.9) yields:

$$\phi_1(\xi) = X_2 \xi^2 + X_1 \xi + X_o + X_{-1} \frac{1}{\xi} + X_{-2} \frac{1}{\xi^2} + \cdots$$
$$= \left(KC + \frac{\$\beta_2}{\overline{C}} \right) R \left(\xi + \frac{m_1}{\xi} + \frac{m_2}{\xi^2} + \cdots \right), \quad |\xi| = 1.$$

It is clear that this equation requires $X_2 = 0$. However, from (5.8), $X_2 \neq 0$, so that we cannot have solutions for this case (Note that the case $X_2 = 0$ corresponds to the case when the inhomogeneity- matrix system is subjected to uniform remote loading (see, for example, Ru et al, 2005).

5.2.2 Case 2: When
$$KC + \frac{\${}_{\beta_2}}{C} = 0, \$D \neq 0$$

In this case (5.9) yields

$$\phi_1(\xi) = X_2\xi^2 + X_1\xi + X_o + X_{-1}\frac{1}{\xi} + X_{-2}\frac{1}{\xi^2} + \cdots$$

$$= i \mathscr{S}\overline{D}R\left(\frac{1}{\xi} + m_1\xi + m_2\xi^2 + \cdots\right), \quad |\xi| = 1.$$

However, from (5.8), we must have

$$\begin{aligned} X_1 &= A_1 R = m_1 \neq 0, \\ X_o &= 2A_2 R^2 m_1 \neq 0, \quad A_1, A_2, R \neq 0, \end{aligned}$$

which again implies that there can be no solutions in this case.

5.2.3 Case 3: When
$$SD \neq 0$$
 and $KC + \frac{S\beta_2}{\overline{C}} \neq 0$

From (5.9), it is clear that we must have

$$m_1=m_3=m_4=\cdots=0.$$

Then, the mapping function (5.5) must take the form:

$$z = \omega(\xi) = R\left(\xi + \frac{m_2}{\xi^2}\right), \quad R, m_2 \neq 0.$$
 (5.10)

In other words, we have just proved the following result.

Theorem 2. If the inhomogeneity-matrix system is subjected to linear remote loading characterized by the quadratic stress functions $\phi_1(z) = A_1 z + A_2 z^2, \psi_1(z) =$ $B_1 z + B_2 z^2$ and the state of (Piola) stress inside the inhomogeneity is uniform, then the

inhomogeneity is a hypotrochoid with three cusps (Muskhelishvili, 1953) with mapping function

$$z = \omega(\xi) = R\left(\xi + \frac{m}{\xi^n}\right), \quad R > 0, \quad 0 < m \le \frac{1}{n}, \quad n = 2,$$

provided that

$$m, \quad \$D, \quad KC + \frac{\$\beta_2}{\overline{C}} \neq 0.$$



Figure 10: Deformed Contour of the Hypotrochoidal Inhomogeneity with 3 Cusps (n = 2). Here, we have taken ($A_1 = 1, A_3 = 0.17, \mu_1 = 0.27, \mu_2 = 0.3, \alpha_1 = \alpha_2 = \beta_1 = \beta_2 = 0.5$)

5.3 COMPLETE SOLUTION OF A HYPOTROCHOIDAL INHOMOGENEITY

We now proceed to find the complete solution in the case corresponding to Theorem 2. To this end, assume that Γ is a hypotrochoid with three cusps and, without loss of generality, assume that the coordinate system is chosen to be symmetric about the principal axes of the hypotrochoid so that the corresponding mapping function is given by:

$$z = \omega(\xi) = R\left(\xi + \frac{m}{\xi^2}\right), \quad R > 0, \quad 0 < m \le \frac{1}{2}.$$
 (5.11)

We seek the solution corresponding to the following uniform Piola stress distribution inside the inhomogeneity:

$$\phi_2(z) = Cz, \quad \psi_2(z) = Dz, \quad z \in S_2,$$
 (5.12)

where C and D are two complex constants to be determined. From (5.7) and (5.11), on $|\xi| = 1$,

$$\phi_1(\xi) = \left(KC + \frac{\$\beta_2}{\overline{C}}\right) R\left(\xi + \frac{m}{\xi^2}\right) + i\$\overline{D}R\left(\frac{1}{\xi} + m\xi^2\right).$$
(5.13)

Also, from (5.1) and (5.11),

$$\phi_1(\xi) = A_1 R \xi + A_2 R^2 \xi^2, \quad |\xi| \to \infty.$$
 (5.14)

It follows from (5.8) and (5.13-14) that

$$\phi_{1}(\xi) = X_{2}\xi^{2} + X_{1}\xi + \frac{X_{-1}}{\xi} + \frac{X_{-2}}{\xi^{2}}, \quad X_{o} = X_{-3} = X_{-4} = \dots = 0, \quad (5.15)$$
$$|\xi| > 1,$$

where

$$X_2 = A_2 R^2 = i \mathscr{S} \overline{D} Rm, \quad X_1 = A_1 R = \left(KC + \frac{\mathscr{S} \beta_2}{\overline{C}} \right) R,$$

$$X_{-1} = \frac{A_2 R^2}{m} = i \mathscr{S} \overline{D} R, \quad X_{-2} = A_1 R m = \left(K C + \frac{\mathscr{S} \beta_2}{\overline{C}} \right) R m,$$

and, from Theorem 2,

$$m, \quad \$D, \quad KC + \frac{\$\beta_2}{\overline{C}} \neq 0.$$

$$\left|\phi'\left(z\right)\right| = F'\left(I\right) \neq 0$$

and

$$\omega'\left(\xi\right)\neq0, \quad |\xi|>1,$$

it is clear that we must have

$$\phi_1'(\xi) \neq 0, \ |\xi| > 1.$$

In other words, from (5.8), we must have that

$$2X_2\xi + X_1 - \frac{X_{-1}}{\xi^2} - \frac{2X_{-2}}{\xi^3} \neq 0, \quad \forall \xi : |\xi| > 1.$$

This implies, together with (5.15), that there are restrictions on the constants A_1, A_2, B_1 and B_2 characterizing the remote stresses.

It remains to determine the analytic function $\psi_1(z)$. From the interface conditions the (5.3)

$$\psi_1(\xi) = i\left(-\alpha_1 \overline{\phi_1(\xi)} - \frac{\beta_1 \overline{\omega(\xi)}\omega'(\xi)}{\phi_1'(\xi)} + \alpha_2 \overline{\phi_2(\xi)} + \frac{\beta_2 \overline{\omega(\xi)}\omega'(\xi)}{\phi_2'(\xi)}\right) + \psi_2(\xi), \quad |\xi| = 1,$$

$$\therefore \quad \phi_{\gamma}'(z) = \frac{d\phi(z)}{dz} = \frac{d\phi(\xi)}{d\xi} \frac{d\xi}{dz} = \frac{\phi'(\xi)}{\omega'(\xi)}, \quad \overline{\phi_{\gamma}'(z)} = \overline{\left(\frac{\phi'(\xi)}{\omega'(\xi)}\right)} = \overline{\frac{\phi'(\xi)}{\omega'(\xi)}},$$

and form (5.11-12) and (5.15), it follows that, for $|\xi| = 1$,

$$\psi_{1}\left(\xi\right) = -i\alpha_{1}\left(\frac{\overline{X_{2}}}{\xi^{2}} + \frac{\overline{X_{1}}}{\xi} + \overline{X_{-1}}\xi + \overline{X_{-2}}\xi^{2}\right) - \frac{i\beta_{1}R^{2}\left(\frac{1}{\xi} + m\xi^{2}\right)\left(1 - \frac{2m}{\xi^{3}}\right)}{2X_{2}\xi + X_{1} - \frac{X_{-1}}{\xi^{2}} - \frac{2X_{-2}}{\xi^{3}}}$$

$$+i\alpha_2\overline{C}R\left(\frac{1}{\xi}+m\xi^2\right)+\frac{i\beta_2R\left(\frac{1}{\xi}+m\xi^2\right)}{C}+DR\left(\xi+\frac{m}{\xi^2}\right).$$
(5.16)

From (5.1) and (5.11), ψ_1 must also satisfy the asymptotic condition

$$\psi_1(\xi) = B_1 R \xi + B_2 R^2 \xi^2, \quad |\xi| \to \infty.$$

Hence, If we adopt ψ_1 from (5.16) for $|\xi| > 1$, we require that

$$-i\alpha_1 \left(\overline{X_{-1}}\xi + \overline{X_{-2}}\xi^2\right) - \frac{i\beta_1 R^2 m}{2X_2}\xi + i\alpha_2 \overline{C}Rm\xi^2 + \frac{i\beta_2 Rm}{C}\xi^2 + DR\xi$$
$$= B_1 R\xi + B_2 R^2 \xi^2, \quad |\xi| \to \infty.$$

Thus, since $X_{-1} = \frac{A_2 R^2}{m}$, $X_{-2} = A_1 Rm$, we require that

$$B_1 = -rac{ilpha_1\overline{A_2}R}{m} - rac{ieta_1m}{2A_2R} + D,$$

$$B_2 = -\frac{i\alpha_1 \overline{A_1}m}{R} + \frac{i\alpha_2 \overline{C}m}{R} + \frac{i\beta_2 m}{RC}$$

From (5.15), by letting $A_1 = \operatorname{Re} A_1 + i \operatorname{Im} A_1$, $C = \operatorname{Re} C + i \operatorname{Im} C$, we can solve for C and D, in terms of A_1 and A_2 . In fact,

$$A_1\overline{C} = K \left|C\right|^2 + \$\beta_2,$$

which yields

$$K \left[(\operatorname{Re} C)^{2} + (\operatorname{Im} C)^{2} \right] - \operatorname{Re} A_{1} \operatorname{Re} C - \operatorname{Im} A_{1} \operatorname{Im} C$$
$$+ i \left(\operatorname{Re} A_{1} \operatorname{Im} C - \operatorname{Im} A_{1} \operatorname{Re} C \right) + \mathscr{B}\beta_{2} = 0,$$

Then, we can derive two separate equations from the above equation by equating real and imaginary parts, respectively:

$$\operatorname{Re} A_1 \operatorname{Im} C - \operatorname{Im} A_1 \operatorname{Re} C = 0$$
, (imaginary part) (5.18)

$$K[(\operatorname{Re} C)^{2} + (\operatorname{Im} C)^{2}] - \operatorname{Re} A_{1} \operatorname{Re} C - \operatorname{Im} A_{1} \operatorname{Im} C + \$\beta_{2} = 0, \quad (\text{real part}).$$
 (5.19)

From (5.18)

$$\operatorname{Im} C = \frac{\operatorname{Im} A_1}{\operatorname{Re} A_1} \operatorname{Re} C, \qquad (5.20)$$

and from (5.21), Equation (5.19) can now be written in the form

$$K (\operatorname{Re} C)^2 \left[\frac{|A_1|^2}{(\operatorname{Re} A_1)^2} \right] - \operatorname{Re} C \left[\frac{|A_1|^2}{\operatorname{Re} A_1} \right] + \mathcal{S}\beta_2 = 0,$$
 (5.21)

where

$$|A_1|^2 = A_1 \overline{A_1} = (\operatorname{Re} A_1)^2 + (\operatorname{Im} A_1)^2.$$

5.3 COMPLETE SOLUTION OF A HYPOTROCHOIDAL INHOMOGENEITY 96

Solving Equation (5.21) in terms of $\operatorname{Re} C$:

Re
$$C = \frac{\text{Re } A_1 \left[|A_1| \pm \sqrt{|A|^2 - 4\beta_2 \mathscr{S} K} \right]}{2K |A|},$$
 (5.22)

Im
$$C = \frac{\text{Im } A_1 \left[|A_1| \pm \sqrt{|A|^2 - 4\beta_2 \mathscr{S}K} \right]}{2K |A|}$$
, (from (5.20)).

Recalling that $C = \operatorname{Re} C + i \operatorname{Im} C$, we finally have the expression of the complex constant C in terms of A_1 :

$$C = \mathcal{J}A_1, \tag{5.23}$$

where

$$\mathcal{I} = \left(\frac{\left|A_{1}\right| \pm \sqrt{\left|A\right|^{2} - 4\beta_{2} \mathcal{S} K}}{2K \left|A\right|} \right)$$

In addition, from (5.15), the unknown complex constant D can be determined by the relation:

$$D = \frac{iR}{\mathcal{S}m}\overline{A_2}.$$
(5.24)

Consequently, the constants C and D are therefore completely determined from (5.23) and (5.24). Also, from (5.17), B_1 and B_2 can be written in terms of A_1 and A_2 :

$$B_{1} = \overline{A}_{2} \frac{iR}{m} \left(\frac{1}{\mathscr{G}} - \alpha_{1} \right) - \frac{1}{A_{2}} \left(\frac{i\beta_{1}m}{2R} \right), \qquad (5.25)$$
$$B_{2} = \overline{A}_{1} \frac{im}{R} \left(\alpha_{2} \overline{\mathscr{I}} - \alpha_{1} \right) + \frac{1}{A_{1}} \left(\frac{i\beta_{2}m}{R \mathscr{I}} \right).$$

SUMMARY

5.3 COMPLETE SOLUTION OF A HYPOTROCHOIDAL INHOMOGENEITY 97

The complete solution of the inhomogeneity-matrix system subjected to linear remote loading described by (5.1) is given by

$$\phi_2(z) = Cz, \quad \psi_2(z) = Dz, \quad |z| \in S_2,$$
(5.26)

$$\phi_1(\xi) = X_2 \xi^2 + X_1 \xi + \frac{X_{-1}}{\xi} + \frac{X_{-2}}{\xi^2}, \qquad (5.27)$$

where

$$X_2 = A_2 R^2 = i \mathscr{B} \overline{D} Rm, \quad X_1 = A_1 R = \left(KC + \frac{\mathscr{B} \beta_2}{\overline{C}} \right) R,$$

$$X_{-1} = \frac{A_2 R^2}{m} = i \$ \overline{D} R, \quad X_{-2} = A_1 R m = \left(KC + \frac{\$ \beta_2}{\overline{C}} \right) R m,$$

and

$$\psi_{1}\left(\xi\right) = -i\alpha_{1}\left(\frac{\overline{X_{2}}}{\xi^{2}} + \frac{\overline{X_{1}}}{\xi} + \overline{X_{-1}}\xi + \overline{X_{-2}}\xi^{2}\right) - \frac{i\beta_{1}R^{2}\left(\frac{1}{\xi} + m\xi^{2}\right)\left(1 - \frac{2m}{\xi^{3}}\right)}{2X_{2}\xi + X_{1} - \frac{X_{-1}}{\xi^{2}} - \frac{2X_{-2}}{\xi^{3}}} + i\alpha_{2}\overline{C}R\left(\frac{1}{\xi} + m\xi^{2}\right) + \frac{i\beta_{2}R\left(\frac{1}{\xi} + m\xi^{2}\right)}{C} + DR\left(\xi + \frac{m}{\xi^{2}}\right), \quad (5.28)$$

where , $|\xi| > 1$. Here, the complex constants C and D are determined by the equations

$$C = \mathcal{J}A_1, \quad D = \frac{iR}{\mathcal{F}m}\overline{A_2},$$

$$J = \left(\frac{|A_1| \pm \sqrt{|A_1|^2 - 4\beta_2 SK}}{2K |A_1|} \right),$$

where, from Remark 8 and Theorem 2, we must have that

$$2X_2\xi + X_1 - \frac{X_{-1}}{\xi^2} - \frac{2X_{-2}}{\xi^3} \neq 0, \quad \forall \xi : |\xi| > 1,$$

C, A_1 , $A_2 \neq 0$ (to guarantee non-uniform remote stresses),

$$m, \ \mathcal{S}D, \ KC + \frac{\mathcal{S}\beta_2}{\overline{C}} \neq 0.$$

Finally, B_1 and B_2 are given by:

$$B_1 = \overline{A_2} \frac{iR}{m} \left(\frac{1}{\mathscr{S}} - \alpha_1 \right) - \frac{1}{A_2} \left(\frac{i\beta_1 m}{2R} \right),$$

$$B_2 = \overline{A_1} \frac{im}{R} \left(\alpha_2 \overline{J} - \alpha_1 \right) + \frac{1}{A_1} \left(\frac{i\beta_2 m}{R J} \right)$$

5.4 SOLUTION OF A HYPOTROCHOIDAL INHOMOGENEITY (GENERAL)

We can extend the results obtained above, to the case when the inhomogeneity-matrix system is subjected to non-uniform (not necessarily linear) remote loading characterized by the analytic functions ϕ_1 and ψ_1 in the form:

$$\phi_1(z) = A_1 z + A_n z^n, \quad \psi_1(z) = B_1 z + B_n z^n, \quad |z| \to \infty, \quad n \ge 2.$$
 (5.29)

Suppose that the inhomogeneity is hypotrochoidal with n + 1 cusps described by the conformal mapping function (Muskhelishvili, 1953),

$$z = \omega(\xi) = R\left(\xi + \frac{m_n}{\xi^n}\right), \quad R > 0, \quad 0 < m_n \le \frac{1}{n}.$$
 (5.30)


Figure 11: Example of Hypotrochoidal Inhomogeneity with n + 1 Cusps.

We assune that the state of Piola stress inside the inhomogeneity is uniform and described by the stress functions:

$$\phi_2(z) = Cz, \quad \psi_2(z) = Dz, \quad z \in S_2.$$
 (5.31)

From (5.1) and (5.30)

$$\phi_1(\xi) = A_1 R \xi + A_n R^n \xi^n, \quad \psi_1(\xi) = B_1 R \xi + B_n R^n \xi^n, \quad |\xi| \to \infty.$$
(5.32)

5.4 SOLUTION OF A HYPOTROCHOIDAL INHOMOGENEITY (GENERAL) 00

Applying the same procedure used in Section 5.4 above, we find that the analytic functions ϕ_1 and ψ_1 for $|\xi| > 1$ are given by

$$\phi_1(\xi) = X_n \xi^n + X_1 \xi + \frac{X_{-1}}{\xi} + \frac{X_{-n}}{\xi^n},$$
(5.33)

where

$$X_n = A_n R^n = i \$ \overline{D} R m_n, \quad X_1 = A_1 R = \left(KC + \frac{\$ \beta_2}{\overline{C}} \right) R,$$

$$X_{-1} = \frac{A_n R^n}{m_n} = i \mathscr{S} \overline{D} R, \quad X_{-n} = A_1 R m_n = \left(KC + \frac{\mathscr{S} \beta_2}{\overline{C}} \right) R m_n,$$

$$K = \frac{\mu_2}{\mu_1} + \alpha_2 \left(1 - \frac{\mu_2}{\mu_1} \right), \quad \mathscr{S} = \left(1 - \frac{\mu_2}{\mu_1} \right),$$

and

$$\psi_{1}\left(\xi\right) = -i\alpha_{1}\left(\frac{\overline{X_{n}}}{\xi^{n}} + \frac{\overline{X_{1}}}{\xi} + \overline{X_{-1}}\xi + \overline{X_{-n}}\xi^{n}\right) - \frac{i\beta_{1}R^{2}\left(\frac{1}{\xi} + m_{n}\xi^{n}\right)\left(1 - \frac{nm_{n}}{\xi^{n+1}}\right)}{nX_{n}\xi^{n-1} + X_{1} - \frac{X_{-1}}{\xi^{2}} - \frac{nX_{-n}}{\xi^{n+1}}}$$
$$+i\alpha_{2}\overline{C}R\left(\frac{1}{\xi} + m_{n}\xi^{n}\right) + \frac{i\beta_{2}R\left(\frac{1}{\xi} + m_{n}\xi^{n}\right)}{C} + DR\left(\xi + \frac{m_{n}}{\xi^{n}}\right), \qquad (5.3)$$

The complex constants C and D are determined by the following general equations

$$C = \mathcal{J}A_1, \quad D = \frac{iR^{n-1}}{\mathcal{S}m_n}\overline{A_n}, \tag{5.35}$$

(5.34)

$$I = \left(\frac{|A_1| \pm \sqrt{|A_1|^2 - 4\beta_2 SK}}{2K |A_1|} \right),$$

Also, from Remark 2 and Theorem 1, we must have that

$$nX_n\xi^{n-1} + X_1 - \frac{X_{-1}}{\xi^2} - \frac{nX_{-n}}{\xi^{n+1}} \neq 0, \quad \forall \xi : |\xi| > 1,$$

 $C, A_1, A_n \neq 0$ (to guarantee non-uniform remote stresses),

$$m_n, \ \ \mathcal{S}D, \ \ KC + \frac{\mathcal{S}\beta_2}{\overline{C}} \neq 0.$$

Consequently, from (5.33) and (5.35), the complex constants B_1 and B_2 in (5.29) are specifically choosen in such a way that

$$B_1 = \overline{A_n} \frac{iR^{n-1}}{m_n} \left(\frac{1}{\mathcal{S}} - \alpha_1\right) - \frac{1}{A_n} \left(\frac{i\beta_1 m_n}{nR^{n-1}}\right),$$

$$Bn = \overline{A_1} \frac{im_n}{R^{n-1}} \left(\alpha_2 \overline{J} - \alpha_1 \right) + \frac{1}{A_1} \left(\frac{i\beta_2 m_n}{R^{n-1} \overline{J}} \right)$$

For the case n = 2, it is clear from (5.29-35) that the general solution reduces to the corresponding results discussed in Section 5.4.

5.5 FURTHER DISCUSSION

As a consequence of the results obtained above, we can draw the following interesting conclusions.

5.5.1 Expressions for I and J Inside the Inhomogeneity

From (2.1) and (2.6), R and J can be calculated as follow:

$$J = \det \mathbf{F} = -\operatorname{Im} \left[w_{,1} \overline{w}_{,2} \right] = y_{1,1} y_{2,2} - y_{1,2} y_{2,1}, \tag{5.36}$$

$$R = tr(\mathbf{F}\mathbf{F}^{T}) = |w_{,1}|^{2} + |w_{,2}|^{2} = y_{p,s}y_{p,s}, \dots p, s = 1, 2,$$
(5.37)

Hence, in the view of (2.1), Equation (2.5) yeilds

$$I = \sqrt{R+2J} = \sqrt{tr(\mathbf{F}\mathbf{F}^{T}) + 2J},$$

$$= \sqrt{(y_{1,1} + y_{2,2})^{2} + (y_{1,2} - y_{2,1})^{2}},$$
(5.38)

where

$$(y_{1,1} + y_{2,2})^2 + (y_{1,2} - y_{2,1})^2 = 2(y_{1,1}y_{2,2} - y_{1,2}y_{2,1}) + y_{11}^2 + y_{22}^2 + y_{12}^2 + y_{21}^2.$$

Equations (2.25-26) give us that

$$w_2(z) = -i\alpha_2 C z + \overline{D} z - \frac{i\beta_2 z}{\overline{C}}.$$
(5.39)

Consequently, I takes the following form within the inhomogeneity

$$I = 2\left(\alpha_2 \left|C\right| + \frac{\beta_2}{\left|C\right|}\right),\tag{5.40}$$

and from Remark 2,

$$\left|\phi_{2}'\left(z\right)\right|=\left|C\right|\neq0.$$

$$I^{2} - 16\alpha_{2}\beta_{2} = 4\left(\alpha_{2} |C| - \frac{\beta_{2}}{|C|}\right)^{2} > 0,$$
(5.41)

which guarantees that F'(I), from (1) is well-defined.

5.5.2 Stress Distribution

From (2.18), the Piola stresses inside the inhomogeneity and at the remote boundary are given by

$$-\sigma_{21} + i\sigma_{11} = (\chi_2)_{,2}, \quad \sigma_{22} - i\sigma_{12} = (\chi_2)_{,1}, \quad (5.42)$$
$$-\sigma_{21}^{\infty} + i\sigma_{11}^{\infty} = (\chi_1)_{,2}, \quad \sigma_{22}^{\infty} - i\sigma_{12}^{\infty} = (\chi_1)_{,1}.$$

From Equation (2.26), (5.29) and (5.31)

$$\chi_1(z) = 2\mu_1 i [(\alpha_1 - 1)(A_1 z + A_n z^n) + i(\overline{B_1}\overline{z} + \overline{B_n}\overline{z^n}) + \frac{\beta_1 z}{\overline{A_1} + n\overline{A_n}\overline{z^{n-1}}}],$$

$$\chi_2(z) = 2\mu_2 i [(\alpha_2 - 1)Cz + i\overline{D}\overline{z} + \frac{\beta_2 z}{\overline{C}}], \qquad (5.43)$$

Thus, the corresponding Piola stress can be found from (41) as follows:

$$-\sigma_{21} + i\sigma_{11} = 2\mu_2 i [i(\alpha_2 - 1)C + \overline{D} + \frac{i\beta_2}{\overline{C}}], \qquad (5.44)$$

$$\sigma_{22} - i\sigma_{12} = 2\mu_2 i [(\alpha_2 - 1)C + i\overline{D} + \frac{\beta_2}{\overline{C}}] \text{ (uniform)}$$

$$-\sigma_{21}^{\infty} + i\sigma_{11}^{\infty} = 2\mu_1 i [i(\alpha_1 - 1)(A_1 + nA_n z^{n-1}) + \overline{B_1} + n\overline{B_n z^{n-1}}], \quad (5.45)$$

$$\sigma_{22}^{\infty} - i\sigma_{12}^{\infty} = 2\mu_1 i [(\alpha_1 - 1)(A_1 + nA_n z^{n-1}) + i(\overline{B_1} + n\overline{B_n z^{n-1}})] \text{ (non-uniform)}.$$

Hence, the Piola stresses inside the inhomogeneity (uniform) and at the remote boundary (non-uniform) can be calculated from (5.44-45).

•

In addition, the Cauchy stresses from (2.12) inside the inhomogeneity are given by

$$\tau_{11} + \tau_{22} = 2\mu_2 [\frac{IF'(I)}{J} - 2],$$

$$\tau_{11} - \tau_{22} + 2i\tau_{12} = 2\mu_2 \frac{F'(I)}{IJ} [(w_2)_{,2}^2 + (w_2)_{,1}^2],$$

which, in our case, from Equation (2.14) and (5.36-41),

$$F'(I) = |\phi'_2(z)| = |C| \neq 0,$$

are also uniform. In fact, from (5.4), we obtain

$$I = 2\left(\alpha_2 \left|C\right| + \frac{\beta_2}{\left|C\right|}\right),\,$$

and from (5.36) and (5.39)

$$J = -\text{Im}\left[(w_2(z))_{,1}\overline{(w_2(z))_{,2}}\right]$$

$$= \alpha_2 |C|^2 + 2\alpha_2\beta_2 + |D|^2 + \frac{\beta_2}{|C|^2},$$

where

$$(w_2(z))_{,1} = -i\alpha_2 C + \overline{D} - \frac{i\beta_2}{\overline{C}}, \ (w_2(z))_{,2} = \alpha_2 C + i\overline{D} + \frac{\beta_2}{\overline{C}}.$$

Consequently, from (2.21), the Cauchy stresses are can be calculated as

$$\tau_{11} + \tau_{22} = 2\mu_2 \left[\frac{2\left(\alpha_2 \left|C\right|^2 + \beta_2\right)}{\alpha_2 \left|C\right|^2 + 2\alpha_2\beta_2 + \left|D\right|^2 + \frac{\beta_2}{\left|C\right|^2}} - 2 \right],$$

$$\tau_{11} - \tau_{22} + 2i\tau_{12} = 2\mu_2 \frac{|C| \left[(w_2(z))_{,2}^2 + (w_2(z))_{,1}^2 \right]}{2\left(\alpha_2 |C| + \frac{\beta_2}{|C|} \right) \left[\alpha_2 |C|^2 + 2\alpha_2\beta_2 + |D|^2 + \frac{\beta_2}{|C|^2} \right]},$$

which, clearly, indicate that the Cauchy stresses inside the inhomogeneity are uniform.

5.6 EXAMPLE: THE CASE n = 3

The general solutions mentioned above can accommodate many different forms of applied remote loading. Among these, we consider the particular class of remote stresses, characterized by the stress functions $\phi_1(z) = A_1 z + A_2 z^3$, $\psi_1(z) = B_1 z + B_2 z^3$, where A_1, A_3, B_1 and B_3 are prescribed complex constants in such a way that

$$B_{1} = \overline{A_{3}} \frac{iR^{2}}{m_{3}} \left(\frac{1}{\mathcal{S}} - \alpha_{1}\right) - \frac{1}{A_{3}} \left(\frac{i\beta_{1}m_{3}}{3R^{2}}\right), \qquad (5.46)$$
$$B_{3} = \overline{A_{1}} \frac{im_{3}}{R^{2}} \left(\alpha_{2}\overline{\mathcal{J}} - \alpha_{1}\right) + \frac{1}{A_{1}} \left(\frac{i\beta_{2}m_{3}}{R^{2}\overline{\mathcal{J}}}\right),$$

where

$$J = \left(\frac{|A_1| \pm \sqrt{|A_1|^2 - 4\beta_2 \mathcal{S}K}}{2K|A_1|}\right), \ K = \frac{\mu_2}{\mu_1} + \alpha_2 \left(1 - \frac{\mu_2}{\mu_1}\right), \ \mathcal{S} = \left(1 - \frac{\mu_2}{\mu_1}\right), \ (5.47)$$

and the corresponding conformal mapping function (hypotrochoid with n = 3) (Muskhelishvili, 1953), from (5.30), is given by

$$z = \omega(\xi) = R\left(\xi + \frac{m_3}{\xi^3}\right), \quad R > 0, \quad 0 < m_3 \le \frac{1}{3},$$
 (5.48)

5.6.1 Complete Solutions

From (5.29) and (5.31), the stress functions at the remote boundary and inside the inhomogeneity, respectively, are given by

$$\phi_1(z) = A_1 z + A_3 z^3, \quad \psi_1(z) = B_1 z + B_3 z^3, \quad |z| \to \infty,$$

$$\phi_2(z) = C z, \quad \psi_2(z) = D z, \quad z \in S_2.$$
(5.49)

From the general solution (5.35), the complex constants C and D are completely determined by the equations

$$X_3 = A_3 R^3 = i \mathscr{S}\overline{D}Rm_3, \quad X_1 = A_1 R = \left(KC + \frac{\mathscr{S}\beta_2}{\overline{C}}\right)R, \quad (5.50)$$

$$X_{-1} = \frac{A_3 R^3}{m_3} = i \mathscr{S} \overline{D} R, \quad X_{-3} = A_1 R m_3 = \left(KC + \frac{\mathscr{S} \beta_2}{\overline{C}} \right) R m_3,$$

$$C = IA_1, \quad D = \frac{iR^2}{Sm_3}\overline{A_3}.$$
 (5.51)

Also, from Remark 8 and Theorem 2, we must have that

C, A_1 , $A_3 \neq 0$ (to guarantee non-uniform remote stresses),

and in the surriounding matrix

$$3X_3\xi^2 + X_1 - \frac{X_{-1}}{\xi^2} - \frac{3X_{-3}}{\xi^4} \neq 0, \quad \forall \xi : |\xi| > 1,$$

$$m_3, \ \ SD, \ \ KC + \frac{S\beta_2}{\overline{C}} \neq 0.$$
 (5.52)

5.6.2 Determination of I and J Inside the Inhomogeneity

From (5.39) together with (5.51), the deformation function inside the inhomogeneity now takes the form

$$w_{2}(z) = -i\alpha_{2} \mathbb{J}A_{1}z - \frac{iR^{2}}{\mathbb{S}m_{3}}A_{3}z - \frac{i\beta_{2}z}{\mathbb{J}A_{1}}.$$
(5.53)

and from (5.40-41) and (5.51)

$$\begin{split} I &= 2\left(\alpha_2 \left| \mathcal{J}A_1 \right| + \frac{\beta_2}{\left| \mathcal{J}A_1 \right|} \right), \\ & |\phi_2'\left(z\right)| = \left| \mathcal{J}A_1 \right| \neq 0, \end{split}$$

$$I^{2} - 16\alpha_{2}\beta_{2} = 4\left(\alpha_{2} \left| \mathbf{J}A_{1} \right| - \frac{\beta_{2}}{\left| \mathbf{J}A_{1} \right|} \right)^{2} > 0,$$
(5.54)

from which I can be calculated. Consequently, F'(I), from (2.21) is well-defined. Also J can now be determined from (5.36) together with (5.53),

$$J = -\operatorname{Im}\left[\left(w_{2}\right)_{,1}\left(\overline{w}_{2}\right)_{,2}\right],$$

where

$$w_2(z) = -i\alpha_2 \mathcal{J}A_1 z - \frac{iR^2}{\mathcal{S}m_3} A_3 z - \frac{i\beta_2 z}{\mathcal{J}A_1}$$

which implies, together with (5.33), that the Cauchy stresses are uniform inside the inhomogeneity (see Equation (5.45) below).

5.6.3 Deformed Contour

From (5.53), we can easily plot the deformed contour (see Figure 12). The fact that the corresponding complex constants C and D satisfy (5.52) guarantee that there is no overlapping.



Figure 12: Deformed Contour of the Hypotrochoidal Inhomogeneity with 4 Cusps (n = 3). Here, we have taken ($A_1 = i, A_3 = 0.25i, \mu_1 = 0.27, \mu_2 = 0.3, \alpha_1 = \alpha_2 = \beta_1 = \beta_2 = 0.5$)

5.6.4 Admissibility Conditions

From (5.36) and (5.53-54), inside inhomogeneity,

$$I^2 > 16\alpha_2\beta_2, \quad J > 0.$$
 (5.55)

As in Ru (2002), (5.55) ensures that there is no overlapping of the deformation field (interior to the inhomogeneity).

5.6.5 The Piola Stress Distribution

From (5.44-45) and (5.51), the Piola stress in the inhomogeneity and the matrix is described, respectively, by:

$$-\sigma_{21} + i\sigma_{11} = 2\mu_2 i [i(\alpha_2 - 1) J A_1 - \frac{iR^2}{Sm_3} A_3 + \frac{i\beta_2}{J A_1}],$$

$$\sigma_{22} - i\sigma_{12} = 2\mu_2 i [(\alpha_2 - 1) \mathcal{J}A_1 + \frac{R^2}{\mathcal{S}m_3}A_3 + \frac{\beta_2}{\mathcal{J}A_1}]$$

$$-\sigma_{21}^{\infty} + i\sigma_{11}^{\infty} = 2\mu_1 i [i(\alpha_1 - 1)(A_1 + 3A_3 z^2) + \overline{B_1} + 3\overline{B_3 z^2}],$$

$$\sigma_{22}^{\infty} - i\sigma_{12}^{\infty} = 2\mu_1 i [(\alpha_1 - 1)(A_1 + 3A_3 z^2) + i(\overline{B_1} + 3\overline{B_3} \overline{z^2})].$$

Again, from the above, the state of (Piola) stress inside the inhomogeneity is uniform.

5.7 CONCLUSIONS

We have discussed the idea that by adjusting the remote (Piola) stress we can design the shape of the inhomogeneity in such a way that the interior (Piola) stress distribution remains uniform. In fact, using complex variable techniques, we have shown that a uniform (Piola) stress distribution can be achieved inside the inhomogeneity when the system is subjected to linear remote loading, in which case, the inhomogeneity is necessarily hyptrochoidal

with three cusps. In this case, we have obtained the complete solution of the corresponding inhomogeneity-matrix system in this. In addition, we also considered the more general case when the system is subjected to nonuniform remote loading characterized by stress functions of the form of n^{th} degree polynomials in the complex variable z describing the matrix. By finding the complete solution of the corresponding system, we have shown that we can again obtain uniform stress inside a hypotrochoidal inhomogeneity with n + 1cusps.

Chapter 6 CONCLUSIONS AND FUTURE WORK

6.1 CONCLUSIONS

In this dissertation, certain important problems concerning a single inhomogeneity-matrix system of harmonic-type undergoing finite plane deformations are considered. For an interface condition, the state of perfect bonding generally referred to as a "perfectly bonded interface" is assumed so that both traction and displacement is continuous across the interface. By successfully incorporating complex variable techniques, we have analyzed the inhomogeneity-matrix system subjected to varying remote loading conditions (not necessarily uniform) and derived novel results of each corresponding problem discussed through Chapters. 3-5.

Chapter 3 addresses and analyzes the problems of an elliptic inhomogeneity subjected to classes of nonuniform remote stress distribution. Despite its practical importance in finite plane elastostatics, so far, there have been no closed form solutions available for such problems. This is due mainly to complications arising from the use of conformal mapping techniques with the ensuing nonlinearities in the mapped plane.

In the present study, we overcome the above-mentioned difficulties and address the issues relating to nonuniform remote loading of harmonic composite materials. In particular, we extend the results in Ru et al (2005) to the case of plane finite deformations of a composite material in which an elliptic elastic inhomogeneity is embedded in the same class of harmonic materials under the assumption of nonuniform remote loading. Using complex-variable methods, we obtain the complete solution for a perfectly bonded elliptic inhomogeneity when the system is subjected to classes of nonuniform remote stress distributions characterized by stress functions described by general polynomials of degree $n \ge 1$ in the corresponding complex variable z describing the matrix. The results obtained are extremely important in that, essentially, they lead to the solution of a class of problems in which the remote loading is characterized by a much wider and more practically realistic class of functions. We also mention here that the solutions obtained in Chapter 3 are general in that they can accommodate the existing results in the literatures (Ru et al, 2005; Ogden & Isherwood, 1978) as special cases of the solutions. In particular, the methodology proposed in Chapter 3 enables us to consider several interesting subjects involving the finite plane deformation of an inhomogeneity-matrix system.

Eshelby's conjecture in finite plane elastostatics is examined in Chapter 4. What is particularly interesting in the present study is to ask if the above statement holds true for the special class of harmonic material considered here. One side of Eshelby's conjecture for this material undergoing finite plane deformations was examined Ru et al (2005). In particular, it was shown there that under the analogous constraints of perfect bonding and uniform remote Piola stress, if the Piola stress within the inhomogeneity is uniform, the

inhomogeneity is necessarily elliptical except for the special case when the remote loading takes a particular restricted form, in which case the shape of the inhomogeneity can be taken as arbitrary. The 'other side' of the conjecture, however, remained unfinished due to the existing complications arising from the use of mapping (conformal) techniques. By virtue of the analogy developed in Chapter 3, we overcome the above-mentioned difficulties and examine, in detail, the converse result. More precisely, it was shown in Chapter 4 that the Piola stress distribution within a perfectly bonded elliptic inhomogeneity is necessarily uniform when the same inhomogeneity-matrix system is subjected to finite plane deformations and uniform remote loading. This statement completes the Eshelby's conjecture for the class of harmonic materials considered here. In addition, the complete solution for this system for any uniform remote stress distribution is obtained which, in turn, reduces to the results in Ru et al (2005). Further, the issue of "uniform stress distribution inside inhomogeneity" provided motivation for the study of designing an inhomogeneity with uniform interior stress (Chapter 5).

The state of uniform interior stress distribution is a desirable state in that, by inducing uniform stress distribution in a system, one can remove unfavorable effects such as stress intensification and concentration both leading to the failure of a system. Therefore, it is of particular interest to see if we can actually achieve this state for an inhomogeneity-matrix system undergoing finite plane deformations in the same class of harmonic materials considered here. In fact, using complex variable techniques and the analogy developed in Chapter 3, it is shown in Chapter 5 that if the Piola stress distribution inside the inhomogeneity is uniform and the system is subjected to linear remote loading, the inhomogeneity is necessarily hypotrochoidal, with no exceptions. The complete solution of the inhomogeneity-matrix system is obtained in this case. In particular, we extend the above results to the more general case when the system is subjected to nonuniform remote loading characterized by stress functions of the form of n^{th} degree polynomials in the complex variable z describing the matrix. In this case, by finding the complete solution of the system, we show that we can again obtain uniform stress inside a hypotrochoidal inhomogeneity with n + 1 cusps. In other words, by adjusting the cusps of a hypotrochoidal inhomogeneity and the remote loading, we can always achieve uniform stress distribution inside the inhomogeneity subjected to a relatively wide class of remote stresses discussed in Section 5.5.

The results derived in this dissertation will enable one to have a clear understanding of the state of the stress of an inhomogeneity-matrix system. In particular, the complete solutions of the corresponding problems in Chapters. 3-5, provide physical relevance to those corresponding "real-world problems". This will eventually enhance the future design of composite materials especially for fiber-reinforced composites.

6.2 FUTURE WORK

The most important extension of our work concerns the incorporation of a more realistic interface model. One which is 'imperfect' in the sense that it can include the effects of voids, cracks and imperfections in the glue layer between the fiber and the matrix. This has been done to a certain extent for linear elasticity (Ru & Schiavone, 1996, 1997; Ru, 1998a, 1998b; Sudak et al, 1999) but not, in any way, for finite elasticity. This is a subject

114

of extreme interest and practical significance from nano-mechanics to traditional areas of continuum mechanics.

It is suggested that we begin with a model of a 'linear interface' embedded in a nonlinear material (Aboudi, 1987; Steif & Hoysan, 1987; Gosz & Achebach, 1992; Benveniste, 1985; Zhong & Meguid, 1996). This is a good starting point for our analysis but since this works is absolutely new, we expect this to be extremely challenging yet immensely new circling area of research.

- Aboudi, J., 1987. Damage in Composite-Modeling of Imperfect Bonding. Composite Science and Technology., 28, 103-128.
- Abeyaratne, R., 1983. Some finite elasticity problems involving crack tips. In modelling problems in crack tip mechanics (Pindera, J.T., ed.) 3-24,. Waterloo: University of Waterloo, Canada.
- Abeyaratne, R. and Horgan, C.O., 1984. The pressurized hollow sphere problem in finite elastostatics for a class of compressible materials. Int. J. Solid. Struct., 20, 715-723.
- Achenbach, J. D. and Zhu, H., 1990. Effect of Interphases on Micro and Macro mechanical Behavior of Hexagonal-Array Fiber Composites. ASME Journal of Applied Mechanics., Vol. 57, 956-963.
- Benveniste, Y., 1985. The Effective Mechanical Behavior of Composite Materials with Imperfect Contact between the Constituents. Mechanics of Materials., 4, 1985, 197-208.
- Bigoni, D., Serkove S. K., Valentini, M. and Movechan, A. B., 1998. Asymptotic Model of Dilute Composite with Imperfectly Bonded Inclusions. International Journal of Solids and Structures., 35, 3239-3258.
- Brennan, J. J., 1988. Interfacial Chemistry and Bonding in Fiber Reinforced Glass and Class-Ceramic Matrix Composites, Ceramic Microstructures and Role of Interface. J.
 A. Pask and A. G. Evans Eds., 387-399, Plenum Press, New York.
- Chadwick, P., 1999. Continuum Mechanics: Concise Theory and Problems, 2nd. Dover Publication. Inc., 31 East 2nd street, Mineola, N. Y. 11501.
- Chen, Y. J., Chern, J. C., Chan, Y. W., Liao, W. C., Wu, A. K., 2007. *Reinforced reactive powder concrete plate under cyclic loading*. Journal of the Chinese Institute of Engineers, Transactions of the Chinese Institute of Engineers, Series A/Chung-kuo Kung Ch'eng Hsuch K'an., Vol. 30, 299-310.
- Christenson, R.M. and Lo, K. H., 1979. Solutions for Effective Shear Properties in Three Phase Sphere and Cylinder Models. Journal of the Mechanics and Physics of Solids., Vol. 27, 315-330.
- Christenson, R.M., 1990. A Critical Evaluation for a Class of Micromechanics Models. Journal of the Mechanics and Physics of Solids., Vol. 38, No. 3, 379-404.

Chung, D. D., 1994. Carbon Fiber Composites. Butterworth-Heinemann.

- Donald R. Askeland, 1990. *The Science and Engineering of Materials, Second S.I. Edition*. Chapmand and Hall.
- Eshelby, J. D., 1957. *The Determination of the Elastic Field of an Ellipsoidal Inclusion and Related Problems*. Proc. R. Soc. London. A., **241**, 376-396.
- Eshelby, J. D., 1959. *The Elastic Filed Outside an Elliptic Inclusion*. Proc. R. Soc. London. A., **252**, 561-569.
- Fung, Y. C., 1977. A First Course In Continuum Mechanics, 2nd Ed. Prentice-Hall, Inc., Englewood Cliffs. New Jersey.
- Gao, F., Jiao, G., Lu and Z., Ning, R., 2007. *Mode II delamination and damage resistance* of carbon/epoxy composite laminates interleaved with thermoplastic particles. Journal of Composite Materials., Vol. 41, No. 1, 111-123.
- Gao, Z., 1995. A Circular Inclusion with Imperfect Interface: Eshelby's Tensor and Related Problems. ASME. Journal of Applied Mechanics., 62, 860-866.
- Ghahremani, F., 1980. Effect of Grain Boundary Sliding on Anelasticity of Polycrystals. International journal of Solids and Sturctures., Vol. 16,825-845.
- Gosz, M., Moran, B. and Achebach, J. D., 1992. Load-Dependent Constitutive Response of Fiber Composite with Compliant Interphases. Journal of the Mechanics and Physics of Solids., 40, No. 8, 1789-1083.
- Gurtin, M. E., 1981. An Introduction to Continuum Mechanics. Academic Press. N. Y.
- Hashin, Z., 1983. Analysis of Composite Materials- A Survey. ASME, Journal of Applied Mechanics, Vol. 50, 481-505.
- Hashin, Z., 1991. The Spherical Inclusion with Imperfect Interface. J. Appl. Mech., 58, 444-449.
- Hang, J. Z. S., Li, Y., Lou, Y. Y., Zhong, Q. D. and Zhang, J. P., 2006. Study on the mechanical properties and corrosion resistance properties of nanometer barite/epoxy composite paint film material. Gongneng Cailiao/Journal of Functional Materials., Vol. 37, No. 12, 2003-2006.
- Hill, R., 1965. *A self-Consistent Mechanics of Composite Materials*. Journal of the Mechanics and Physics of Solids, Vol. 13, 213-222.

- Horgan, C.O., 1989. Some remarks on axisymmetric solutions in finite elastostatics for compressible materials. Proceedings of the Royal Irish Academy, Section A (invited paper in P. M. Quinlan Birthday issue)., 89, 185-193.
- Horgan, C.O., 1995. On axisymmetric solutions for compressible nonlinearly elastic solids. J. Appl. Math. Phys. (ZAMP) (invited paper in P. M. Naghdi Birthday issue)., 46, S107-S125.
- Horgan, C. O. and Chan, A. M., 1999. The Pressurized Hollow Cylinder or Disk Problem for Functionally Graded Isotropic Linearly Elastic Materials. Journal of Elasticity., Vol. 55, No. 1, 43-59.
- Horgan, C.O., 2001. Equilibrium solutions for compressible nonlinearly elastic materials. Invited Chapter in Nonlinear Elasticity: Theory and Applications (ed. by R. W. Ogden and Y. Fu). London Mathematical Society Lecture Notes., No. 283, Cambridge University Press, 135-159.
- Huang, Y. and Hu, K.X., 1995. A Generalized Self-Consistent Mechanics Method for Solid Containing Elliptical Inclusions. ASME. Journal of Applied Mechanics., Vol. 62, 566-572.
- Irie. M, 1989. Development and Aplication of Shape Memory Polymers. CMC., 24.
- Jafari, A.H., Abeyaratne, R. and Horgan, C. O., 1984. The finite deformation of a pressurized circular tube for a class of compressible materials. J. Appl. Math. Phys. (ZAMP), 35, 227-246.
- Jayaraman, K., Reinfsnider, K. L. and Swain, R. E., 1993. Elastic and Thermal Effects in the Iterphase: Part I. Comments on Characterization Methods. Journal of Composite, Technology and Research. JCTRER, 15 No. 1, 3-13.
- Jacob, Aboudi, 1991. Mechanics of Composite Materials: A Unified Micromechanical Approach. Elsevier Publishers.
- John, D. W., 1999. Biaxial Fatigue and Leakage Characteristics of Fiber Reinforced Composite Tubes. Ph.D. Thesis. University of Alberta.
- John, F., 1960. *Plane Strain Problems for a Perfectly Elastic Material of Harmonic Type.* Comm. Pure Appl. Math., XIII, 239-290.
- Jones, J. P. and Whittier, J. S., 1967. *Waves at a Flexibility Bonded Interface*. Journal of Applied Mechanics., Vol. 34, 905-909.

- Karthikeyan, C.S., Sankaran, S. and Kishore, 2007. Investigation of bending modulus of fiber-reinforced syntactic foams for sandwich and structural applications. Polymers for Advanced Technologies., Vol 18, No. 32, 254-256.
- Kim, C. I. and Schiavone, P., 2007a. An Elliptic Inhomogeneity Subjected to a General Class of Nonuniform Remote Loadings in Finite Elasticity. Mathematics and Mechanics of Solids. (Accepted for publication).
- Kim, C. I. and Schiavone, P., 2007b. A Circular Inhomogeneity Subjected to Nonuniform Remote Loading in Finite Plane Elastostatics. International Journal of Non-Linear Mechanics. (Accepted for publication).
- Kim, C. I., Vasudevan, M. and Schiavone, P., 2007. Eshelby's Conjecture in Finite Plane Elastostatics. (Submitted).
- Kim, C. I. and Schiavone, P., 2007c. Designing an Inhomogeneity with Uniform Interior Stress in Finite Plane Elastostatics. (Submitted).
- Kim, J. O. and Bau, H. H., 1992. A Study of the Fiber-Matrix Interface in Composite Materials. ASME. Journal of Applied Mechanics., Vol. 59, S163-S165.
- Knowles, J.K. and Sternberg, E., 1975. On the singularity induced by certain mixed boundary value problems in linearized and nonlinear elastostatics. Int. J. Solid. Struct., 11, 1173 – 1201.
- Lipton, R. and Vernescu B., 1995. Variational Methods, Size Effects and External Microstructures for Elastic Composite with Imperfect Inteface. Mathematical Models and Methods in Applied Sciences., 5, 1139-1173.
- Li, X. and Steigmann, D.J., 1993. Finite plane twist of an annular membrane. Quart. J. Mech. Appl. Math., 46, 601 625.
- Malvern, L. E., 1969. Introduction to the Mechanics of a Continuous Medium. Prentice-Hall., Englewood Cliffs, N. J.
- Nemat-Nasser, S. and Hori, M., 1993. *Micromechanics: Overall Properties of Heterogeneous Materials*. North-Holland., New York.
- Ogden, R. W. and Isherwood, D. A., 1978. Solution of Some Finite Plane-Strain Problems for Compressilbe Elastic Solids. Quarterly Journal of Mechanics and Applied Mathematics., **31**, 219-249.

- Pindera, M. J., Aboudi, J. and Arnod, S. M., 1995. Limitations of the Uncoupled, RVEbased Micromechanical Approach in the Analysis of Functionally Graded Composite. Mechanics of Materials., Vol. 20, 77-94
- Ru, C. Q. & Schiavone, P., 1996. On the Elliptic Inclusion in Anti-Plane Shear. Math. Mech. Solids., 1, 327-333.
- Ru, C. Q. and Schiavone, P., 1997. A Circular Inclusion with Circumferentially Inhomogeneous Interface In Anti-Plane Shear. Proc. R. Soc. London A., 453, 2551-2572.
- Ru, C. Q., 1998a. Interface Design of Neutral Elastic Inclusions. International Journal of Solids and Structures., 35, 599-572.
- Ru, C. Q., 1998a. Effect of Interphase Layers on Thermal Stresses within an Elliptical Inclusion. J. Appl. Phys., 84, 4872-4879.
- Ru, C. Q., 1998b. A Circular Inclusion with Circumferentially Inhomogeneous Sliding Interface in Plane Elastostatics. J. Appl. Mech., 65, 30-38.
- Ru, C. Q., 2002. On Complex-Variable Formulation for Finite Plane Elastostatics of Harmonic Materials. Acta Mech., 156, 219-234.
- Ru, C. Q., P. Schiavone, L. J. Sudak, A. Mioduchowski, 2005. Uniformity of Stresses Inside an Elliptic Inclusion in Finite Plane Elastostatics. International Journal of Non-Linear Mechanics., 40, 281-287.
- Schiavone, P., 2002. On an Elastic Circular Inhomogeneity with Imperfect Interface in Antiplane Shear. Journal of Applied Mechanics. ASME., Vol. 69.
- Schiavone, P., 2003. Neutrality of the Elliptic Inhomogeneity in the Case of Nonuniform Loading. Int. J. Eng. Sci., 41, 2081-2090.
- Sendeckyj, G. P., 1970. *Elastic Inhomogeneity Problems in Plane Elastosatics*. Int. J. Solids Structures., **6**, 1535-1543.
- Shen, H., Schiavone, P, Ru, C. Q. and Mioduchowski, A., 2000a. An Elliptic Inclusion with Imperfect Interface in Antiplane Shear. International Journal of Solids and Structures., 37, 4557-4575.
- Shen, H., Schiavone, P. Ru, C. Q. and Mioduchowski, A., 2000b. Aanlysis of Internal Stress in an Elliptic Inclusion with Imperfect Interface in Plane Elasticity. Mathematics and Mechanics of Solids., 5, 501-512.

- Steif, P. S. and S. F., Hoysan, 1987. An Energy Method for Calculating the Stiffness of Aligned Short-Fiber Composites. Mechanics of Materials., 6, 197-210.
- Sudak, L., Ru, C. Q., Schiavone, P. and Mioduchowski, A., 1999. A Circular Inclusion with Inhomogeneously Imperfect Interface in Plane Elasticity. Journal of Elasticity., 55, 19-41.
- Schmauder, S., Mullet, W.H. and McMeeking, R.M., 1992. Residual Stresses at Interface in Fiber Reinforced Composite. Residual Stresses-III Science and Technology, H. Fujiwara, T. Abe and K. Tanaka Eds., 589-594, Elsevier Applied Science, London.
- Timoshenko, S. P. and Goodier, J. N. 1970. *Theory of Elasticity, 3rd Ed.*, McGraw-Hill Book Co., New York.
- Tjong, S.C., Wang, G.S. and Mai, Y. W., 2005. High cycle fatigue response of in-situ Albased composites containing TiB2 and Al2O3 submicron particles. Composites Science and Technology., Vol. 65, No. 10, 1537-1546.
- Truesdell, C. and Toupin, R. A., 1960. *The Classical Field Theories*. In: S. Flugged, Ed., Handbuch derphysick III/1, 226-793. Springer Verlag, Berlin.
- Truesdell, C. and Noll, W., 1965. *The Nonlinear Field Theories of Mechanics*. In: S. Flugged, Ed., Handbuch derphysick **III/3**. Springer Verlag, Berlin.
- Tupper, M., Munshi, N., Beavers, F., Gall, K., Mikuls, M., Jr. and Meink, T., 2001. Developments in elastic memory composite materials for spacecraftdeployable structures. Aerospace Conference, IEEE Proceedings., Vol. 5, Issue, 2541 – 2547.
- Varley, E. and Cumberbatch, E., 1980. Finite deformation of elastic materials surrounding cylindrical holes. J. Elast., 10, 341 405.
- Van Vliet, D., Schiavone, P. and Mioduchowski, A., 2003. On the Design of Neutral Elastic Inhomogeneities in the case of Nonuniform Loading. Math. Mech. Solids., 8, 161-169.
- Wang, G.F., Schiavone, P. and Ru, C-Q., 2005. Harmonic Shapes in Finite Elasticity Under Non-uniform Loading. ASME. J. Appl. Mech., 72, 691 - 694.
- Zhang, C. S. and Ni, Q. Q., 2007. Bending behavior of shape memory polymer based laminates. Composite Structures., Vol 78, No. 2, 153-161.
- Zhang, Y. and Zhao, Y. P., 2007. A discussion on modeling shape memory alloy embedded in a composite laminate as axial force and elastic foundation. Materials and Design., Vol. 28, No. 3, 1016-1020.

Zhong Z. and Meguid S. A., 1996. On the Eigenstrain Problem of a Spherical Inclusion with an Imperfect Bonded Interface. ASME. Journal of Applied Mechanics., 63, 877-883.

Appendix A DERIVATIONS

A.1 DERIVATION OF RESULTS IN CHAPTER. 3

From the general solution presented in Chapter 3, we derived the corresponding results from Ru et al (2005). In fact, by setting

$$A_n = B_n = 0, \quad n \ge 2,$$

we derive from (3.1) that the complex potentials characterizing the remote stress are taken to be

$$\phi_1(z) = A_1 z, \quad \psi_1(z) = B_1 z \quad |z| \to \infty.$$

Then, the general solution (3.19B) and (3.19D) yield only that

$$0 = \sum_{n=a}^{b} [A_n] \xi^n = \sum_{n=a}^{b} [X_{n,(n,r=0)}] \xi^n = \sum_{n=a}^{b} [KC_n + iS\overline{D_n}m^n] \xi^n,$$

$$0 = \sum_{n=a}^{b} [B_n] \xi^n = \sum_{n=a}^{b} [-i\alpha_1 \overline{X_{n,(n,r=n)}} + i\alpha_2 \overline{C_n}m^n + D_n] \xi^n, \quad a \ge 2, b \ge a,$$

where

$$\overline{X_{n,(n,r=n)}} = K\overline{C_n}m^n - iSD_n,$$

which, in turn leads to

$$KC_n + iSD_nm^n = 0,$$
$$i(\alpha_1K - \alpha_2)C_nm^n + \overline{D_n}(1 - \alpha_1S) = 0$$

For the class of harmonic materials discussed here,

$$K, \ (1 - \alpha_1 S) \neq 0.$$

Therefore, to satisfy the above equations, it is shown in Chapter 4 (see Cases. 1-4) that we must have

$$C_n = D_n = 0, \quad n \ge 2,$$

which implies that the subsidiary complex constants are taken to be

$$C_{n,\alpha} = D_{n,\alpha} = 0, \quad n \ge 2.$$

Hence, Equation (3.19A) and (3.19C) become:

$$A_1 R\xi = \xi \left(KC_1 R + iS\overline{D_1}Rm \right) + \lim_{|\xi| \to \infty} \left(S \frac{\beta_2 R \left(\xi + \frac{m}{\xi} \right)}{\overline{C_1}} \right)$$

$$B_1R\xi = \xi \left[-i\alpha_1 \left(K\overline{C_1}Rm - iSD_1R \right) + i\alpha_2\overline{C_1}Rm + D_1R \right]$$

$$+\lim_{|\xi|\to\infty}\left[\left(\frac{\beta_2 R\left(\xi+\frac{m}{\xi}\right)}{\overline{C_1}}\right)i\left(1-\alpha_1 S\right)\right] -\lim_{|\xi|\to\infty}\left[\frac{i\beta_1 R^2\left(\frac{1}{\xi}+m\xi\right)\left(1-\frac{m}{\xi^2}\right)}{\phi_1'\left(\xi\right)}\right],$$

where, from (3.17),

where, from (3.17),

$$\phi_1\left(\xi\right) = \xi \left(KC_1R + iS\overline{D_1}Rm\right) + \frac{1}{\xi} \left(KC_1Rm + iS\overline{D_1}R\right) + S\frac{\beta_2 R\left(\xi + \frac{m}{\xi}\right)}{\overline{C_1}},$$

$$\phi_1'\left(\xi\right) = KC_1R + iS\overline{D_1}Rm + S\frac{\beta_2R}{\overline{C_1}} - \frac{1}{\xi^2}\left(KC_1Rm + iS\overline{D_1}RS + \frac{\beta_2Rm}{\overline{C_1}}\right)$$

Thus, at the remote boundary, we obtain,

$$A_1R = KC_1R + iS\overline{D_1}Rm + S\frac{\beta_2R}{\overline{C_1}}$$

$$B_1 R = -i\alpha_1 \left(K\overline{C_1}Rm - iSD_1R \right) + i\alpha_2 \overline{C_1}Rm + D_1R + \frac{\beta_2 Rm}{C_1} i \left(1 - \alpha_1 S \right) - \frac{i\beta_1 Rm}{A_1},$$
$$(\because A_1 = KC_1 + iS\overline{D_1}m + S\frac{\beta_2}{\overline{C_1}}).$$

Consequently, by setting $R = \lambda$, $Rm = \lambda_1$ and recall $K = \frac{\mu_2}{\mu_1} + \alpha_2 \left(1 - \frac{\mu_2}{\mu_1}\right)$,

 $S = \left(1 - \frac{\mu_2}{\mu_1}\right)$, we obtain the solution for the corresponding elliptic inhomogeneity problem under the assumption of uniform remote stresses (Ru et al, 2005) characterized by the complex potentials:

$$\phi_1\left(z\right) = A_1 z, \quad \psi_1\left(z\right) = B_1 z \quad |z| \to \infty$$

where

$$A_{1}\lambda = \left[\left(\frac{\mu_{2}}{\mu_{1}} + \alpha_{2} \left(1 - \frac{\mu_{2}}{\mu_{1}} \right) \right) C_{1} + \left(1 - \frac{\mu_{2}}{\mu_{1}} \right) \frac{\beta_{2}R}{\overline{C_{1}}} \right] \lambda + i \left(1 - \frac{\mu_{2}}{\mu_{1}} \right) \overline{D_{1}}\lambda_{1}$$
$$B_{1}\lambda = -i\alpha_{1}\overline{X_{1}} - \frac{i\beta_{1}\lambda_{1}}{A_{1}} + i \left(\alpha_{2}\overline{C_{1}} + \frac{\beta_{2}\lambda_{1}}{C_{1}} \right) \lambda_{1} + D_{1}\lambda,$$

and

$$X_1 = \left[\left(\frac{\mu_2}{\mu_1} + \alpha_2 \left(1 - \frac{\mu_2}{\mu_1} \right) \right) C_1 + \left(1 - \frac{\mu_2}{\mu_1} \right) \frac{\beta_2 R}{\overline{C_1}} \right] \lambda_1 + i \left(1 - \frac{\mu_2}{\mu_1} \right) \overline{D_1} \lambda.$$

Finally, from (3.5), we obtain the complex potentials ϕ_2 and ψ_2 inside the inhomogeneity, for $C_n = C_{n,\alpha} = D_n = D_{n,\alpha} = 0, \quad n \ge 2,$

$$\phi_2(z) = C_1 z, \quad \psi_2(z) = D_1 z, \quad z \in S_2.$$

A.2 DERIVATION OF RESULTS IN CHAPTER. 4

In this section, we show that the solution derived in Chapter 4 can be reduced to the results presented in Ru et al (2005). We first proceeds by noting the difference between the mapping functions (conformal). In Ru et al (2005), the infinite region outside an ellipse in the z - plane is mapped (conformally) to the infinite region outside the unit circle in the $\xi - plane$ by the following mapping function (see Muskhelishvili, 1953):

$$z = \omega(\xi) = \lambda \xi + \frac{\lambda_1}{\xi}, \quad \lambda \neq 0,$$

which, with $\xi = \rho e^{i\theta}$ and $\rho = 1$ will map the ellipse

$$x_1 = \left(\lambda \rho + \frac{\lambda_1}{\rho}\right) \cos \theta, \quad x_2 = \left(\lambda \rho - \frac{\lambda_1}{\rho}\right) \sin \theta,$$

(where $\lambda, \lambda_1 \in \mathbb{R}$ and $\theta \in [0, 2\pi]$) and its exterior region in the complex z-plane, onto and outside, respectively, the unit circle in the complex $\xi - plane$ (note that ρ is designed to have unit value). Such a mapping function is convenient when the forms of complex potentials ϕ_2 and ψ_2 inside the inhomogeneity are prescribed, yet it is not suitable for the purpose of identifying the forms of complex potentials describing the state of stresses inside inhomogeniety.

Therefore, in Chapter 4, we adopt the following mapping function (conformal).

$$z = \omega(\xi) = R\left(\xi + \frac{1}{\xi}\right), \quad R > 0,$$

which maps the ellipse

$$x_1 = R\left(\rho + \frac{1}{\rho}\right)\cos\theta, \quad x_2 = R\left(\rho - \frac{1}{\rho}\right)\sin\theta,$$

where R, $\rho > 0$ and $\theta \in [0, 2\pi]$ and its exterior region in the complex z-plane, onto and outside, respectively, the circle of radius ρ in the complex $\xi - plane$. Consequently, the region S_1 is transformed into the exterior of the circle with radius R_* and the region $S_2 \setminus D$ into the ring $1 < |\xi| < R_*$ in the $\xi - plane$. The segment D and the boundary Γ are mapped onto the circles $|\xi| = 1$ and $|\xi| = R_*$, respectively, where (see Figure 6),

$$R_* = \frac{a + \sqrt{a^2 - 4R^2}}{2R} > 1, \qquad a = R\left(R_* + \frac{1}{R_*}\right).$$

The only difference here is that ρ is not a fixed value so that the mapping function accommodates the region inside an elliptic inhomogeneity. By comparing the above mapping functions, we derive the relations:

$$\lambda = R\rho, \quad \lambda_1 = \frac{R}{\rho},$$

and, since at the interface $\rho = R_*$, the above equations are now taken to be

$$\lambda = RR_*, \quad \lambda_1 = \frac{R}{R_*}.$$

Next, the complex potentials in Chapter 4 and in Ru et al (2005) are given, respectively,

$$\phi_1(z) = Az + O_{(1)}, \quad \psi_1(z) = Bz + O_{(2)}, \quad |z| \to \infty,$$

$$\phi_2(z) = C_o + \frac{C_1}{R}z, \quad \psi_2(z) = D_o + \frac{D_1}{R}z, \quad |z| \in S_2,$$
(Chapter 4),

and

$$\phi_1\left(z\right)=Az, \ \ \psi_1\left(z\right)=Bz, \ \ |z|\to\infty,$$

$$\phi_2(z) = Cz, \ \ \psi_2(z) = Dz, \ \ |z| \in S_2, \ (\text{Ru et al}, 2005).$$

From Cases. 1-4 in Chapter 4 together with (4.38), we have that, for $O_{(1)} = O_{(2)} = 0$,

$$\phi_1(z) = Az, \quad \psi_1(z) = Bz, \quad |z| \to \infty,$$

$$\phi_2(z) = \frac{C_1}{R}z, \quad \psi_2(z) = \frac{D_1}{R}z, \quad |z| \in S_2,$$

$$(\because C_o = D_o = 0).$$

Then, we define the complex constants in such a way that

$$\frac{C_1}{R} = C, \quad \frac{D_1}{R} = D,$$

and recall

$$\lambda = RR_*, \quad \lambda_1 = \frac{R}{R_*}.$$

Therefore, from (4.38)

$$ARR_* = \left(KC_1 + \frac{iS\overline{D_1}}{R_*^2} + \frac{S\beta_2 R^2}{\overline{C_1}}\right)R_*$$

$$= KCRR_* + \frac{iS\overline{D}R}{R_*} + \frac{S\beta_2 RR_*}{\overline{C}} = KC\lambda + iS\overline{D}\lambda_1 + \frac{S\beta_2\lambda}{\overline{C}} = A\lambda,$$

$$\frac{K_{-1}}{R_*} = \left(KC_1 + iS\overline{D_1}R_*^2 + \frac{S\beta_2R^2}{\overline{C_1}}\right)\frac{1}{R_*}$$

$$= KC\frac{R}{R_*} + iS\overline{D}RR_* + \frac{S\beta_2R}{\overline{C}R_*} = KC\lambda_1 + iS\overline{D}\lambda + \frac{S\beta_2}{\overline{C}}\lambda_1 = X_1,$$

$$BRR_{*} = \left(-\frac{i\alpha_{1}\overline{X_{-1}}}{R_{*}^{2}} - \frac{i\beta_{1}R}{AR_{*}^{2}} + \frac{i\alpha_{2}\overline{C_{1}}}{R_{*}^{2}} + \frac{i\beta_{2}R^{2}}{C_{1}R_{*}^{2}} + D_{1}\right)R_{*}$$
$$= -\frac{i\alpha_{1}\overline{X_{1}}R_{*}}{R_{*}} - \frac{i\beta_{1}R}{AR_{*}} + \frac{i\alpha_{2}\overline{C}R}{R_{*}} + \frac{i\beta_{2}R}{CR_{*}} + DRR_{*}$$
$$-i\alpha_{1}\overline{X_{1}} - \frac{i\beta_{1}}{A}\lambda_{1} + i\alpha_{2}\overline{C}\lambda_{1} + \frac{i\beta_{2}}{C}\lambda_{1} + D\lambda = B\lambda.$$

Consequently, replacing $K = \frac{\mu_2}{\mu_1} + \alpha_2 \left(1 - \frac{\mu_2}{\mu_1}\right)$, $S = \left(1 - \frac{\mu_2}{\mu_1}\right)$,

$$\begin{split} A_1\lambda &= \left[\left(\frac{\mu_2}{\mu_1} + \alpha_2 \left(1 - \frac{\mu_2}{\mu_1}\right)\right) C_1 + \left(1 - \frac{\mu_2}{\mu_1}\right) \frac{\beta_2 R}{\overline{C_1}} \right] \lambda + i \left(1 - \frac{\mu_2}{\mu_1}\right) \overline{D_1} \lambda_1 \\ X_1 &= \left[\left(\frac{\mu_2}{\mu_1} + \alpha_2 \left(1 - \frac{\mu_2}{\mu_1}\right)\right) C_1 + \left(1 - \frac{\mu_2}{\mu_1}\right) \frac{\beta_2 R}{\overline{C_1}} \right] \lambda_1 + i \left(1 - \frac{\mu_2}{\mu_1}\right) \overline{D_1} \lambda, \\ B_1\lambda &= -i\alpha_1 \overline{X_1} - \frac{i\beta_1 \lambda_1}{A_1} + i \left(\alpha_2 \overline{C_1} + \frac{\beta_2 \lambda_1}{C_1}\right) \lambda_1 + D_1\lambda, \end{split}$$

we find that these results are identical to those presented in Ru et al (2005) for the solution of the elliptic inhomogeneity problem under uniform remote stresses.